Queensland Competition Authority

Queensland Rail 2025 Draft Access Undertaking

Draft decision

June 2024

We wish to acknowledge the contribution of the following staff to this report:

Pag Arao-Arao, Eugene Clark, Duncan Coutts, Richard Creagh, Russell Jutlah, Adam Liddy, Stephen Wisenthal

© Queensland Competition Authority 2024

The Queensland Competition Authority supports and encourages the dissemination and exchange of information. However, copyright protects this document. The QCA has no objection to this material being reproduced, made available online or electronically, but only if the QCA is recognised as the owner of the copyright and this material remains unaltered.

Contents

DRAF	TDECISION	1
SUBN	1ISSIONS	3
1	OUR INVESTIGATION	5
1.1 1.2 1.3 1.4	Background Queensland Rail's 2025 DAU Our regulatory task Human rights considerations	5 5 6 10
2	PREAMBLE, APPLICATION AND SCOPE	11
2.1 2.2 2.3	Preamble Scope of the 2025 DAU Other administrative matters	12 14 18
3	NEGOTIATION FRAMEWORK	20
3.1 3.2 3.3 3.4	Overview of the negotiation framework Facilitating the negotiation of access terms Arrangements for prioritising access applications Dispute resolution mechanism	21 22 26 28
4	STANDARD ACCESS AGREEMENT (SCHEDULE H)	30
 4.1 4.2 4.3 4.4 4.5 4.6 	Consistency Train path optimisation Insurance Dispute resolution escalation Assignment of Queensland Rail's rights under SAA Performance indicators	32 32 35 40 41 42
5	PRICING RULES	44
5.1 5.2 5.3 5.4	Overview of pricing rules in the 2025 DAU Guiding the negotiation of access charges Enabling greater price differentiation Service-specific criteria for the Mount Isa line	44 45 46 52
6	OPERATING REQUIREMENTS	54
6.1 6.2 6.3	Network management principles Operating requirements manual User groups	55 62 63
7	REPORTING REQUIREMENTS	67
7.1 7.2 7.3	Overview of reporting obligations in the 2025 DAU Performance reporting Further reporting on ad hoc planned possessions	67 68 70

8	WEST MORETON TARIFF APPROACH	72
8.1	Queensland Rail's proposal	74
8.2	Developing the proposed reference tariff	74
8.3	Capital spending: indicator and approval	79
8.4	Capacity assessment	82
8.5	Risk mitigation	83
8.6	Tariff structure	88
8.7	Loss capitalisation and 2020 undertaking	90
8.8	Appropriate reference tariff	92
9	TARIFF BUILDING BLOCKS	95
9.1	Building blocks approach to regulatory pricing	96
9.2	Coal volumes	97
9.3	Asset base	104
9.4	Rate of return	107
9.5	Capital expenditure	122
9.6	Asset lives	125
9.7	Appreciation	126
9.8	Maintenance expenditure	127
9.9	Operating expenditure	129
9.10	Metropolitan system tariff	131
APPE	ENDIX A: THE ACCESS REGIME	132
APPE	NDIX B: TRACK QUALITY	133
APPE	ENDIX C: BETA DATA	137
APPENDIX D: SUBMISSIONS RECEIVED		139
GLOSSARY		140
REFERENCES		141

Draft decision

On 10 November 2023, Queensland Rail submitted a voluntary draft access undertaking (2025 DAU) to us for approval in accordance with section 136 of the *Queensland Competition Authority Act 1997* (QCA Act).

We assessed the appropriateness of all aspects of 2025 DAU, and considered all submissions received in accordance with the statutory requirements, including having regard to the approval criteria in section 138(2) of the QCA Act.

Our draft decision is to refuse to approve Queensland Rail's 2025 DAU, having regard to the statutory criteria and the stakeholder submissions we received.

While we have identified a number of potential amendments to the DAU, there are some provisions where our preliminary position is that they are appropriate to approve. For the most part, these are unchanged from the 2020 undertaking. Matters that may be appropriate to approve include:

- the negotiation provisions in Part 2
- the pricing rules in Part 3
- the dispute resolution provisions in Part 6.

However, what stands out in this draft decision is the number of matters that Queensland Rail and its customers have not resolved. A key concern of stakeholders has been that Queensland Rail has not engaged in genuine consultation but has rather proposed changes from the 2020 undertaking on a take it or leave it basis. Stakeholders have suggested a number of specific amendments to the regulatory arrangements proposed by Queensland Rail in the 2025 DAU. We encourage Queensland Rail to work with its customers to resolve such matters and find agreed approaches for us to consider.

We would be inclined to approve the provisions in any pending undertaking where the parties have reached agreement, subject to our statutory obligations and having regard to the interests of those not party to negotiations. Amendments settled through negotiation are preferable as stakeholders are best placed to manage and allocate risks among themselves.

Non-tariff matters that remain unresolved or where we consider it is appropriate to amend the 2025 DAU include:

- several aspects of the standard access agreement (SAA) (Schedule H)
- notification times and treatment of the master train plan (MTP) in the network management principles in Part 4 and Schedule F
- reporting on ad hoc planned possessions in Part 5.

A significant unresolved matter is the West Moreton reference tariff and the related provisions in Schedules D and E of the 2025 DAU. Stakeholders said Queensland Rail gave insufficient details of its proposed costs, pricing and risk mitigation measures, and provided the information only shortly before the DAU was lodged.

This late consultation means the tariff proposal does not appear to reflect feedback from customers on their preferences. It is also uncertain whether Queensland Rail's proposed volumes will be required and whether, if they are contracted, they can be delivered. We have suggested a potential way forward – that is to escalate existing tariffs by the consumer price index (CPI) – but our preference is that the parties negotiate a reference tariff that has regard to the legitimate interests of both Queensland Rail and its customers, and promotes efficient investment in, and operation and use of, the West Moreton system.

Our preliminary assessment of the 2025 DAU against the relevant criteria is set out in detail throughout this draft decision, including reasons for our draft decision, ways in which we consider the DAU should be amended, and matters where we consider the parties should pursue agreed positions. Where we have suggested it is best that Queensland Rail and its stakeholders find agreed positions, we have in most cases provided an indication of what we think might be an appropriate approach.

The draft decision does not represent any final views and has no force as such.

Next steps

Our final decision will be informed by submissions on this draft decision – our analysis and application of statutory assessment criteria may change before the final decision.

We invite submissions from interested parties regarding the draft decision by **Tuesday 23 July 2024.** We encourage stakeholders, particularly Queensland Rail, to engage with other parties and to present agreed positions as fully as possible, including suggested drafting of proposed amendments to the 2025 DAU. Parties may also justify proposals in terms of the approval criteria in section 138(2) of the QCA Act. We will take account of all submissions received within the stated timeframe. Please refer to the 'Submissions' section below for additional information.

We encourage stakeholders to use the period for initial submissions as an opportunity to find common ground and bring forward agreed positions. We ask that parties have regard to the benefits of early involvement in the process. As we said in our statement of regulatory intent in November, should a stakeholder raise a significant issue, or propose a major policy change, early engagement will afford other stakeholders natural justice by giving them sufficient opportunity to consider and comment on matters that affect them.

We also intend to provide a period for collaborative submissions. This final round of submissions will give Queensland Rail and its customers an opportunity to identify any further areas of consensus. We anticipate the due date for collaborative submissions will be in September 2024.

As part of the consultation after this draft decision, we may release targeted position papers or discussion papers where we consider it is appropriate for parties to have an opportunity to comment on particular matters.

Submissions

Closing date for submissions: 23 July 2024

The role of the QCA

The Queensland Competition Authority (QCA) is Queensland's economic regulator. We are an independent statutory body¹ that promotes competition as the basis for enhancing efficiency and growth in the Queensland economy. One of our primary roles is to help prevent monopoly businesses operating in Queensland, particularly in the provision of key infrastructure, from inappropriately using their market power through unfair pricing or restrictive access arrangements.

The draft decision outlines our preliminary assessment of Queensland Rail's 2025 DAU in accordance with the relevant statutory criteria and the reasons why we do not consider it is appropriate to approve the undertaking. The draft decision represents our preliminary view and is intended to give stakeholders insight into that view and to encourage further contributions. Our final decision will be informed by submissions on the draft decision, and our analysis and application of statutory assessment criteria may change prior to the final decision. The draft decision does not represent any final views and has no force *per se*.

Submissions

Public involvement is an important element of our decision-making processes. Therefore, we invite submissions from interested parties regarding the draft decision.

We will take account of all submissions received within the stated timeframes.

Submissions, comments or inquiries regarding the draft decision should be directed to:

Queensland Competition Authority

GPO Box 2257, Brisbane QLD 4001

Tel 07 3222 0555

www.qca.org.au/submissions

Confidentiality

In the interests of transparency, and to promote informed consultation, we intend to make all submissions publicly available. However, if a person making a submission believes information in it is confidential, they should claim confidentiality over the relevant information and state the basis for that claim. We will assess confidentiality claims in accordance with the *Queensland Competition Authority Act 1997*.²

¹ Established under the *Queensland Competition Authority Act* 1997.

² Considering particularly the public interest and if disclosure of the relevant information is likely to damage a person's commercial activities.

Queensland Rail's 2025 Draft Access Undertaking

Claims for confidentiality should be clearly noted on the front page of a submission, and relevant sections of the submission marked as confidential. The submission should also be provided in both redacted and unredacted versions. In the redacted version, all information claimed as confidential should be removed or hidden. In the unredacted version, all information should be exposed and visible. These measures will make it easier for us to make the remainder of the document publicly available. A confidentiality claim template is available at **www.qca.org.au/submission-policy.**

The template gives guidance on the type of information that may help us to assess a confidentiality claim. We encourage stakeholders to use this template when making confidentiality claims.

Public access to submissions

Subject to any confidentiality constraints, submissions will be available for public inspection at our Brisbane office, or on our website at **www.qca.org.au**. If you experience any difficulty gaining access to documents, please contact us on **07 3222 0555**.

1 Our investigation

1.1 Background

Queensland Rail owns and operates a 6,600 kilometre rail network, including the commuter lines in south-east Queensland, and the West Moreton, Mount Isa and North Coast systems. Queensland Rail provides access to declared services for the purposes of Queensland's third-party access regime established under Part 5 of the QCA Act. Appendix A provides an outline of this access regime and Queensland Rail's declared services.

Queensland Rail's 2020 access undertaking (AU2) came into effect on 1 July 2020 and sets out the terms and conditions under which Queensland Rail provides access to the declared service on its rail infrastructure. It also outlines the process required for an access seeker to negotiate access to the services, and the way disputes in relation to access are to be resolved. AU2 will expire on 30 June 2025.³

On 10 November 2023, Queensland Rail submitted the 2025 DAU to us for approval in accordance with section 136 of the QCA Act. It is intended that the undertaking will take effect immediately, as the 2025 access undertaking (AU3), after the 2020 undertaking (AU2) expires.

1.2 Queensland Rail's 2025 DAU

An access undertaking for a service means a written undertaking that sets out details of the terms on which an owner or operator of the service undertakes to provide access to the service.⁴

Queensland Rail's 2025 DAU sets out the proposed terms and conditions under which Queensland Rail will provide access to the services covered by the undertaking during its term. The term of the DAU commences on the DAU's approval date until its expiry on 30 June 2030, unless an earlier terminating date arises.

An overview of the arrangements proposed in the 2025 DAU is outlined in Figure 1. Queensland Rail's proposed regulatory arrangements largely reflect the current regulatory arrangements, set out in the 2020 undertaking.

In this regard, Queensland Rail submitted that it was only seeking changes from the 2020 undertaking on an exceptions basis, where it considered improvements could be made. Queensland Rail said the 2020 undertaking provisions had been tried and tested and it had proposed only minor changes to provide business certainty to its customers.⁵

As part of these arrangements, a reference tariff is to apply to coal handling services on the West Moreton Route service.⁶ Queensland Rail has recalculated the reference tariff and proposed a price 24% higher than customers have been paying under the 2020 undertaking.

³ Unless an earlier terminating date is triggered under AU2.

⁴ QCA Act, Sch. 2.

⁵ Queensland Rail, sub. 1, p. 3.

⁶ The West Moreton Route service means the use of the West Moreton system and the Metropolitan system (see Appendix A for further explanation of the parts (or route services) of the Queensland Rail service that have been declared under s. 84(1)(b) of the QCA Act).

Figure 1: Overview of the proposed arrangements included in the 2025 DAU

nnlie	cation of the undertaking
Part see whe	 1 defines the duration and scope of the access undertaking, the non-discriminatory treatment of access ekers and access holders, as well as outlining Queensland Rail's obligations for extending the network ere an access seeker is willing to fund an extension. 6 outlines transitional arrangements for matters undertaken as part of the previous regulatory period.
ame	ework for negotiating access
• So • So • So co	2 sets out the framework for negotiating access rights. To support the negotiation process: chedule A outlines obligations for Queensland Rail to provide preliminary and capacity information chedule B specifies information to be provided by access seekers as part of an access application chedule I identifies principles of negotiation if an access seeker or access holder wants to build a rail nection to its private infrastructure.
	6 outlines a dispute resolution mechanism to address instances where disputes between the parties arise ard access agreement
Sche	adule H provides a standard access agreement to facilitate the negotiation of access that sets out the ndard terms and conditions of access to Queensland Rail's network.
pera	ting and reporting requirements
ent • So • So	 t 4 establishes the operating requirements that govern how Queensland Rail is to deliver train service itlements. The operating requirements and expectations are further detailed in: chedule F outlines the network management principles chedule G outlines operating requirements manual. t 5 sets out Queensland Rail's reporting responsibilities throughout the regulatory period.
ricing	g rules for negotiating access
	3 prescribes pricing rules that are to apply in negotiating access charges. Edule J outlines how annual regulatory fees are to be allocated between train services.
efere	ence tariffs for the Metropolitan and West Moreton systems
des Sche app	edule D reports the reference tariffs that are to apply to those train services that accord with the scription of a reference train service in the Metropolitan and West Moreton systems. Edule E outlines the process for rolling-forward the regulatory asset base (including the process for proving capital expenditure) throughout the regulatory period for the purpose of calculating the relevant erence tariffs.

1.3 Our regulatory task

Submissions

In accordance with section 136 of the QCA Act, we are required to consider the 2025 DAU and either approve, or refuse to approve, it.⁷ As part of our assessment, we must publish the 2025 DAU and consider relevant submissions on it (ss. 138(3)(c) and (d)).

On 13 November 2023, we published the 2025 DAU, a notice of investigation (that invited interested parties to make submissions by 2 February 2024), ^{8,9} and a Statement of Regulatory

⁷ If we refuse to approve the 2025 DAU, we must provide a written notice stating the reasons for the refusal and the way in which we consider it is appropriate to amend the 2025 DAU (s. 136(5)). In this circumstance, notification would occur when we release the final decision on the 2025 DAU.

⁸ Section 146 of the QCA Act provides for us to issue a notice of investigation to commence an investigation for deciding whether to approve the DAU. A notice of investigation states our intention to conduct the investigation and invites interested parties to make written submissions on the proposed DAU.

⁹ On 23 November 2023, we published a list of topics to assist stakeholders in preparing their submissions.

Intent.¹⁰ In response to our initial invitation for comment, we received submissions on the 2025 DAU from the following parties:

- Aurizon Coal and Bulk
- Aurizon Network Pty Ltd
- Centrex Limited
- GrainCorp Operations Limited
- New Hope Group
- Pacific National
- Phosphate International Ltd t/a North West Phosphate
- Qube Logistics
- Yancoal Australia Limited.

On 22 February 2024, we provided stakeholders, including Queensland Rail, a further three weeks to comment on the new matters raised in the submissions received. In response to our follow-up invitation, we received further submissions from the following parties:

- Glencore
- Mount Isa Line Users
- New Hope Group
- Pacific National
- Queensland Rail
- Rail Operator Group
- Yancoal.

Appendix D lists the submissions we received on the 2025 DAU and provides the submission numbers used in footnotes.

Factors affecting approval

Section 138 of the QCA Act outlines the factors affecting the approval of a DAU. In particular, we may approve the 2025 DAU only if we consider it appropriate to do so having regard to each of the matters set out in section 138(2) of the QCA Act (Box 1).

¹⁰ The Statement of Regulatory Intent outlines our intended approach to managing information-gathering processes, stakeholder consultation and assessment timeframes throughout our investigation.

Box 1: Section 138(2) of the QCA Act

The authority may approve a DAU only if it considers it appropriate to do so having regard to each of the following:

- a. the object of this part;
- b. the legitimate business interests of the owner or operator of the service;
- c. if the owner and operator of the service are different entities the legitimate business interests of the operator of the service are protected;
- d. the public interest, including the public interest in having competition in markets (whether or not in Australia);
- e. the interests of persons who may seek access to the service, including whether adequate provision has been made for compensation if the rights of users of the service are adversely affected;
- f. the effect of excluding existing assets for pricing purposes;
- g. the pricing principles mentioned in section 168A;
- h. any other issues the authority considers relevant.

The 'object of this part' as referred to in section 138(2)(a) is set out in section 69E:

The object of this part is to promote the economically efficient operation of, use of and investment in, significant infrastructure by which services are provided, with the effect of promoting effective competition in upstream and downstream markets.

The pricing principles set out under section 168A are:

The pricing principles in relation to the price of access to a service are that the price should:

- a. generate expected revenue for the service that is at least enough to meet the efficient costs of providing access to the service and include a return on investment commensurate with the regulatory and commercial risks involved; and
- b. allow for multi-part pricing and price discrimination when it aids efficiency; and
- c. not allow a related access provider to set terms and conditions that discriminate in favour of the downstream operations of the access provider or a related body corporate of the access provider, except to the extent the cost of providing access to other operators is higher; and
- d. provide incentives to reduce costs or otherwise improve productivity.

This draft decision outlines our preliminary assessment of the 2025 DAU, having regard to the matters in section 138(2) of the QCA Act and all stakeholders' submissions that were received within the consultation periods.

In assessing the 2025 DAU, we have considered all aspects of the undertaking afresh and had regard to the section 138(2) factors in every aspect of this draft decision. In some cases, the assessment of whether it is appropriate to approve the 2025 DAU, having regard to the factors

affecting approval, gives rise to competing considerations. In such cases, we weighed up the competing considerations as appropriate.

Agreed outcomes

As outlined in Appendix A, the access regime in Queensland is based on a negotiate-arbitrate framework, which envisages that, in the first instance, access to a declared service should be procured on the basis of terms and conditions that are commercially agreed between the access seeker and the provider of the declared service.¹¹

We have repeatedly emphasised our desire for consultation and consensus, including in our decision on Queensland Rail's 2020 DAU, and in our communication about the 2025 DAU. Such approaches have been demonstrated to deliver superior outcomes for Aurizon Network and Dalrymple Bay Infrastructure (DBI) and their respective customers and we see no reason why similar outcomes cannot be achieved by Queensland Rail and its customers if there is focused consultation on the issues raised in this draft decision.

Consistent with the negotiate-arbitrate framework, we place weight on agreed outcomes and consider it highly beneficial for parties to engage with each other, to the extent possible, to explore salient issues and potential areas of consensus in relation to the 2025 DAU. In respect of all aspects of the access undertaking, we still, in accordance with our statutory role, undertake an independent review to confirm that they comply with the criteria in section 138(2) of the QCA Act, including the interests of potential entrants and the public.

Draft decision structure

Our draft decision is structured in two parts (Figure 2):

- Part A our assessment of the 2025 DAU provisions
- Part B our assessment of the proposed reference tariffs and related provisions for the West Moreton and Metropolitan systems.

Figure 2: Structure of the draft decision

Part A – Our assessment of the 2025 DAU provisions		
Chapter	Торіс	
2	Preamble, application and scope of the undertaking	
3	Framework for negotiating access	
4	Standard access agreement	
5	Pricing rules for negotiating access	
6	Operating requirements	
7	Reporting requirements	

¹¹ QCA, Summary guide: Access disputes under the QCA Act, June 2019, p. 1.

Part B – Our assessment of the proposed reference tariffs		
Chapter	Торіс	
8	Reference tariff approach	
9	Tariff building blocks	

1.4 Human rights considerations

Although we do not consider this draft decision to be a decision for the purposes of the *Human Rights Act 2019* (HR Act), pursuant to section 58(1)(b) of the HR Act, the following human rights may potentially be relevant to our final decision:

- rights potentially related to climate change (right to life, equality and non-discrimination and right of children to protection of their best interests)
- the right to freedom of movement (for passengers in Queensland).

It is not envisaged that the final decision will limit any of the rights mentioned above, particularly as:

- rights potentially related to climate change these rights are unlikely to be limited by a decision concerning approval of an access undertaking, as any such decision would likely not have a material effect on the volume of coal exported and consumed overseas. This will primarily depend on market factors such as demand and price
- right of freedom of movement the effect of the final decision is likely to be positive or neutral.

It is therefore envisaged that a final decision (that is consistent with the approach taken in this preliminary assessment) is likely to be compatible with human rights under section 8(a) of the HR Act.

2 Preamble, application and scope

The preamble of the 2025 DAU is an introductory section that provides background and high-level context for the undertaking.

Part 1 of the 2025 DAU outlines the application and scope of the undertaking. It includes provisions relating to consistency and fairness in the treatment of users;¹² requirements in negotiating, developing and funding network extensions in circumstances where an access seeker notifies Queensland Rail that it is willing to fund an extension;¹³ and master planning and extension coordination, applicable to situations where Queensland Rail receives a request to prepare a 'system master plan' relating to its proposed expansion projects.¹⁴

Part 6 of the 2025 DAU includes various administrative provisions, including transitional provisions that set out how matters undertaken or established under AU2 will continue during the AU3 regulatory period.¹⁵

Part 7 of the 2025 DAU provides definitions of key terms¹⁶ and outlines rules for interpreting the undertaking.¹⁷

Overview of the draft decision

Our draft decision is that it is appropriate to approve the preamble and transitional provisions in Part 6 of the 2025 DAU and that it may be appropriate to approve Part 1 of the 2025 DAU.

Preamble, application and scope – summary

Proposal	Clause	QCA draft decision
Preamble		
Queensland Rail did not propose changes to the preamble. ¹⁸	n/a	It is appropriate to approve the preamble. See section 2.1.
Stakeholders proposed further incentivising road-to-rail modal shift ¹⁹	n/a	It is not necessary to require amendments to the preamble regarding a road to rail modal shift. See section 2.1.
Scope		
Stakeholders proposed requiring greater consistency and harmonisation with other rail infrastructure managers. ²⁰	n/a	It is not appropriate to amend the scope of the 2025 DAU to require greater harmonisation. See section 2.2.2.

¹² 2025 DAU, cl. 1.3.

¹³ 2025 DAU, cl. 1.4.

¹⁴ 2025 DAU, cl. 1.5.

¹⁵ 2025 DAU, cl. 6.5. These transitional arrangements include rules relating to matters and access applications established under AU2, reporting periods and approval dates.

¹⁶ 2025 DAU, cl. 7.1.

¹⁷ 2025 DAU, cl. 7.2.

¹⁸ Queensland Rail, sub. 1.

¹⁹ Aurizon Coal and Bulk, sub. 2, p. 26-27; Mount Isa Line Users, sub. 17, p. 2; Rail Operators Group, sub. 15, pp. 1-4; Pacific National, sub. 13, pp. 2, 3-6.

²⁰ Aurizon Coal and Bulk, sub. 2, pp. 10-12; GrainCorp, sub. 4, p. 3; Pacific National, sub. 7, pp. 14-15; Rail Operators Group, sub. 15, pp. 11-12; Qube, sub. 8, pp. 2-5.

Stakeholders proposed including interstate services using the dual gauge link between Acacia Ridge and the Port of Brisbane in the 2025 DAU. ²¹	cl. 1.2	These services are likely not a declared service under the Treasurer's declaration of 2020. We invite submissions on this issue. See section 2.2.
Other matters		
Queensland Rail proposed maintaining transitional provisions ²²	cl. 6.5	It is appropriate to approve this. See section 2.3.

2.1 Preamble

The preamble provides high-level context for Queensland Rail's 2025 DAU. Queensland Rail did not propose any changes to the preamble of the 2020 access undertaking.

We do not consider that it is appropriate to require any amendments to the preamble. Importantly, the high-level context provided by Queensland Rail is factually correct. Moreover, we do not consider that making minor amendments to the preamble will affect the operation of the 2025 DAU.

In their submissions, stakeholders raised issues broadly concerning the extent to which the 2025 DAU incentivises a modal shift of freight from road transport to rail transport.

Incentivising a road-to-rail modal shift

We do not consider it necessary to require that amendments to the preamble include a commitment by Queensland Rail to support road-to-rail conversion and grow rail volumes.

Aurizon Coal and Bulk considered that the preamble should include a clear commitment by Queensland Rail to:

- promote competition in the rail haulage market
- promote the economically efficient investment, use and operation of the network
- grow rail volumes including by supporting and incentivising emerging demand, retaining volumes on rail and supporting 'road to rail' modal conversion.²³

The Mount Isa Line Users submitted that there was no mechanism in the 2025 DAU that could be used to hold Queensland Rail to account in acting in a way that supported this objective.²⁴ The Rail Operators Group considered that the 2025 DAU should provide stronger obligations on Queensland Rail to offer access arrangements that effectively encouraged a modal shift of freight from road to rail. The Rail Operators Group considered that there must be a strong price incentive to offset the unreliability of the North Coast line and Mount Isa line.²⁵

Pacific National considered that Queensland Rail had not sufficiently justified its 2025 DAU having regard to the matters in section 138(2) of the QCA Act, primarily because it provided neither incentives to encourage modal shift from road to rail nor support to grow competition with road transport.²⁶ Pacific National and the Rail Operators Group also submitted that additional mechanisms should be incorporated into the 2025 DAU to encourage a modal shift of freight from

²¹ Aurizon Coal and Bulk, sub. 2, pp. 30-33; Rail Operators Group, sub. 15, p. 12.

²² Queensland Rail, sub. 1, cl. 6.5.

²³ Aurizon Coal and Bulk, sub. 2, p. 26; Mount Isa Line Users (sub. 17, p. 2) supported Aurizon Coal and Bulk's submission.

²⁴ Mount Isa Line Users, sub. 17, p. 2.

²⁵ Rail Operators Group, sub. 15, pp. 1, 4.

²⁶ Pacific National, sub. 13, pp. 2, 3-6.

road to rail. The Rail Operators Group considered this would promote the economically efficient operation and use of the network.²⁷

Queensland Rail considered that the assumption that there was a misalignment between Queensland Rail's objective of being a champion of rail freight and its commercial positions was unfounded. Queensland Rail submitted that it inherently pursued growth and sustainability, discrediting any perceived conflict between these objectives.²⁸

We consider it appropriate for the overarching regulatory framework to restrict Queensland Rail from exercising market power to the extent that it prevents or hinders access seekers from accessing Queensland Rail's network on reasonable terms and conditions. This includes enabling access seekers to obtain access to Queensland Rail's network, where they are willing to pay access charges that recover the efficient costs incurred in providing that access (and where sufficient capacity is available). We consider constraining market power in this manner promotes the economically efficient use and operation of Queensland Rail's network, and appropriately balances the legitimate business interests of Queensland Rail and access seekers.

In making this draft decision, we have outlined those regulatory arrangements that we consider are required as part of an approved access undertaking for the AU3 regulatory period (within Queensland Rail's overarching regulatory framework) to provide for this.

As such, we do not consider it is necessary to require amendments to the preamble to include a commitment by Queensland Rail to support road to rail conversion. In any case, we do not consider that amending the preamble in this manner would affect the operation of the undertaking.

We have set out in the relevant chapters of this draft decision whether it is appropriate to require further amendments to the 2025 DAU to encourage more freight from road to rail.

In making this draft decision, we have had regard to the public interest in accordance with section 138(2)(d) of the QCA Act.

The Rail Operators Group considered that moving more freight by rail would serve the public interest through improvements to road safety, less congestion on the roads and environmental benefits.²⁹ Pacific National considered that the 2025 DAU did not meet the public interest objective of shifting traffic from road to rail.³⁰ Pacific National considered the preamble should be expanded to acknowledge the positive benefits rail freight provides relative to road.³¹

Queensland Rail submitted that while safety and environmental benefits associated with moving more freight by rail were crucial, a balance with economic viability was imperative.³²

We consider that it is important to have regard to any positive externalities associated with increasing rail haulage in assessing whether it is appropriate to approve the 2025 DAU. This is recognised by approving regulatory arrangements that prevent Queensland Rail from exercising market power to the extent that it prevents or hinders access seekers from accessing Queensland Rail's network on reasonable terms and conditions.

²⁷ Pacific National, sub. 7, p. 9; Rail Operators Group, sub. 15, p. 3.

²⁸ Queensland Rail, sub. 14, p. 43.

²⁹ Rail Operators Group, sub. 15, pp. 3-4.

³⁰ Pacific National, sub. 13, p. 5. Pacific National (sub. 7. p. 4) also considered that pricing of the North Coast line should prioritise growth of rail freight, reflecting the positive economic externalities rail provides compared to road freight and the reduced costs to government and communities in terms of accidents, congestion, and emissions. Centrex (sub. 10, p. 3) submitted that the aspect of community service is currently not being adequately considered in relation to Queensland's mineral wealth.

³¹ Pacific National, sub. 7, p. 29.

³² Queensland Rail, sub. 14, pp. 43-44.

We do not consider it is appropriate to require Queensland Rail to include in the 2025 DAU further commitments or support for particular customers or industry groups (including through subsidised access charges) beyond this. In particular:

- pricing policies that do not enable Queensland Rail to generate expected revenue for the service that is at least enough to meet the efficient costs of providing access to its network³³ may reduce incentives to invest in the network and have adverse consequences for parties seeking to access the rail infrastructure³⁴
- favourable terms and conditions to particular customers or industry groups to promote one mode of freight transport over another may distort competitive outcomes in related markets.

In this regard, Queensland Rail submitted that ultimately pricing structures must consider factors like maintenance costs, infrastructure development and financial sustainability.³⁵

We consider that measures to directly assist industry or to encourage more freight from road to rail are better considered and provided through means other than access regulation, within the context of wider transport policy and industry assistance objectives.

Summary 2.1

It is appropriate to approve Queensland Rail's proposed preamble to the 2025 DAU.

It is not necessary to require amendments to the preamble to include a commitment by Queensland Rail support road to rail conversion and grow rail volumes.

2.2 Scope of the 2025 DAU

Queensland Rail's proposed scope of the 2025 DAU is substantially consistent with the scope that was approved as part of its 2020 access undertaking.³⁶

Queensland Rail considered that, similar to the 2020 undertaking, a five-year term – 1 July 2025 to 30 June 2030 – was appropriate. In defining the terminating date, the 2025 DAU also removes a reference from AU2 to a circumstance in which the Minister had not declared Queensland Rail's services.³⁷ Stakeholders did not express any concerns regarding the proposed duration of the 2025 DAU.

We consider that a five-year term appropriately balances the benefits of providing certainty to stakeholders about the terms and conditions of access for a reasonable period of time and flexibility to deal with changing circumstances. In addition, removing the reference to a circumstance

³³ Including a return on investment commensurate with the regulatory and commercial risks involved.

³⁴ We do not consider that this will promote economically efficient investment in Queensland Rail's network. Neither does it appropriately balance the legitimate business interests of Queensland Rail and access seekers.

³⁵ Queensland Rail submitted that in considering the issues around rail access pricing, there is a tension between the objective to enable rail operators to effectively compete with road, while also setting a charge that enables sufficient ongoing maintenance and renewal of the rail infrastructure.

³⁶ New Hope (sub. 5, p. 28) and Yancoal (sub. 9, p. 24) submitted that definition of 'network' needs to refer to infrastructure the use of which forms part of the declared service. We consider that the definition of 'network' does refer to infrastructure the use of which forms part of the declared service, consistent with the definition of 'network' in the 2020 access undertaking.

³⁷ 2025 DAU, cl. 7.1, 'Terminating Date'.

involving no ministerial declaration is appropriate as this contingency is no longer relevant following the Minister's declaration of Queensland Rail's below-rail services in 2020.

We did not receive any stakeholder submissions in relation to Queensland Rail's proposed provisions relating to the scope of the 2025 DAU, other than essentially seeking to expand the scope and effect of the DAU to address certain concerns. These concerns are considered below.

2.2.1 Dual gauge track from Acacia Ridge to the Port of Brisbane

Some stakeholders raised concerns in their submissions regarding the status of interstate (standard gauge) services that use Queensland Rail's dual gauge link between Acacia Ridge and the Port of Brisbane and submitted that these services should be included within the scope of the 2025 DAU.³⁸ Aurizon Coal and Bulk suggested that there was some ambiguity regarding whether or not interstate services on the Acacia Ridge-Port of Brisbane corridor were included under AU2 and the Minister's declaration of Queensland Rail's services.³⁹

The Rail Operators Group and Aurizon Coal and Bulk submitted that this corridor had critical commercial and economic significance because of the containerised rail freight services it provided.⁴⁰ They also noted concerns in dealing with Queensland Rail regarding dual gauge access to this corridor⁴¹ which, in their view, demonstrated Queensland Rail's market power in relation to this corridor⁴² and its failure to consider integration with the national rail network, contrary to government policy and strategic objectives.⁴³

In its responding submission, Queensland Rail submitted that:

- the Metropolitan system was not a declared service and should not be within the scope of the 2025 DAU⁴⁴
- its services were declared by reference to defined routes, each of which traversed a combination of network systems, and that the Metropolitan system was not declared, except insofar as it was used in combination with another rail system, in accordance with the definitions contained in the Minister's declaration of Queensland Rail's services⁴⁵
- we could not require an amendment to include the dual gauge corridor in the 2025 DAU
- the QCA Act did not require Queensland Rail to provide an access undertaking for services that were not subject to declaration
- any change to the declared services would require a further application for declaration⁴⁶
- with respect to its interactions with users (notably Aurizon Coal and Bulk) concerning access for interstate containerised freight service through the corridor, Queensland Rail submitted that it fulfilled its obligations in access applications and negotiations.⁴⁷

We acknowledge the issues raised respectively by stakeholders and Queensland Rail concerning the status of interstate services that use Queensland Rail's dual gauge link between Acacia Ridge

³⁸ Aurizon Coal and Bulk, sub. 2, pp. 30-33; Rail Operators Group, sub. 15, p. 12.

³⁹ Aurizon Coal and Bulk, sub. 2, pp. 30-33.

⁴⁰ Rail Operators Group noted that the corridor connects the national standard gauge rail network with Brisbane's only open access container rail terminal, and it also provides a narrow gauge rail network link between Queensland's largest container rail terminal at Acacia Ridge and the state's largest container port, and provides narrow gauge rail access to export coal and grain terminals (sub. 15, p. 10).

⁴¹ Rail Operators Group, sub. 15, p. 11; Aurizon Coal and Bulk, sub. 2, pp. 32-33.

⁴² Aurizon Coal and Bulk, sub. 2, pp. 32-33.

⁴³ Rail Operators Group, sub. 15, p. 11.

⁴⁴ Queensland Rail, sub. 13, p. 53.

⁴⁵ Queensland Rail, sub. 13, pp. 53-54.

⁴⁶ Queensland Rail, sub. 13, p. 54.

⁴⁷ Queensland Rail, sub. 13, pp. 27-28.

and the Port of Brisbane and whether these services should be included within the scope of the 2025 DAU.

In forming our preliminary views on this matter, we note that there are 6 route services covered under the Queensland Treasurer's June 2020 decision to declare services pursuant to section 84 of the QCA Act, as outlined in Appendix A. The Metropolitan system is not a declared route service in and of itself. However, this system is declared in circumstances where its use 'is referred to *as part* of the relevant [declared] service' [emphasis added].⁴⁸ As a result, we consider that use of the Metropolitan system for traffic that is not related to a relevant declared service, such as trains travelling to or from Acacia Ridge via the Australian Rail Track Corporation (ARTC) interstate line, likely does not constitute use of a relevant declared service.

With respect to traffic that runs only on the Metropolitan system, each declared route service is comprised of a primary system and at least one secondary system. Our initial view is that use of a declared service must entail use of a primary system, but not necessarily a secondary system. We consider that the Metropolitan system is a secondary system for most of the declared route services, noting that it is described 'as part of the relevant service'.

Based on the foregoing considerations and the submissions to date, our preliminary view is that interstate services using the dual gauge link between Acacia Ridge and the Port of Brisbane are likely not a declared service under the Treasurer's declaration of 2020. We invite parties' submissions on this issue for further consideration following which we will finalise our position in the final decision.

Any changes to the declared services sought by an interested party would need to be separately addressed in accordance with the declaration process in Part 5 of the QCA Act.⁴⁹

2.2.2 Harmonisation with other rail networks

We do not consider that it is appropriate to amend the scope of the 2025 DAU to require Queensland Rail commit to greater consistency and harmonisation with other rail infrastructure managers.

A number of stakeholders called for amendments to the 2025 DAU to provide for greater consistency in network standards and operating requirements across rail networks.⁵⁰ Pacific National submitted that the 2025 DAU should include a requirement for Queensland Rail to use best endeavours to align processes and systems with other jurisdictions.⁵¹ Aurizon Coal and Bulk considered that Queensland Rail should pursue increased consistency in access negotiation frameworks, the standard contracting terms for access and applied access management methods, wherever possible.⁵² Qube and the Rail Operators Group also considered that the 2025 DAU should commit to measurable actions to deliver government policy and strategy objectives.⁵³

⁴⁸ Queensland Government, <u>Gazette</u> (Extraordinary), vol. 384, no. 31, 1 June 2020, Notice of a decision to declare parts of a service that are themselves a service under sections 84-87, para 3(c).

⁴⁹ Under this process, any person may apply to us for a recommendation to the relevant Minister that a service provided by a facility be declared for third party access under the Queensland access regime.

⁵⁰ Aurizon Coal and Bulk, sub. 2, pp. 10-12; GrainCorp, sub. 4, p. 3; Pacific National, sub. 7, pp. 14-15; Rail Operators Group, sub. 15, pp. 11-12.

⁵¹ Pacific National, sub. 7, pp. 13-14.

⁵² Aurizon Coal and Bulk, sub. 2, p. 3. The Rail Operators Group (sub. 15, p. 11) also considered that the complexity of an interstate train path moving across multiple networks was clearly demonstrated by requirements to comply with different access agreements, rolling-stock standards, network rules, operating conditions and pricing principles.

⁵³ Qube, sub. 8, pp. 2-5; Rail Operator Group, sub. 15, p. 2, 11. In this regard, Qube submitted that improvements to the national freight network, including streamlined regulation and improved inter-operability across networks, were identified as key deliverables in the National Freight and Supply Chain Strategy and the National Rail Action Plan.

Queensland Rail's access regime provides for access seekers to obtain access to Queensland Rail's network on reasonable terms and conditions. Within this regulatory framework, an approved access undertaking is to facilitate the negotiation of access on reasonable terms and conditions.

We have assessed whether it is appropriate to approve the terms and conditions proposed by Queensland Rail in its 2025 DAU.

We do not consider it is appropriate to require Queensland Rail to amend the processes or arrangements in the 2025 DAU primarily to achieve improved alignment with those arrangements applied in other jurisdictions. From the information available, we do not consider that amending arrangements in the 2025 DAU to align with those applied in another jurisdiction would provide for improvements to the provision of access to Queensland Rail's network. In assessing whether the arrangements in the 2025 DAU are appropriate to approve, we have given consideration to the specific characteristics of Queensland Rail's network and its customers.

Moreover, we do not consider that our 2025 DAU investigation is the right forum to attempt to establish consistent processes and standards across jurisdictions. Amongst other things, this would require engagement between all the relevant parties (across jurisdictions) to identify and consider opportunities for reform. We understand there are working groups and organisations that have been established to do just this.⁵⁴ Queensland Rail considered that it would be counterproductive to attempt to achieve national consistency based on submissions from individual operators, cutting across ongoing industry-wide initiatives to achieve a national rail market.⁵⁵

We acknowledge that greater operational harmonisation between rail infrastructure managers may reduce complexity and have benefits for rail operators and end users. Pacific National submitted that a transition to harmonised safeworking, licencing and rolling-stock conditions, amongst others, would aid the efficiency of rail freight, reduce operational complexity and support future opportunities for investment.⁵⁶ Aurizon Coal and Bulk and the Rail Operator Group also considered that there was benefit from improved harmonisation of the performance metrics used by Australia's freight rail networks.⁵⁷

However, it is not clear that amending specific arrangements in the 2025 DAU to better align with those applied in another jurisdiction will necessarily realise these benefits. As yet, no nationally consistent approach has been agreed to, or adopted, by rail infrastructure managers with respect to these operational or reporting arrangements. Queensland Rail considered that benefits arising from more consistent and harmonised performance reporting would only arise if all other rail infrastructure managers adopted the same measures.⁵⁸

Furthermore, making amendments to the 2025 DAU does not prevent alternative regulatory arrangements being considered and applied in other jurisdictions in future. Therefore, any attempt to align arrangements does not guarantee consistency between jurisdictions over time.

While we do not consider that amendments to the 2025 DAU are an effective way to achieving greater operational harmonisation with other rail infrastructure managers, we encourage Queensland Rail to work toward this objective in consultation with industry. Queensland Rail submitted that it supported the Australian Railway Association (ARA) priorities, including moving towards a national rail market and national interoperability, and that it was participating in the

⁵⁴ For example, Queensland Rail (sub. 14, p. 6) submitted that the ARA's strategy for national interoperability was the responsibility of an expanded interoperability Working Group including operators, rail infrastructure managers and relevant supply chain members.

⁵⁵ Queensland Rail, sub. 14, pp. 5-6.

⁵⁶ Pacific National, sub. 7, p. 14.

⁵⁷ Aurizon Coal and Bulk, sub. 2, p. 40; Rail Operators Group, sub. 15, p. 14.

⁵⁸ Queensland Rail, sub. 14, pp. 5-8.

initiatives supporting ARA's Rail Freight Strategy 2023-2025.⁵⁹ We recognise that both Queensland Rail and Aurizon Coal and Bulk supported the case for a more harmonised approach to rail access regulation within Australia.⁶⁰ In the event that future harmonisation efforts are hampered by Queensland Rail's access undertaking then it can be amended.

Summary 2.2

It is appropriate to approve Queensland Rail's proposed scope of the 2025 DAU.

The interstate services on the Acacia Ridge - Port of Brisbane dual gauge link are likely not a declared service under the Treasurer's declaration of 2020.

It is not appropriate to amend the scope of the 2025 DAU to require Queensland Rail commit to greater consistency and harmonisation with other rail infrastructure managers.

2.3 Other administrative matters

The 2025 DAU establishes transitional provisions (cl. 6.5), which provide for matters that have begun or have been settled under AU2 to be considered under the AU2 regulatory arrangements or continue to apply once the AU3 regulatory period begins.

Queensland Rail's proposed transitional arrangements in the 2025 DAU reflect those approved as part of its 2020 access undertaking.

These transitional provisions provide for processes and agreements that are completed under the regulatory framework to continue following the expiry of the 2020 access undertaking, giving regulatory certainty to access seekers and access holders. Such certainty promotes the efficient use of Queensland Rail's network and appropriately balances the legitimate business interests of Queensland Rail, access seekers and access holders.

We did not receive any stakeholder submissions in relation to Queensland Rail's proposed transitional arrangements.

The definitions of key terms in Part 7 of the 2025 DAU are, to the extent relevant, discussed in other sections of this draft decision, where those terms relate to individual sections of Queensland Rail's 2025 DAU.

⁵⁹ Queensland Rail, sub. 14, pp. 5-6.

⁶⁰ Aurizon Coal and Bulk, sub. 2, p. 10-11; Queensland Rail, sub. 14, p. 5.

Summary 2.3

It is appropriate to approve Queensland Rail's proposed transitional arrangements (cl. 6.5 of the 2025 DAU).

3 Negotiation framework

Queensland Rail's 2025 DAU provides that the terms of access to the declared infrastructure be negotiated between Queensland Rail and access seekers. The 2025 DAU also establishes a dispute resolution mechanism, which applies to disputes arising under the undertaking or in relation to the negotiation of access terms.⁶¹

Overview of the draft decision

Our draft decision is that it is appropriate to approve Queensland Rail's proposed negotiation framework.⁶²

Clause	QCA draft decision
Part 2	Appropriate to approve. See section 3.2.
cl. 2.9	Appropriate to approve. See section 3.3.
cls. 6.1, 6.2	Appropriate to approve. See section 3.4.
	Part 2 cl. 2.9

Negotiation framework (Part 2 and Part 6) – summary

⁶¹ Disputes about rights or obligations under agreements are to be dealt with in accordance with the provisions in those agreements, rather than the dispute resolution mechanism in the 2025 DAU (2025 DAU, cl. 6.1.2(c)).

⁶² However, we identified minor typological errors in the drafting of cls. 2.9.2 and 2.9.4(d), and a minor referencing error in cl. 2.9.1(c).

⁶³ Queensland Rail, sub. 1, part 2.

⁶⁴ Queensland Rail, sub. 1, cl. 2.9.

⁶⁵ Queensland Rail, sub. 1, cls. 6.1 and 6.2.

3.1 Overview of the negotiation framework

Part 2 of the 2025 DAU sets out the framework for negotiating access rights. It outlines key steps to be undertaken as part of the negotiation process (see Figure 3) and the information parties may be required to provide to as part of this process.⁶⁶





Access rights are granted to the access seeker when parties execute an access agreement (cl. 2.11)

The negotiation framework also outlines principles and procedures for:

- the disclosure and use of confidential information during negotiations⁶⁷
- dealing with multiple access seekers that are seeking access rights relating to the same traffic task⁶⁸
- determining how access applications will be prioritised if there are multiple access applications from access seekers competing for limited available capacity⁶⁹
- the treatment of an access application for renewing existing access rights.⁷⁰

Part 6 of the 2025 DAU outlines a dispute resolution mechanism, which applies a staged approach to resolving disputes (see Figure 4).⁷¹

⁶⁶ These steps include Queensland Rail's obligations to provide preliminary and capacity information (in conjunction with schedule A of the 2025 DAU) and access seekers' obligations to provide certain information in access applications (in conjunction with schedule B of the 2025 DAU).

⁶⁷ 2025 DAU, cl. 2.2.

^{68 2025} DAU, cl. 2.6.

⁶⁹ 2025 DAU, cl. 2.9.

⁷⁰ 2025 DAU, cl. 2.10.

⁷¹ 2025 DAU, cl. 6.1. The SAA contains a separate dispute resolutions mechanism that applies where disputes arise in relation to an access agreement (see chapter 4).

Figure 4: Staged approach to resolving disputes in the 2025 DAU

Representatives of the parties must first meet and use reasonable endeavours to resolve the dispute

If unresolved, the dispute is escalated and senior representatives of the parties are to use reasonable endeavours to resolve it

If the parties are unable to resolve the dispute, either party may refer the dispute to the QCA to make a determination

Source: 2025 DAU, cls. 6.1.3 and 6.1.4.

Part 6 also sets out our obligations and procedures for making a determination in relation to a dispute.⁷² Under the 2025 DAU, Queensland Rail is bound by any determination we make to resolve a dispute.⁷³

3.2 Facilitating the negotiation of access terms

We consider that Queensland Rail's proposed negotiation framework in the 2025 DAU adequately facilitates access seekers and Queensland Rail's negotiation of the terms and conditions for access to Queensland Rail's network.

The negotiation framework provides access seekers with clear guidance on what is required to apply for access, and a level of certainty about the timeframes associated with negotiating access. In this regard, the framework clearly outlines the steps of the negotiation process, the timeframes associated with each of these steps, and the obligations for parties in negotiating access.⁷⁴ For instance, the framework specifies the information that parties are required to provide throughout the negotiation process.

Furthermore, we consider that Queensland Rail's proposed negotiation framework provides for Queensland Rail and access seekers to enter negotiations from an appropriately informed position. The framework establishes the arrangements for parties to exchange information required to assess the access rights sought and, ultimately, negotiate the terms of access.⁷⁵ Amongst other things, the 2025 DAU specifies:

- preliminary and capacity information Queensland Rail is to provide access seekers (as outlined in Schedule A of the 2025 DAU)⁷⁶
- information that access seekers should provide as part of an access application (as outlined in Schedule B of the 2025 DAU)⁷⁷

⁷⁵ Additionally, the 2025 DAU (cl. 2.2) establishes obligations for parties in relation to the treatment of confidential information to support the provision of commercially sensitive information throughout the negotiation process.

⁷² 2025 DAU, cl. 6.1.4 and cl. 6.2.

⁷³ 2025 DAU, cl. 6.1.4(b).

⁷⁴ Aurizon Coal and Bulk considered that the relevant clauses could be presented more sequentially, in line with the negotiation process (see sub. 2, pp. 51-52). We do not consider that rearranging the order in which the relevant clauses are presented is necessary, or beneficial to access seekers.

⁷⁶ 2025 DAU, cl. 2.1.

⁷⁷ 2025 DAU, cl. 7.1 and Sch. B.

- information to be provided by Queensland Rail as part of an indicative access proposal, relating to the relevant operating characteristics, estimated access charges and associated costs and asset values⁷⁸
- Queensland Rail's obligations to provide information that is reasonably required by the access seeker during negotiations, consistent with its obligations for negotiating access under the QCA Act.⁷⁹

Aurizon Coal and Bulk considered that Queensland Rail's standard response timeframes were not compatible with the market demand for these services. Aurizon Coal and Bulk submitted that many of its haulage opportunities required a rapid response, with timeframes for tender preparation measured in weeks.⁸⁰ It considered the following amendments would improve the timeliness of information provided to access seekers:

- no requirement for an access application for timetabled MTP services to include advice on specific train schedules, which would allow an indicative access proposal to be provided based on the assumption that the path will be scheduled using existing available capacity⁸¹
- shorter indicative access proposal response times where it was assumed that the path will be scheduled using existing available capacity, given Queensland Rail would not be required to undertake capacity analysis at this stage of the process.⁸²

Separately, Aurizon Coal and Bulk also considered that the 2025 DAU should be amended to not require access seekers to finalise the operating plan before developing the interface risk management plan.⁸³ Aurizon Coal and Bulk considered that finalising the operating plan while the interface risk management plan was being developed better reflected the iterative process involved.⁸⁴

In response, Queensland Rail considered that it was fair and reasonable to require an access seeker to provide a complete access application to allow Queensland Rail to complete its assessment of available capacity. Queensland Rail submitted that the availability of capacity was a key element of an indicative access proposal and was important in determining if a future access agreement would be feasible and able to be implemented by the access seeker's target start-up date.⁸⁵

We consider that Queensland Rail's proposed negotiation framework provides for parties to be appropriately informed at each stage of the negotiation process. The benefits are that:

- negotiating parties can decide whether to progress to the next stage of negotiations
- the correct information can be used in finalising access terms and policy documents.

We consider that this is in the legitimate interests of Queensland Rail and access seekers.

Nonetheless, we acknowledge the importance of the timely provision of information for access seekers. Importantly, the negotiation framework provides for Queensland Rail to be responsive to the requests and timeframes of specific access seekers, where this is possible. Queensland Rail submitted that an indicative access proposal can be qualified with various assumptions where required, with the intent of those being resolved during the negotiation period should the operator

⁷⁸ 2025 DAU, cl. 2.4.2.

⁷⁹ 2025 DAU, cl. 2.7.2 (a).

⁸⁰ Aurizon Coal and Bulk, sub. 2, p. 34. Furthermore, Aurizon Coal and Bulk considered that uncertainty around the specific scheduled paths to be applied means that information on access pricing is often not available prior to the requirement for it to propose rail haulage commercial terms to customers.

⁸¹ In these circumstances, the specific path would need to be confirmed through the negotiation period.

⁸² Aurizon Coal and Bulk, sub. 2, pp. 4, 34.

⁸³ 2025 DAU, cl. 2.7.2 (a)(iii)(B).

⁸⁴ Aurizon Coal and Bulk, sub. 2, p. 51.

⁸⁵ Queensland Rail, sub. 14, pp. 30-31.

wish to progress on this basis. Queensland Rail submitted that it has provided preliminary information to access seekers on numerous occasions.⁸⁶

The 2025 DAU also provides for Queensland Rail to cease negotiations where certain circumstances exist, which we consider appropriately balances the legitimate interests of Queensland Rail and access seekers.⁸⁷ Queensland Rail may cease negotiations with an access seeker where:

- an access seeker fails to comply with provisions of the undertaking
- an access seeker has concurrent requests for access that Queensland Rail reasonably believes to be duplicate requests for access
- an access seeker does not satisfy prudential requirements
- Queensland Rail considers an access seeker has no reasonable likelihood of using the requested access rights or complying with an access agreement
- Queensland Rail considers the proposed access rights sought may adversely affect the safety of passenger train services.⁸⁸

We note that Queensland Rail's proposed negotiation process is, except for proposed amendments to the queuing mechanism, unchanged from the framework in its 2020 AU.

Queensland Rail submitted that no disputes have been lodged in relation to the negotiation process. This indicates to Queensland Rail that the majority of access seekers are reasonably satisfied with the information and process that Queensland Rail provides.⁸⁹

Supporting negotiations and constraining Queensland Rail's market power

We do not consider that amendments to the negotiation framework in the 2025 DAU are required to further support negotiations between Queensland Rail and access seekers.

Aurizon Coal and Bulk considered that the 2020 access undertaking had not been effective in supporting genuine negotiations for access – with Queensland Rail continuing to exercise market power in its access negotiations for services where no reference tariffs applied.⁹⁰ In support of this, Aurizon Coal and Bulk, Glencore and Centrex submitted that the costs of operating and maintaining the Mount Isa line were comparably higher than other rail corridors for transporting containerised products and have increased significantly in recent years.⁹¹ Moreover, stakeholders submitted there had been a decline in operational performance. For instance, Aurizon Coal and Bulk submitted that its members observed there had been a deterioration in transit times.⁹⁵

⁸⁶ Queensland Rail, sub. 14, pp. 30-31.

⁸⁷ Aurizon Coal and Bulk (sub. 2, p. 52) questioned whether a clause that explicitly provided for Queensland Rail to cease negotiation if it believed the proposed operation might adversely affect a passenger service was necessary. Aurizon Coal and Bulk considered that should Queensland Rail assess the risk of an operation to be above the level it was willing to accept and identified no adequate controls as part of a standard interface risk assessment, the effect would be the same. While it may be argued that the 2025 DAU already provides for Queensland Rail to cease negotiations in these circumstances, we consider that it is reasonable and provides regulatory certainty, if Queensland Rail clearly specifies those circumstances where it may cease negotiations with an access seeker.

⁸⁸ 2025 DAU, cl. 2.8.

⁸⁹ Queensland Rail, sub. 14, p. 29.

⁹⁰ Aurizon Coal and Bulk, sub. 2, pp. 12-13.

⁹¹ Aurizon Coal and Bulk, sub. 2, pp 16–18; Centrex, sub. 10, pp. 3-4; Glencore, sub. 11, pp. 3-5.

⁹² Aurizon Coal and Bulk submitted that in recent years there had been an increase in the distance of the Mount Isa line that was subject to speed restrictions and sectional run times.

⁹³ Aurizon Coal and Bulk submitted that Queensland Rail had restricted access to the corridor for 4 weeks a year in a recent access agreement renewal (reducing the provision of pathing from 52 weeks to 48 weeks per annum).

⁹⁴ Aurizon Coal and Bulk, sub. 2, pp. 18–19.

⁹⁵ Rail Operators Group, sub. 15, p. 4.

Pacific National considered that access charges on the North Coast line were not commensurate with the condition of the lines nor the performance levels required to operate an effective rail freight market.⁹⁶

In response, Queensland Rail considered that the cost comparisons presented by stakeholders did not account for characteristics such as geographical location, traffic density and the types of containerised freight being transported. Furthermore, the cost information did not account for inflation or the effects of changing volumes on the network. In relation to operational performance, Queensland Rail submitted that the largest portion of transit time delays and train cancellations over the past five quarters were due to above-rail rather than below-rail causes. Queensland Rail also submitted that there had been a negligible change in the total distance of temporary speed restrictions over the last five years.⁹⁷

We consider that Queensland Rail has an incentive to attract, and negotiate access with, access seekers that are willing to pay access charges that recover at least the incremental costs incurred by Queensland Rail in providing access to its network. Any revenue generated from a customer above this level will contribute to the recovery of the substantial fixed costs associated with operating and investing in the network.

For those systems that do not have a reference tariff, overall we do not consider that Queensland Rail is able to generate expected revenue for the provision of access that is in excess of the efficient costs it incurs in providing access to access holders.⁹⁸ The Department of Transport and Main Roads provides transport service contract payments to support Queensland Rail's provision of rail infrastructure for the majority of its systems, as they are not financially viable based solely on access revenue.⁹⁹ While access revenues on the Mount Isa line largely cover incremental operating costs, it does not generate sufficient access revenues to cover the total economic cost of providing the service.¹⁰⁰

Where this is the case, we consider that Queensland Rail has an incentive to reduce inefficiencies on the network, given it may not be able to pass on all of the costs it incurs to access holders when negotiating the terms of access. For instance, Queensland Rail submitted that access charges on the Mount Isa line had not increased in real terms for several years.¹⁰¹ Queensland Rail submitted that even where costs had increased (for instance, due to rising materials prices and labour shortages) there had been no transfer of cost increases to users on the Mount Isa line.¹⁰²

Moreover, the 2025 DAU provides for access seekers to refer a dispute that arises in negotiating the terms of access to us to make a determination. As outlined in section 3.4, Queensland Rail's

¹⁰¹ Queensland Rail submitted that standard access charges for bulk mineral concentrates and intermodal freight on the Mount Isa line had only increased by CPI since FY18, except for in July 2019, where intermodal access charges were reduced by 5.2% in real terms.

⁹⁶ Pacific National, sub. 13, p. 5.

⁹⁷ Queensland Rail, sub. 14, pp. 19-20, 23.

⁹⁸ Including a return on investment commensurate with the regulatory and commercial risks involved.

⁹⁹ Queensland Rail, sub. 14, p. 16. In 2022-23, Queensland Rail reported \$190 million in total access revenue, against operating expenses of over \$409 million for its regional network.

¹⁰⁰ Queensland Rail, sub. 14, pp. 17-18. The return on assets is calculated on an assets value of up to \$331 million (as at 30 June 2023) as compared to an economic value of more than \$1.4 billion as estimated by a DORC valuation methodology conducted by Queensland Rail. While we have not conducted a valuation of the Mount Isa line, we note it is approximately three times as long as the West Moreton system, which has a regulatory asset base of more than \$500 million (see section 9.3.1 of this draft decision). The Mount Isa line is rated at a 20-tonne axle load, while the West Moreton axle load is 15.75 tonnes.

¹⁰² Queensland Rail, sub. 14, pp. 20, 22.

proposed dispute resolution mechanism provides for disputes to be resolved in a fair and timely manner.¹⁰³

Centrex and North West Phosphate considered that Queensland Rail should be required to publish floor and price ceiling prices to improve transparency and aid the price differentiation process.¹⁰⁴ In this regard, North West Phosphate submitted that junior miners currently suffered from information asymmetry and were thus at a distinct resource disadvantage in contract and price negotiations. Glencore also considered that Queensland Rail should be obliged to publish details of how it had calculated the efficient incremental cost of providing their service, as well as the ceiling price.¹⁰⁵

We consider that Queensland Rail's proposed negotiation framework provides for Queensland Rail and access seekers to enter negotiations from an appropriately informed position. As part of an indicative access proposal, Queensland Rail is to outline:

- the cost of providing the access, including the capital, operating and maintenance costs (consistent with s. 101(2)(b) of the QCA Act)
- asset value including the valuation methodology (consistent with s. 101(2)(c) of the QCA Act).¹⁰⁶

Summary 3.1

It is appropriate to approve Queensland Rail's proposed negotiation process (Part 2 of the 2025 DAU).

3.3 Arrangements for prioritising access applications

Queensland Rail's proposed negotiation process includes the following arrangements for prioritising the negotiation of access applications:

- a queuing mechanism that establishes how access applications are to be prioritised where it is not possible for Queensland Rail to fulfil requests for access rights from multiple access seekers¹⁰⁷
- provisions that prioritise access holders looking to renew access rights over new access seekers, when negotiating access.

¹⁰³ Glencore (sub. 11, pp. 3-5) submitted that there was currently no opportunity to challenge Queensland Rail's assumed WACC for non-reference services. Glencore also considered that Queensland Rail had sought to recover costs for capital projects where either Glencore had doubted the efficiency of; or which had never realised the asserted efficiency benefits of; or had not actually been developed. We consider that access seekers are able to challenge such matters as part of negotiating the level of access charges and have the ability to refer a dispute to us to make a determination where such matters cannot be resolved.

¹⁰⁴ Centrex, sub. 10, p. 4; North West Phosphate, sub. 6, p. 2. Mount Isa Line Users (sub. 17, p. 2) supported Centrex's and North West Phosphate's submissions on these matters.

¹⁰⁵ Glencore, sub. 11, pp. 4-5. Glencore submitted that past attempts to use the legislative rights to information under the QCA Act have failed to produce any material additional transparency.

¹⁰⁶ Where a reference tariff does not apply (2025 DAU, cl. 2.4.2 (e)).

¹⁰⁷ 2025 DAU, cl. 2.9.

Queensland Rail proposed changes to the queuing mechanism provisions that it considered simplified the drafting but retained the principles outlined in the 2020 access undertaking.¹⁰⁸

We consider that Queensland Rail's proposed queuing mechanism provides a consistent, nondiscriminatory and transparent framework for prioritising access applications where capacity constraints exist.

The principles and procedures for establishing a queue remain consistent with those outlined in Queensland Rail's 2020 access undertaking. Queensland Rail is to:

- identify whether an access application is to be placed in a queue
- notify the access seeker that its application is a queued access application and identify its position in the queue
- establish the order of the queue based on the date that Queensland Rail receives the applications.¹⁰⁹

The 2025 DAU provides for Queensland Rail to change the order of the queue in certain circumstances.¹¹⁰

The 2025 DAU queuing mechanism does not apply to access holders who are seeking to renew access rights that are about to expire.¹¹¹ Instead, access holders who are renewing access rights will be given priority over a new access seeker in negotiating an access agreement, where they are competing for the same access rights.¹¹²

Aurizon Coal and Bulk and Pacific National considered that the proposed queuing mechanism may affect the ability of existing access holders to renegotiate contracted paths in order to maintain the continuity of their business. Aurizon Coal and Bulk submitted that the queuing mechanism may give an access seeker priority over an access holder looking to renew its access entitlement, simply because of the order that the applications are received.¹¹³

To address this, Aurizon Coal and Bulk considered that the 2025 DAU should also provide for Queensland Rail to place an access holder at the start of the queue, where an access holder provides notification that it wishes to renew its access entitlement within 120 days of its access agreement expiring.¹¹⁴ Aurizon Coal and Bulk considered that this approach:

- provides for Queensland Rail to continue to have the right to re-order the queue
- provides additional confidence to access holders around their ability to negotiate for renewal of their access rights
- does not detract from Queensland Rail's legitimate business interests in renegotiating access charges in accordance with an approved access undertaking, or in preferencing allocation of the capacity to a different access seeker who places a higher value on the capacity.¹¹⁵

¹⁰⁸ Queensland Rail, sub. 1, p. 60. Queensland Rail proposed simplified drafting amendments to make clear that where there are multiple applications for the same traffic task, those applications will be regarded as a single access application for the purpose of the queue (see 2025 DAU, cl. 2.9.2). Queensland Rail is to assign the same position in any applicable queue where the customer has not yet nominated an alternate access seeker.

¹⁰⁹ 2025 DAU, cl. 2.9.3 (a), (b).

¹¹⁰ 2025 DAU, cl. 2.9.4(b). These circumstances include where the parties end the negotiation of access rights, in accordance with the 2025 DAU; the change is required for compliance with the 2025 DAU; Queensland Rail prioritises an application from a coal-carrying train services (using West Moreton) for an access agreement of at least 10 years ahead of an application seeking a term less than 10 years; Queensland Rail prioritises an application from other train services based on the present value of contribution to common costs.

¹¹¹ 2025 DAU, cl. 2.9.1(c).

¹¹² 2025 DAU, cl. 2.10.

¹¹³ Aurizon Coal and Bulk, sub. 2, p. 35; Pacific National, sub. 13, p. 7.

¹¹⁴ Pacific National (sub. 13, p. 7) supported Aurizon Coal and Bulk's submission.

¹¹⁵ Aurizon Coal and Bulk, sub. 2, pp. 4, 35-36.

Queensland Rail's 2025 Draft Access Undertaking

We consider that the renewal provisions outlined in the 2025 DAU provide sufficient certainty, by applying a non-discriminatory, transparent and balanced approach for users to manage contractual positions over time. As such, we consider that Queensland Rail's proposed renewal arrangements promote the efficient use of the network (s. 138(2)(a)).

However, we encourage Queensland Rail and other stakeholders to further consider and, if possible, reach consensus on amendments to the renewal provisions that provide for an access holder to be placed at the start of the queue, if that access holder provides notification that it wishes to renew its access agreement within 120 days of it expiring. We will have regard to consensus positions reached amongst stakeholders as part of our assessment of Queensland Rail's 2025 DAU.

New Hope and Yancoal considered that existing users in the West Moreton system should have stronger renewal rights than those proposed in the 2025 DAU.¹¹⁶ We discuss this matter in section 8.5.4 as part of our consideration of an appropriate reference tariff for the West Moreton system.

Summary 3.2

It is appropriate to approve Queensland Rail's proposed arrangements for prioritising access applications (clause 2.9 of the 2025 DAU).

3.4 Dispute resolution mechanism

While the 2025 DAU facilitates negotiation between access seekers and Queensland Rail, disputes between the parties may still arise. An effective dispute resolution mechanism is an important element of the negotiation framework.

We consider that Queensland Rail's proposed dispute resolution mechanism in the 2025 DAU promotes successful negotiations and makes parties accountable for their conduct under the access undertaking.

The mechanism outlines procedures for a dispute to be raised and then resolved between the parties. Where a dispute remains unresolved, the relevant parties may refer that dispute to us for a determination, providing for the dispute to be resolved in a timely manner.¹¹⁷ When disputes are resolved in a fair and timely way, parties can be confident that negotiations will proceed in a meaningful manner in accordance with the intent, obligations and processes of the undertaking.

Queensland Rail's proposed dispute resolution mechanism provides certainty as to the binding nature of our determinations, including orders as to the payment of costs.¹¹⁸ This will increase stakeholder confidence that disputes will be resolved in a fair and timely manner. This is in the interests of all potential disputing parties and will promote the efficient use of the declared service (ss. 138(2)(a), (b), (e) and (h)).

We note that Queensland Rail's proposed dispute resolution mechanism is unchanged from the mechanism contained in its 2020 AU.

¹¹⁶ New Hope, sub. 5, p. 26; Yancoal sub. 9, p. 21.

¹¹⁷ A dispute may also be referred directly to us if a party fails to comply with the requirements to use reasonable endeavours to resolve the dispute according to the resolution by escalation procedures in cl. 6.1.3.

¹¹⁸ The 2025 DAU requires that before a QCA determination commences, all parties agree in a legally binding way to be bound by the determination.

Queensland Rail's 2025 Draft Access Undertaking

Pacific National considered that the following amendments to Queensland Rail's proposed dispute resolution mechanism would facilitate a more efficient and less costly dispute resolution process:

- a reduction in the timeframes for dispute escalation
- the inclusion of a mediation or conciliation step prior to a dispute being referred to us for a determination
- setting an expectation around the timing around our determination on a dispute, such as outlining a timeframe for making a determination.¹¹⁹

We consider Queensland Rail's proposed timeframes applying to each stage of the dispute resolution process¹²⁰ provide parties with sufficient opportunity to attempt to resolve a dispute in a timely manner, before being able to refer that dispute to us for a determination. A reduction in these timeframes may limit opportunities for parties to resolve the dispute before it is escalated or referred to us for a determination.

Queensland Rail's proposed dispute resolution mechanism provides sufficient opportunity for parties to resolve a dispute, with parties to use reasonable endeavours to resolve a dispute as it is escalated to senior representatives. The ability to then refer that dispute to us for a determination provides further incentive for parties to resolve a dispute.

Within the dispute resolution process, it is always open for the parties to voluntarily engage in mediation. However, requiring that an additional step of mediation be included in the dispute resolution mechanism may simply increase the costs and timeframes associated with resolving a dispute.

We also do not consider it appropriate to restrict our decision-making process to resolve a dispute. While we seek to resolve disputes in a timely manner, restricting our decision-making process may have implications for our ability to effectively resolve a dispute – which is not in the interests of the relevant parties.

Summary 3.3

It is appropriate to approve Queensland Rail's proposed dispute resolution mechanism (cls. 6.1 and 6.2 of the 2025 DAU).

¹¹⁹ Pacific National, sub. 7, pp. 19-21.

¹²⁰ In accordance with cl. 6.1.3 of the 2025 DAU, following a dispute notice, the relevant parties have 5 days to meet and use reasonable endeavours to resolve the dispute before the matter is escalated to senior representatives of the parties. The senior representatives then have 10 days to use reasonable endeavours to resolve the dispute before the matter is escalated to each party's chief executive officer. If the matter is not resolved within 10 days from the date it is referred to each party's chief executive officer, the parties may refer that dispute to us for a determination.

4 Standard access agreement (Schedule H)

An access agreement must be consistent with the terms of the SAA, unless the parties agree otherwise.¹²¹ Queensland Rail's proposed SAA is schedule H of the 2025 DAU (the proposed SAA). The proposed SAA sets out the standard terms and conditions for access to Queensland Rail's network.¹²²

Overview of the draft decision

Our view is that Queensland Rail's SAA should maximise efficiency in use and operation of the network.¹²³ It should also allocate risks to the party best able to manage those risks.

Our draft decision is that it is appropriate for Queensland Rail to make some amendments to the proposed SAA to provide a more balanced allocation of risks. We also encourage further engagement between the parties and invite submissions, preferably on a consensus basis, including proposed drafting, on a number of issues, as outlined below. On all these issues, we consider that it is in the interests of all parties that the proposed SAA is clearly drafted and workable (ss. 138(2)(b), (d), (e) and (h)).

Proposal	Clause	QCA draft decision
Train path optimisation		
Train path rescheduling: Allow Queensland Rail to reschedule train paths where an operator consistently has poor reliability performance. ¹²⁴	n/a	It may be appropriate to amend the treatment of train path rescheduling. See section 4.2.1.
Train service levels: Specify train service levels in a way that allows flexibility in train scheduling. ¹²⁵	Sch. 2, att. 1	It may be appropriate to amend the treatment of train service level requirements. See section 4.2.2.
Ad hoc train service: Specify timeframes for Queensland Rail to respond to access holders' requests for ad hoc train services. ¹²⁶	cl. 8.3	It is appropriate to specify timeframes for responding to ad hoc train service requests. See section 4.2.3.
Path resumption: Amendments involving resumption utilisation threshold, resumption trigger, access holder entitlements and ability to resume or reschedule paths. ¹²⁷	cl. 21.1	It may be appropriate to amend the path resumption requirements. See section 4.2.4.

Standard access agreement (Schedule H) – summary

¹²⁴ Aurizon Coal and Bulk, sub. 2, pp. 47-48.

¹²¹ 2025 DAU, cl. 2.11.2(a).

¹²² References to clauses and schedules in this chapter are to the proposed SAA in Schedule H of the 2025 DAU, unless otherwise specified.

¹²³ These goals are consistent with s. 138(2)(a) and the object of Part 5 of the QCA Act under s. 69E of the QCA Act.

¹²⁵ Aurizon Coal and Bulk, sub. 2, pp. 47-48.

¹²⁶ Pacific National, sub. 7, pp. 23-24.

¹²⁷ Aurizon Coal and Bulk, sub. 2, pp. 48-50.

Proposal	Clause	QCA draft decision	
Insurance			
Limitation on claims: Remove requirement for claimants' provision of 'full details of the Claim'. ¹²⁸	cl. 13.2	It is appropriate to specify that details are required only 'to the extent known'. See section 4.3.1.	
Operator's obligation to obtain and	maintain in	surance	
Responsibility for coverage: Change responsibility for coverage of operator's associates, agents, consultants, contractors and their subcontractors. ¹²⁹	cl. 16.1	It may be appropriate to amend the coverage requirements under cl. 16.1. See section 4.3.2.	
Coverage and deductible levels: Amend coverage and deductible levels. ¹³⁰	cl. 16.1	It may be appropriate to amend certain coverage and deductible levels of insurances held under cl. 16.3. See section 4.3.2.	
Insurer			
Financial strength rating: Change insurer's minimum financial strength rating to 'A-'. ¹³¹	cl. 16.3	It may be appropriate to amend the minimum strength rating requirement in cl. 16.3. See section 4.3.3.	
Captive insurance companies: Recognise wholly owned captive insurance companies as suitable insurers. ¹³²	cl. 16.3	It is not appropriate to amend the captive insurance provisions as proposed. See section 4.3.3.	
Essential terms and conditions: Remove disallowance of exclusions, endorsements or alterations that adversely amend coverage without Queensland Rail's written consent. ¹³³	cl. 16.4(b)	It is not appropriate to amend the essential terms and conditions as proposed. See section 4.3.4.	
Insurance claims: Change requirements regarding disclosure of occurrence, details and progress of other insurance claims. ¹³⁴	cl. 16.9	It may be appropriate to amend the notification requirements to allow certain exceptions. See section 4.3.5.	
Dispute resolution escalation			
Escalation process: Adopt escalation timeframes and processes outlined in the dispute and complaint resolution process (cl. 6.1 of the 2025 DAU). ¹³⁵	cl. 19.2	It may be appropriate to amend the dispute resolution process under the proposed SAA. See section 4.4.	

.

¹²⁸ Pacific National, sub. 7, p. 24.
¹²⁹ Queensland Rail, sub. 1, p. 66.
¹³⁰ Aurizon Coal and Bulk, sub. 2, p. 52; Pacific National, sub. 7, p. 24-25.
¹³¹ Pacific National, sub. 7, pp. 25-26.
¹³² Aurizon Coal and Bulk, sub. 2, p. 52.
¹³³ Pacific National, sub. 7, pp. 25-26.
¹³⁴ Pacific National, sub. 7, p. 26.
¹³⁵ Pacific National, sub. 7, p. 26.

Proposal	Clause	QCA draft decision	
Assignment of Queensland Rail's rights under SAA			
Assignment: Allow assignments by Queensland Rail where it ceases or no longer expects to have a right to operate all or part of the network. ¹³⁶	cl. 22.1(a)	It is not appropriate to approve the proposal in its current form. See section 4.5.	
Performance indicators			
New performance indicators: Change performance indicators to others associated with aggregate system performance reporting. ¹³⁷	Sch. 5, cl. 1	It may be appropriate to amend the performance indicators. See section 4.6.	

4.1 Consistency

Queensland Rail has not proposed to significantly change the existing SAA under AU2, other than changes to the scope of operators' insurance obligations and the circumstances in which Queensland Rail may assign its rights to another party. Stakeholders have also proposed changes to other areas involving train path optimisation, insurance requirements, dispute resolution escalation and performance indicators.

In assessing these issues, having regard to all the relevant factors (as outlined in chapter 1), we consider that it is appropriate to retain consistency between the existing SAA under AU2 and the proposed SAA as far as possible, subject to material changes in circumstances. In particular, in our view, it is appropriate to place weight on whether:

- the proposed SAA appropriately allocates existing responsibilities and obligations under the new arrangements between the rail operator and the end user
- the risk that each party bears remains unchanged or, if the risk profile does change, it is justifiable and appropriate.

Having regard to the criteria in section 138(2) of the QCA Act, we consider this approach to be appropriate because:

- we have previously accepted the existing SAA under AU2 as being appropriate
- it is in all parties' interests to have a substantial degree of consistency and continuity in the rights and obligations under the various forms of access arrangements, especially as Queensland Rail is currently a party to other access agreements based on the existing SAA
- Queensland Rail and users are each best placed to identify relevant material changes in their respective circumstances, including risk profiles, that may merit divergence from the existing SAA.

4.2 Train path optimisation

Stakeholders proposed amendments to improve train path optimisation in the following areas:

- train path rescheduling
- train service levels

¹³⁶ Queensland Rail, sub. 1, pp. 66-67. ¹³⁷ Aurizon Coal and Bulk, sub. 2, pp. 44-46.
- ad hoc train service
- path resumption.

These proposed amendments are considered in turn in this section.

4.2.1 Train path rescheduling

Aurizon Coal and Bulk proposed that the proposed SAA be amended to allow Queensland Rail to reschedule train paths where an operator had consistent poor reliability performance and to require an operator to use its best endeavours to negotiate variations to agreements defining network entry and exit times to accommodate the varied schedule.¹³⁸ It noted that, unlike Queensland Rail's proposed SAA, most Australian access frameworks included a limited entitlement to reschedule services in circumstances where an operator or access holder had consistent poor reliability performance, which could help enable schedule optimisation through the introduction of a new efficient scheduled path.

Aurizon Coal and Bulk also proposed requiring operators to use their best endeavours to negotiate a varied entry or exit time that aligned with the revised schedule offered by Queensland Rail (and more closely resembled actual network usage), saying this would also support schedule optimisation.

Our view is that Aurizon Coal and Bulk's proposal may support schedule optimisation and network utilisation by enabling Queensland Rail to reschedule train paths in response to consistent poor reliability performance of an operator or access holder. The requirement for operators to negotiate network entry or exit times that reflect actual practice also provides greater flexibility for the parties and supports Queensland Rail's ability to optimise the schedule.

Our view is that it may be appropriate to approve Aurizon Coal and Bulk's proposal because schedule optimisation is consistent with the objective of economically efficient operation under Part 5 of the QCA Act (s. 138(2)(a)) and is in the interests of Queensland Rail, rail operators and access holders, and in the public interest (ss. 138(2)(b), (d), (e) and (h)). Schedule optimisation also may incentivise the transfer of freight from road transport to rail transport, with consequent environmental and public interest benefits through the reduction of vehicle emissions associated with climate change (ss. 138(2)(d) and (h)).

We invite stakeholders' submissions, preferably on a consensus basis, on Aurizon Coal and Bulk's proposal regarding train path rescheduling, including proposed drafting. With respect to the parties' ability to provide joint submissions, we note that the Australian Competition and Consumer Commission (ACCC) on 1 February 2024 granted authorisation with conditions for rail operators to collectively negotiate with Queensland Rail and other rail network owners regarding the terms and conditions on which the operators acquire below-rail access to the owners' respective below-rail networks.¹³⁹

4.2.2 Train service levels

Aurizon Coal and Bulk proposed that train service levels in Schedule 2, Attachment 1 be specified in a way that allows some flexibility for train scheduling in accordance with the train service level, rather than including fixed network entry or exit times.¹⁴⁰ It submitted that this proposed

¹³⁸ Aurizon Coal and Bulk, sub. 2, pp. 47-48.

 ¹³⁹ ACCC, <u>Application for revocation of AA1000425 and the substitution of authorisation AA1000644</u>, determination,
 1 February 2024.

¹⁴⁰ Aurizon Coal and Bulk, sub. 2, pp. 47-48.

amendment would provide Queensland Rail with additional opportunities to modify existing schedules to facilitate the inclusion of new train services, or to otherwise optimise the train schedule.

We consider that providing a higher degree of flexibility for train scheduling would support schedule optimisation, improve network utilisation and be beneficial to all parties (ss. 138(2)(b), (d) and (e)). As noted in section 4.2.1, schedule optimisation is also consistent with the objective of economically efficient operation under Part 5 of the QCA Act, and it may incentivise the transfer of freight from road transport to rail transport, with consequent environmental and public interest benefits (ss. 138(2)(a) and (h)). Our view is that it may be appropriate to approve the proposed amendment, and we invite submissions from Queensland Rail and stakeholders on proposed drafting for Schedule 2, Attachment 1.

4.2.3 Ad hoc train service

Pacific National proposed adding a provision to the proposed SAA that specified timeframes for Queensland Rail within which to respond to an access holder's request for ad hoc train services. It proposed that Queensland Rail should respond either within 48 hours or 7 days (non-business days inclusive), depending on whether the request was for ad hoc service within 2 weeks or exceeding 2 weeks from the request date. Pacific National submitted that these timeframes would provide more clarity and certainty and mitigate concerns with variability of Queensland Rail's response times.¹⁴¹

In our view, it is in the interests of all parties to enhance timeliness, predictability and certainty in pathing request outcomes (ss. 138(2)(b), (d), (e) and (h)). These improvements in ad hoc train service delivery may also incentivise the transfer of freight from road transport to rail transport, with consequent environmental and public interest benefits (ss. 138(2)(a) and (h)). We acknowledge Pacific National's suggestion that the proposed timeframes should be based on calendar days, and not business days.

However, we consider that business days remains an appropriate metric to avoid onerously short response times, such as where a request is received by Queensland Rail on a Friday, and a response is due by Sunday, contrary to the legitimate business interest of Queensland Rail (s. 138(2)(b)). The use of business days is also consistent with other timeframes in Queensland Rail's DAU.

It is appropriate to amend clause 8.3(a) to add the proposed timeframes (in business days) for Queensland Rail's response to an access holder's request for ad hoc train services.

4.2.4 Path resumption

Aurizon Coal and Bulk proposed that the path resumption provisions in clause 21.1 be amended to:

- modify the resumption utilisation threshold to be either at least 50% utilisation over 3 months or at least 75% utilisation over 6 months, with a path only measured as 'utilised' where it is used for a train service of at least 50% of its usual length or weight
- add a new resumption trigger, namely the loss of a connecting path on an adjoining network, except where the operator has continued to operate a modified train service not reliant on that connecting path
- remove the provisions that enable an access holder to contest the resumption once the underutilisation trigger has been met

¹⁴¹ Pacific National, sub. 7, pp. 23-24.

Queensland Rail's 2025 Draft Access Undertaking

• allow Queensland Rail the option of either resuming a path or rescheduling it to the nearest otherwise available time.

Aurizon Coal and Bulk submitted that these amendments would improve Queensland Rail's ability to effectively address path hoarding, if this occurs. In Aurizon Coal and Bulk's view:

- the existing provisions provide a comparatively low utilisation threshold of 50%
- the proposed resumption trigger would allow a more efficient path resumption process where services operate over multiple networks
- path resumption decisions should be based only on a factual utilisation test and that the existing allowance for path resumption disputes by access holders is no longer necessary, and creates uncertainty and the potential for delay
- enabling Queensland Rail, where the underutilisation threshold is triggered, to either resume or reschedule a path to the nearest available time mitigates the risk that it may be disincentivised to use its path resumption powers in certain instances.¹⁴²

We consider that it may be appropriate to approve Aurizon Coal and Bulk's proposal because strengthening Queensland Rail's ability to address network underutilisation is consistent with the object of economically efficient operation under Part 5 of the QCA Act (s. 138(2)(a)) and is in the interests of Queensland Rail, rail operators, access holders and in the public interest (ss. 138(2)(b), (d), (e) and (h)).

We invite submissions on this issue, including proposed drafting, preferably on a consensus basis.

Summary 4.1

- 1. **Train path rescheduling:** it may be appropriate to approve the proposed amendment.
- 2. **Train service levels:** it may be appropriate to approve the proposed amendment.
- 3. **Ad hoc train service:** it is appropriate to amend clause 8.3(a) to add the proposed timeframes (in business days) for Queensland Rail's response to an access holder's request for ad hoc train services.
- 4. **Path resumption:** it may be appropriate to approve the proposed amendment.

We invite submissions, including proposed drafting, from stakeholders on points 1, 2 and 4 and would be minded to approve proposals agreed by Queensland Rail and its customers.

4.3 Insurance

Queensland Rail and stakeholders proposed amendments to insurance requirements in the SAA involving the following areas that we consider in turn in this section:

limitation on claims

¹⁴² Aurizon Coal and Bulk, sub. 2, pp. 48-50.

- operator's obligation to obtain and maintain insurance
- insurer
- essential terms and conditions of insurance
- insurance claims.

4.3.1 Limitation on claims

Pacific National proposed removing a requirement in clause 13.2(a) that prevented claims between parties to the SAA unless 'full details of the Claim' were provided to the other party within one year after the occurrence from which the claim arose. Pacific National submitted that the term 'full details' was vague and could be interpreted as including quantum or a final amount, which may take longer than one year to determine.¹⁴³ 'Claim' is a defined term under clause 28.1 and essentially involves any claim, cause of action, proceeding, liability, suit or demand.

Clause 13.2(a) bars claims between parties unless notice and full details of the claim are provided within the prescribed one-year period. In our view, timely provision of these elements is consistent with ensuring that contentious matters are expeditiously brought forward for formal resolution. Expeditious resolution of claims is in the interests of all parties (ss. 138(2)(b), (d) and (e)). Accordingly, we consider it is not appropriate to remove the requirement for provision of full details of claims.

However, Pacific National's concern – that the term 'full details' could be construed as requiring indication of a quantum or final amount that may not be determinable within one year from the occurrence – can be addressed by limiting the details required, such as by adding the term 'to the extent known' to clause 13.2(a). The clause then reads 'notice and full details, to the extent known, of the Claim have been given to the other Party...'

We consider that our proposed amendment appropriately meets the parties' and public interests in promoting timely awareness and improving the resolution of claims and the clarity and workability of access agreements by omitting vague or impracticable requirements (ss. 138(2)(b), (d) and (e)).

4.3.2 Operator's obligation to obtain and maintain insurance

Coverage of operator's associates, agents and other parties

Queensland Rail proposed changing clause 16.1 of the existing SAA in AU2 to replace the requirement that an operator's public liability policy cover its associates, agents, consultants, contractors and their subcontractors with a requirement that the operator must ensure that those parties take out their own insurance.¹⁴⁴ Stakeholders generally did not oppose this proposal in principle.¹⁴⁵ However, several stakeholders expressed concern that the proposal, as currently drafted, was unduly burdensome and unfeasible,¹⁴⁶ uneconomical and inefficient for operators to fulfill,¹⁴⁷ vague in regard to certain terminology,¹⁴⁸ and excessive in scope.¹⁴⁹

¹⁴³ Pacific National, sub. 7, p. 24.

¹⁴⁴ Queensland Rail, sub. 1, p. 66.

¹⁴⁵ Aurizon Coal and Bulk, sub. 2, p. 52; New Hope Group, sub. 5, pp. 29-30, Yancoal, sub. 9, p. 30.

¹⁴⁶ Pacific National, sub. 7, pp. 24-25.

¹⁴⁷ New Hope Group, sub. 5, pp. 29-30; Yancoal, sub. 9, p. 30

¹⁴⁸ Pacific National, sub. 7, pp. 24-25 (regarding the term 'ensure'); Yancoal, sub. 9, p. 30 (regarding the term 'sufficient insurance').

¹⁴⁹ New Hope Group, sub. 5, pp. 29-30; Yancoal, sub. 9, p. 30. New Hope and Yancoal each submitted that the proposed amendments appear to extend beyond the stated rationale and include additional types of insurance. Both parties suggested revisions to the proposed amendment to address these concerns.

We consider that it is in all parties' interests, including the public interest, that public liability insurance policies adequately cover an operator's associates, agents, consultants, contractors and their subcontractors. Given the scale and scope of Queensland Rail's operations, this helps to avoid critical gaps in coverage against public liability injury or damage claims (ss. 138(2)(b), (d), (e) and (h)). We also consider that appropriate allocation of responsibility for maintaining this coverage is consistent with the objective of economically efficient operation under Part 5 of the QCA Act (s. 138(2)(a)) and in the interests of Queensland Rail, rail operators, access holders and in the public interest (ss. 138(2)(b), (d), (e) and (h)).

We acknowledge the concerns expressed by stakeholders regarding the burden, workability, affordability, efficiency, clarity and scope of Queensland Rail's proposal. Given these concerns, our preliminary view is that it may not be appropriate to approve Queensland Rail's proposal as drafted. It may be appropriate for Queensland Rail to amend the proposed drafting to address the concerns raised consistent with the parties' interests.

We therefore invite submissions, preferably on a consensus basis, including proposed drafting, from stakeholders in this regard.

Coverage and deductible levels

Stakeholders proposed various amendments to the coverage and deductible levels of insurances held under clause 16.3, including the limit of liability in clause16.1(a)(iv)(B);¹⁵⁰ the scope of coverage described in clause 16.1(a)(iv)(C); the maximum deductible for operators' public liability insurance required under clause 16.1(a)(iv)(D);¹⁵¹ the requirement for operators to hold carriers' liability insurance under clause 16.1(a)(v);¹⁵² the maximum deductible in operators' carriers' liability policy of insurance under clause 16.1(a)(v)(C);¹⁵³ and removing references to 'without limitation' in clause 16.1(a).¹⁵⁴

We consider that it is in all parties' interests, including the public interest, that public liability insurance policies adequately cover an operator's associates, agents, consultants, contractors and their subcontractors. Given the scale and scope of Queensland Rail's operations, this helps avoid critical gaps in coverage against public liability injury or damage claims (ss. 138(2)(b), (d), (e) and (h)). We also consider that requirements for appropriate insurance limits and deductibles are consistent with the objective of economically efficient operation under Part 5 of the QCA Act (s. 138(2)(a)) and in the interests of Queensland Rail, rail operators, access holders and in the public interest (ss. 138(2)(b), (d), (e) and (h)).

It may be appropriate to amend certain coverage and deductible levels of insurances held under clause 16.3. We invite submissions, preferably on a consensus basis, including proposed drafting, from stakeholders in this regard (ss. 138(2)(b), (d), (e) and (h)).

¹⁵⁰ Aurizon Coal and Bulk, sub. 2, p. 52.

¹⁵¹ Aurizon Coal and Bulk, sub. 2, p. 52; Pacific National, sub. 7, pp. 24-25.

¹⁵² Pacific National, sub. 7, pp. 24-25.

¹⁵³ Aurizon Coal and Bulk, sub. 2, p. 52.

¹⁵⁴ Aurizon Coal and Bulk, sub. 2, p. 52.

4.3.3 Insurer

Minimum financial strength rating

Pacific National proposed that the minimum Standard & Poor's financial strength rating held by access holders' and operators' insurers under clause 16.3 should be reduced from an A to an A–. Pacific National noted that this change would reflect common industry practice.¹⁵⁵

We consider it is important that insurers' minimum strength rating is commensurate with the parties' needs and generally reflects industry standards. It may be appropriate to amend the minimum strength rating in clause 16.3, and we invite submissions from stakeholders, including Queensland Rail in this regard, preferably on a consensus basis.

Captive insurance companies

Aurizon Coal and Bulk submitted that Queensland Rail should recognise the use of captive insurance companies by operators in the coal industry and proposed that clause 16.3 should be amended to include 'any insurance policy required to be effected and maintained by the Access Holder and the operator pursuant to clause 16 may at any time, be placed in whole or in part with a wholly owned captive insurance company.'¹⁵⁶

Captive insurance companies are insurers that provide insurance for and are owned by a parent company. Aurizon Coal and Bulk did not justify the need or identify the drivers behind its suggestion that the use of captive insurance companies should be recognised by Queensland Rail. Clause 16.3 protects the interests of access holders and the operator by requiring insurers based on a prescribed, independently determined financial strength rating. This standard aligns with the important objective that insurers are financially capable of covering claims and that liabilities are not inappropriately externalised. Allowing captive insurance companies to provide insurance without appropriate controls does not provide for these important safeguards and is not in the interests of access holders.

We consider that it is not appropriate to make the proposed amendment, having regard to the interests of access holders, operators, and the public (ss. 138(2)(b), (d) and (e)). However, we encourage the parties to consider this issue further and to propose drafting on reasonably acceptable controls that may be appropriate.

4.3.4 Essential terms and conditions of insurance

Pacific National proposed deleting the requirement under clause 16.4(b) that access holders and the operator must ensure that their insurances do not contain any exclusions, endorsements or alterations that adversely amend the cover provided, without the written consent of Queensland Rail.¹⁵⁷ Pacific National submitted that all insurance policies contained these types of retractions of cover, not all of which would impact relevant activities. As a result, it was unreasonable to expect rail operators to seek Queensland Rail's approval each time an insurance policy was renewed or amended.

In our view, the requirement under clause 16.4(b) to obtain Queensland Rail's written consent for insurance terms that adversely amend the cover provided supports the objective that acceptable insurance coverage remains in place, which is in the interests of all parties (ss. 138(2)(b), (d) and (e)).

¹⁵⁵ Pacific National, sub. 7, pp. 25-26.

¹⁵⁶ Aurizon Coal and Bulk, sub. 2, p. 52.

¹⁵⁷ Pacific National, sub. 7, pp. 25-26.

It is also not evident what types of irrelevant activities may be covered by this provision, as submitted by Pacific National. We also note that clause 16.4(b) requires that Queensland Rail's written consent not be unreasonably withheld or delayed. This requirement helps promote predictability and expediency in administering the provision.

Having regard to the factors in section 138(2), we consider it is not appropriate to amend clause 16.4(2) as proposed by Pacific National.

4.3.5 Insurance claims

Clause 16.9 of the proposed SAA requires that parties notify Queensland Rail of reasonable details of insurance claims related to the SAA and keep Queensland Rail informed of developments. Pacific National proposed that this should be replaced with a requirement to notify Queensland Rail of the existence of claims likely to vitiate required insurance, except in certain circumstances. These exceptional circumstances involved situations where notification could prejudice the insurance outcome, breach confidentiality obligations or circulate commercially sensitive information.¹⁵⁸ Pacific National submitted that the current requirements could prejudice the outcome of claims by Queensland Rail and that notification of claims unrelated to the SAA was unreasonable, except where the claim would vitiate required insurance.

As defined in clause 28.1, 'Insurance' means the insurances to be effected and maintained in accordance with clause 16. Clauses 16.1 and 16.2 set out the operator and access holder's respective obligations to obtain and maintain insurance in accordance with their activities, works, obligations and responsibilities under the SAA. As claims under the parties' insurance involves matters under the SAA, Queensland Rail presumably has a direct interest in the status and outcome of all claims, and not simply in circumstances where insurance is likely to be vitiated (s. 138(2)(b)). For instance, the timing and status of insurance payouts may have implications for the creditworthiness of the operator or access holder. In addition, clause 16.9 is similar to requirements in other railways' access agreements.¹⁵⁹

As a result, we consider it is not appropriate to replace the requirement that parties notify Queensland Rail of reasonable details of insurance claims related to the SAA and keep Queensland Rail informed of developments.

We note Pacific National's suggestion that it is unreasonable to be required to notify of insurance claims not related to the subject matter of the SAA. In our view, this concern is addressed by clause 16.9(a)(i) of the proposed SAA, which provides for 'reasonable details of the claim *relevant to or arising out of the subject matter of this agreement*' [emphasis added].

We acknowledge Pacific National's concern that notifying Queensland Rail may be inappropriate and harmful to access holder's interests in certain other instances, namely where the information could prejudice the insurance outcome, breach confidentiality obligations or circulate commercially sensitive information. However, we note that the proposed SAA includes provisions that may relate to these concerns. Notably, clauses 24.1 and 24.2 of the proposed SAA outline obligations and exceptions in respect of parties' 'confidential information' as defined in clause 28.1.¹⁶⁰ It may be appropriate, and in the interests of access seekers and holders and Queensland Rail, to amend the

¹⁵⁸ Pacific National, sub. 7, p. 26.

¹⁵⁹ For example, Aurizon Network Pty Ltd's SAA – Coal, cl. 23.8(b) and (c), 2017 AU, 19 December 2019.

¹⁶⁰ This definition includes reference to disclosure of information that 'would reasonably be expected to adversely affect the commercial interests' of the disclosing party. In addition, clause 9.6 of the proposed SAA limits access to 'commercially sensitive' information, albeit in the conduct of inspections and audits.

proposed SAA to except notification under clause 16.9 in certain limited instances (ss. 138(2)(b), (e) and (h)). We invite submissions from stakeholders in this regard, preferably on a consensus basis.

Summary 4.2

- 1. **Limitation on claims:** it is appropriate to amend clause 13.2(a) to specify details under this provision are required only 'to the extent known'.
- 2. **Coverage of operator's associates, agents and other parties:** it may be appropriate to amend Queensland Rail's proposal. We invite stakeholders' submissions, including proposed drafting.
- 3. **Coverage and deductible levels:** it may be appropriate to approve the proposal. We invite stakeholders' submissions, including proposed drafting.
- 4. **Insurers' minimum financial strength rating:** it may be appropriate to approve the proposed amendment. We invite submissions from stakeholders, including Queensland Rail.
- 5. **Captive insurance companies:** it is not appropriate to approve the proposed amendment.
- 6. **Essential terms and conditions of insurance:** it is not appropriate to approve the proposed amendment.
- 7. **Insurance claims:** it may be appropriate to amend clause 16.9 to exempt the requirements for notification or information in certain circumstances. We invite stakeholders' submissions.

4.4 Dispute resolution escalation

Consistent with the 2020 undertaking SAA, the proposed SAA includes a dispute resolution process in clause 19. This section considers a proposed amendment to this process.

Pacific National proposed that clause 19.2 mirror the escalation timeframes and processes outlined in clause 6.1 (Dispute and complaint resolution process) of the 2025 DAU, on the basis that this would improve clarity and understanding of requirements.¹⁶¹

We acknowledge Pacific National's interest in improving the dispute resolution process in the proposed SAA. We consider that an effective and efficient dispute resolution regime under the proposed SAA is in the interests of all parties, including the operator and access holders, and the public, and is consistent with the objective of efficient operations under Part 5 of the Act (ss. 138(2)(a), (b), (d), (e) and (h)). Notably, enabling alternate dispute resolution processes, such as mediation, may facilitate an expedient resolution of disputes and avoid costs associated with court proceedings.

The existing dispute resolution regime has been in place for a substantial period, and there may be opportunities for Queensland Rail and stakeholders to engage with each other on whether any aspects of this process should be modified (including for reasons of modernisation).

¹⁶¹ Pacific National, sub. 7, p. 26.

We have provided guidance to promote workable and effective dispute resolution processes in relation to other declared services, notably the Dalrymple Bay Coal Terminal (DBCT) service.¹⁶² We consider that this guidance may be helpful to Queensland Rail and stakeholders in further engagement on this issue.

It may be appropriate to amend the dispute resolution process under the proposed SAA, and we therefore encourage Queensland Rail and other stakeholders to consider whether Pacific National's proposal is appropriate and to engage with each other on whether there are other aspects of the dispute resolution process in the proposed SAA that could operate more effectively with appropriate amendments.

Summary 4.3

It may be appropriate to amend the dispute resolution process under the proposed SAA. We encourage Queensland Rail and stakeholders to consider the proposed amendment to the dispute resolution escalation process and to engage with each other on whether other aspects of this process should be modified.

4.5 Assignment of Queensland Rail's rights under SAA

This section considers Queensland Rail's proposal regarding the circumstances in which it may assign its rights under the proposed SAA.

In the proposed SAA, Queensland Rail proposed adding a new clause (cl. 22.1(a)) to provide for the assignment of its rights under the SAA in circumstances where it ceases (or no longer expects) to have a right to operate all or part of the network.¹⁶³

Stakeholders were generally not supportive of the proposed change from the 2020 undertaking, and they proposed revised drafting. In particular, Pacific National submitted that, to ensure a reasonable balance between the parties on matters that may significantly impact their operations, assignments under clauses 22.1(a) and (b) should require other parties' prior consent, which would not be unreasonably withheld.¹⁶⁴ New Hope and Yancoal each submitted that they did not have concerns with Queensland Rail's rationale for assignment in circumstances where it ceased to have a right to operate the network; however, in their view, the proposed amendment, as drafted, appeared to go beyond that and inadvertently included certain inappropriate circumstances.¹⁶⁵

We acknowledge the importance of enabling Queensland Rail to assign its rights under a SAA to another party in appropriate circumstances, including where it ceases to have a right to operate all or part of the network (s. 138(2)(b)). We note that there is a difference in the eligibility standards of assignees under clauses 22.1(a) and (b) of the proposed SAA in terms of their necessary expertise and resourcing such that assignees under clause 22.1(a) must be able to provide the relevant access rights whereas assignees under clause 22.1(b) more broadly need to be able to meet Queensland

¹⁶² QCA, <u>Arbitration of disputes in relation to the DBCT service</u>, version 3, guideline, December 2021.

¹⁶³ Queensland Rail, sub. 1, pp. 66-67.

¹⁶⁴ Pacific National, sub. 7, pp. 26-27.

¹⁶⁵ New Hope Group, sub. 5, pp. 30-31; Yancoal, sub. 9, pp. 30-31. These parties submitted that inappropriate circumstances permitted under the proposal included situations where the parts of the network for which operatorship had changed were not relevant to the access rights under the agreement, or it was merely expected or not actually certain that operatorship had changed.

Rail's obligations under the access agreement. As Pacific National noted, assignments can significantly impact the operations of access holders and seekers (s. 138(2)(e), (h)).

We consider it important, and in the interests of all parties, that assignees who take on responsibilities under clause 22.1 should be capable of fulfilling Queensland Rail's obligations and provide the service on the same or substantially similar terms as Queensland Rail, failing which, contractual arrangements between the parties may essentially be voided.

Given these considerations and the concerns raised by stakeholders regarding the potential impact and scope of Queensland Rail's proposal, we consider that it is not appropriate to approve the proposal in its current form. We encourage Queensland Rail and stakeholders to discuss these issues further with a view to proposing a revised treatment.

Summary 4.4

Whilst acknowledging the legitimacy of the issue Queensland Rail is seeking to address, Queensland Rail's proposed treatment of assignment of its rights under the SAA is not appropriate to approve in its current form. We encourage Queensland Rail and stakeholders to discuss these issues further with a view to proposing a revised amendment.

4.6 Performance indicators

In this section, we consider a stakeholder's proposal to amend performance indicators in the proposed SAA.

Aurizon Coal and Bulk and the Rail Operators Group proposed that the list of performance indicators in Schedule 5, clause 1 be replaced with indicators that are consistent with reporting on aggregate system performance.¹⁶⁶ Aurizon Coal and Bulk submitted that the minimum list of key performance indicators (KPIs) in the existing SAA do not provide the most useful information for assessing performance under an access agreement for regularly scheduled train services. It also submitted that information similar to the aggregate system performance indicators, but specific to each train service specified in an individual access agreement, would improve consistency in assessing performance and enable an operator to understand the performance of its train service, relative to that of the entire system.

The Rail Operators Group submitted that clear, accountable obligations on rail infrastructure managers and objective key performance indicators supported market outcomes and modal shift. In the Rail Operators Group's view, individual operators lack the commercial leverage to require Queensland Rail's inclusion of performance obligations in individual access agreements. The Rail Operators Group submitted that although it is beyond the scope of our current process, adopting a consistent suite of key performance indicators across all rail infrastructure managers concerning adjacent networks would support standardisation of one aspect of access, provide a useful source of comparative data and assist in identifying constraints and areas for investment.¹⁶⁷

 ¹⁶⁶ Aurizon Coal and Bulk, sub. 2, pp. 44-45; Rail Operators Group, sub. 15, pp. 5-6.
 ¹⁶⁷ Rail Operators Group, sub. 15, pp. 5-6.

Clause 6.7 of the proposed SAA provides that Queensland Rail will provide monthly reports to each party, documenting Queensland Rail's performance regarding the performance levels set out in Schedule 1. This clause also allows for parties, upon request, to negotiate in good faith for agreement on additional performance levels and an associated reporting regime within 12 months (or as otherwise agreed) of the SAA's commencement date.

Clause 6.7 also provides for resolution of performance level-related disputes. We consider that clause 6.7 provides adequate flexibility, in the interests of all parties, to agree on additional performance indicators beyond the indicators in Schedule 5, clause 1 or, in the absence of an agreement, to seek resolution thereof (ss. 138(2)(b), (d), (e) and (h)). However, we note the concern raised by the Rail Operators Group that individual rail operators do not have commercial leverage to require Queensland Rail to include performance obligations in individual access agreements. The benefits of at least some degree of standardisation and comparability of key performance indicators are also consistent with the parties' interests and the goal of economic efficiency in Queensland Rail's operations (ss. 138(2)(a), (c), (e) and (h)).

Our preliminary view is that it may be appropriate to amend the list of performance indicators in Schedule 5, clause 1. We encourage the parties to engage further with each other on appropriate revisions to this list, and we invite stakeholders' further submissions in this regard .

Summary 4.5

It may be appropriate to approve amendment of the list of performance indicators in Schedule 5, clause 1 of the proposed SAA. We encourage the parties to engage further with each other on this issue, and we invite stakeholders' further submissions, including proposed drafting.

Pricing rules

Queensland Rail proposed that charges for access to the declared service be determined in accordance with the pricing rules outlined in Part 3 of the 2025 DAU.

Overview of the draft decision

Our draft decision is that it is appropriate to approve Queensland Rail's proposed pricing rules.

Pricing rules (Part 3) – summary

Proposal	Clause	QCA draft decision
Pricing rules		
Queensland Rail proposed pricing rules unchanged from those in the 2020 undertaking.	Part 3	It is appropriate to approve Queensland Rail's proposed pricing rules. See section 5.2.
Price differentiation		
Stakeholders proposed enabling greater price differentiation between access holders. ¹⁶⁸	Part 3	It is not appropriate, at this time, to require amendments to the pricing rules to provide for greater price differentiation. See section 5.3.
Mount Isa criteria		
Stakeholders proposed including service-specific criteria for the Mount Isa line. ¹⁶⁹	n/a	It is not appropriate to require amendments obliging Queensland Rail to develop service- specific criteria for the Mount Isa line. See section 5.4.

Overview of pricing rules in the 2025 DAU 5.1

The pricing rules specify that for those train services that accord with the description of a reference train service (i.e. coal trains using the West Moreton route service), the relevant reference tariff applies. We have separately assessed Queensland Rail's proposed West Moreton reference tariffs (see chapters 8 and 9).

For all other train services, the 2025 DAU prescribes pricing rules that apply in negotiating access charges (see Figure 5).¹⁷⁰

¹⁶⁸ Aurizon Coal and Bulk, sub. 2, pp. 27-28; Glencore, sub. 11, p. 3; Centrex, sub. 10, pp. 2-3; North West Phosphate, sub. 6, p. 2. ¹⁶⁹ Aurizon Coal and Bulk, sub. 2, p. 29; Mount Isa Line Users, sub. 17, pp. 2-3.

¹⁷⁰ The rules will apply in the order listed in Figure 5, to address any conflicts that may arise between the pricing rules (2025 DAU, cl. 3.5).

Figure 5: Pricing rules prescribed in the 2025 DAU

Price differentiation – Establishes the extent access charges can vary among customers. Queensland Rail will not differentiate between access seekers where the characteristics of the train service are alike and the access seekers operate in the same end market. (cl. 3.3)	
Pricing limits – Establishes an upper and lower bound for price negotiations. The expected revenue from	
access charges must fall between the incremental cost and standalone cost of providing access. (cl. 3.2)	
Network utilisation – Provides for Queensland Rail to determine access charges above the price ceiling in	
certain circumstances where capacity is insufficient to meet the requests of all access seekers. (cl. 3.1.2)	
Revenue adequacy – Provides for Queensland Rail to recover at least enough expected revenue to meet	
the efficient costs of providing access, including a return on investment commensurate with the regulatory and commercial risks involved. (cl. 3.1.1)	

The pricing rules also outline that:

- Queensland Rail or an access seeker may require reasonable and balanced rate review provisions in an access agreement to enable access charges to be adjusted to account for changes over time¹⁷¹
- access charges may include a QCA levy component to account for our annual regulatory fees, which is to be allocated between train services in accordance with Schedule J of the 2025 DAU.¹⁷²

5.2 Guiding the negotiation of access charges

We consider that the pricing rules establish appropriate bounds to guide the negotiation of access charges.

The pricing rules require that negotiated access charges be set such that expected revenue from that access seeker does not:

- exceed a ceiling revenue limit the standalone cost of providing access to any train service (or group of train services)
- fall below the floor revenue limit the incremental cost of providing access to any train service (or group of train services).

These pricing limits result in access charges (and transport service payments) for each part of the network being at least sufficient to meet the incremental cost of providing access and establish an upper bound for price negotiations that reflects the efficient costs of providing access.¹⁷³

Furthermore, within these bounds, the pricing limits provide flexibility to differentiate access charges, to reflect the differing characteristics of the various markets that use Queensland Rail's network. However, Queensland Rail's proposed pricing rules apply restrictions on the extent that it can differentiate access charges between users. In particular, Queensland Rail must not differentiate

¹⁷¹ 2025 DAU, cl. 3.6.

¹⁷² 2025 DAU, cl. 3.7.

¹⁷³ The formula for calculating the ceiling revenue limit is consistent with a building block approach, whereby the revenue a firm is allowed to earn reflects the estimated efficient costs of providing the relevant service, including an appropriate return on investment.

between users where the characteristics of the train service are alike and the access seekers are competing in the same end market.^{174,175}

We consider that together the pricing limits and price differentiation rules provide an appropriate balance between:

- promoting the efficient use of the declared service
- limiting the risk of distorting competition in dependent markets for the service.¹⁷⁶

Some stakeholders submitted that enabling further price differentiation amongst users was appropriate. We consider those submissions below.

The pricing rules also provide for Queensland Rail to generate expected revenue from access charges and transport service contract payments (if applicable) that is at least enough to meet the efficient costs of providing access.¹⁷⁷ We consider this promotes efficient operation of and investment in the declared service (s. 138(2)(a)). Furthermore, these arrangements are consistent with pricing principles in section 168A of the QCA Act (s. 138(2)(g)).

In certain circumstances where capacity is insufficient to meet the requests of all access seekers, the ceiling revenue limit for a train service may be revised to reflect an access seeker's willingness to pay. Enabling access charges to reflect access seekers' willingness to pay promotes both the efficient use of, and investment in, the network, especially in circumstances where available capacity is constrained. We consider that these arrangements appropriately balance the legitimate interests of Queensland Rail and users of the network (ss. 138(2)(b), (e) and (h)).

We note that the pricing rules proposed in the 2025 DAU are the same as the pricing rules applied in the 2020 undertaking.

Summary 5.1

It is appropriate to approve Queensland Rail's proposed pricing rules (Part 3 of the 2025 DAU).

5.3 Enabling greater price differentiation

We do not consider it necessary to require amendments to the pricing rules to include a commitment by Queensland Rail to provide for greater price differentiation between access holders.

A number of stakeholders considered that the regulatory arrangements needed to provide for more opportunities to price-differentiate between users.

¹⁷⁴ 2025 DAU, cl. 3.3(d).

¹⁷⁵ The 2025 DAU provides for us to direct Queensland Rail to offer an access holder an access charge that is charged to another access holder whose train service characteristics are alike and are competing in the same end market (2025 DAU, cl. 3.9).

¹⁷⁶ Queensland Rail is not vertically integrated into above-rail freight operations.

¹⁷⁷ Including a return on investment commensurate with the regulatory and commercial risks involved (2025 DAU, cl. 3.1.1). The pricing rules also enable Queensland Rail to require reasonable rate review provisions in negotiated access agreements to adjust access charges to reflect changes over time (see 2025 DAU, cl. 3.6); and to charge users a levy to recover the annual fees it pays to us (2025 DAU, cl. 3.7).

Aurizon Coal and Bulk submitted that Queensland Rail adopted a rigid approach to the specification of access proposals and was unwilling to negotiate around these proposals, instead applying a 'take it or leave it' approach to the access terms offered.¹⁷⁸ Moreover, Aurizon Coal and Bulk considered that Queensland Rail used its regulatory obligation to not unfairly discriminate between access seekers and access holders as a shield against attempts to negotiate greater price differentiation and achieve more flexible access terms.¹⁷⁹ Glencore also submitted that in its experience prices were provided on a take it or leave it basis.¹⁸⁰

Aurizon Coal and Bulk submitted that Queensland Rail's inflexible and high-cost access arrangements significantly hindered Aurizon Coal and Bulk's ability to respond to opportunities to attract freight from road on the Mount Isa line.¹⁸¹ Aurizon Coal and Bulk considered it important that the 2025 DAU did not enable Queensland Rail to simply set access charges on a 'take it or leave it' basis, further entrenching Queensland Rail's ability to impose pricing outcomes and limit effective negotiation with access seekers.¹⁸²

We consider that pricing flexibility promotes the efficient use of rail infrastructure, enabling Queensland Rail to adjust prices in response to competition from alternative modes of transport and expand the demand for its service by targeting customers that are more price-sensitive.

Queensland Rail's pricing rules provide scope for Queensland Rail to differentiate access charges, to reflect the differing characteristics of the various markets that utilise the network.

We also consider that Queensland Rail has an incentive to increase the revenue it recovers from access charges. Queensland Rail is not vertically integrated into above-rail freight operations. As such, within the bounds of the pricing rules, we consider that Queensland Rail has an incentive to:

- negotiate access to the declared service at a price above the price floor, where there is available capacity
- differentiate prices to reflect the willingness to pay of different users that are competing in different end markets.

The benefits associated with further price differentiation need to be balanced with the objective of having negotiated access charges that:

- enable Queensland Rail to generate expected revenue that is at least enough to meet the efficient cost of providing access to its network¹⁸³
- do not discriminate between access holders to the extent that has implications for competition in downstream markets.

The 2025 DAU does not allow Queensland Rail to differentiate between users where the characteristics of the train service are alike and the access seekers are competing in the same end market.

We have considered stakeholders' submissions calling for specific amendments to the 2025 DAU to provide for additional price differentiation in the negotiation of access charges. For the reasons outlined below, we do not consider that it is appropriate to require amendments to the pricing rules to provide for greater price differentiation between access holders.

¹⁷⁸ Aurizon Coal and Bulk was concerned that such an approach imposes commercial risks on access holders regardless of their ability to manage these risks with customers, with Queensland Rail accepting limited accountability in the provision of access.

¹⁷⁹ Aurizon Coal and Bulk, sub. 2, pp. 12, 23.

¹⁸⁰ Glencore, sub. 11, p. 3.

¹⁸¹ Aurizon Coal and Bulk, sub. 2, pp. 12-13.

¹⁸² Aurizon Coal and Bulk, sub. 2, p. 28.

¹⁸³ We consider that this appropriately balances the legitimate business interests of Queensland Rail, access holders and access seekers, as well as promotes the economically efficient investment in the network.

5.3.1 Differentiation for multi-product trains

We do not consider it is appropriate to amend the 2025 DAU to require Queensland Rail to differentiate access charges for different products on multi-product trains, as part of negotiations with access holders.

Aurizon Coal and Bulk considered that the pricing rules should clearly permit differentiation for different products on multi-product trains to grow rail volumes, through supporting and incentivising emerging demand.¹⁸⁴ It said this increased sophistication in applying access charges would likely be necessary to enable operators to compete effectively for mode contestable freight. Aurizon Coal and Bulk considered that the current regulatory arrangement created sufficient flexibility to allow this form of price differentiation, but Queensland Rail's risk aversion meant it was unwilling to apply this pricing flexibility.¹⁸⁵

Queensland Rail considered that such an approach could lead to sub-optimal outcomes for both Queensland Rail and access seekers. Amongst other things, Queensland Rail considered that multicommodity pricing significantly increased the complexity of negotiations and contract agreements, which might lead to prolonged negotiation processes, encourage disputes and hinder market efficiency. Queensland Rail also said the practical implementation of differentiation within multicommodity services posed administrative burdens and challenges in accurately managing, tracking, reporting and billing different commodities.¹⁸⁶

We consider price differentiation may promote the efficient use of rail infrastructure. However, in practice, differentiating access charges between products on a multi-product train based on their differing market characteristics to achieve more efficient outcomes can be challenging.

For Queensland Rail to efficiently price-differentiate between different products, it requires sufficient information on the customers' willingness to pay for the various products being carried on a multi-product train. In this regard, HoustonKemp (in a report commissioned by Queensland Rail) stated that Queensland Rail:

- had limited knowledge of its end customers and the goods that were carried within each container
- did not have oversight of the contents of containers that were carried on its network, which meant it would not be able to verify the charge that should be applied to each train.¹⁸⁷

Adding to these complexities, a customer's willingness to pay will continue to change to reflect market conditions related to the respective products. Glencore considered it was 'very dangerous' to provide a different pricing outcome for different anticipated values of products where pricing for minerals can be quite cyclical and vary significantly during an undertaking term.¹⁸⁸

We consider that solutions for these matters need further consideration from the affected parties before such an access pricing approach can be implemented in practice for a multi-product train.

¹⁸⁴ Aurizon Coal and Bulk, sub. 2, pp. 26-27. Mount Isa Line Users (sub. 17, p. 2) supported Aurizon Coal and Bulk's submission.

¹⁸⁵ Aurizon Coal and Bulk, sub. 2, pp. 26-27.

¹⁸⁶ Queensland Rail, sub. 14, pp. 47-48.

¹⁸⁷ Queensland Rail, sub. 14, pp. 109-110.

¹⁸⁸ Glencore, sub. 11, p. 7.

Queensland Rail's 2025 Draft Access Undertaking

5.3.2 Differentiation for premium pathways

Aurizon Coal and Bulk submitted that the pricing rules should also clearly permit differentiation to reflect the different market value of non-premium paths compared to premium paths within a given market.

In response, Queensland Rail considered that this approach was worth further consideration, as in theory it could promote allocative efficiency given:

- train paths and priority would be allocated to customers who valued them the most
- it potentially allowed Queensland Rail to recover additional revenue, through price differentiation.¹⁸⁹

While certain stakeholders may support adopting this pricing approach, implementing price differentiation to reflect the different market value of non-premium paths is likely to require revisions to current contractual arrangements between Queensland Rail and access holders. For instance, Queensland Rail may need to renegotiate access charges to reflect the extent to which the relevant train services are considered to be either premium or non-premium.

In this regard, Queensland Rail submitted that consideration of how such an approach affected contractual relationships governing access and service would need to be evaluated and any issues addressed. Queensland Rail acknowledged that this would represent a significant and material change to how it operated its network and could have a material effect on other stakeholders.¹⁹⁰

We encourage stakeholders to discuss this matter, and collaborate and, where possible, reach consensus on it. We will have regard to consensus positions reached amongst stakeholders as part of our assessment of Queensland Rail's 2025 DAU.

5.3.3 Price differentiation to support emerging demand

We do not consider it is appropriate to amend the 2025 DAU to require Queensland Rail to further differentiate access charges for junior miners, as part of negotiations with access holders.

Aurizon Coal and Bulk considered that the pricing rules should further clarify the circumstances in which Queensland Rail may apply price differentiation in order to achieve the following negotiation objectives:

- promoting the economically efficient investment in, and use and operation of, the network
- growing rail volumes including by supporting and incentivising emerging demand, retaining volumes on rail and supporting 'road to rail' modal conversion.¹⁹¹

Glencore considered that it would assist in increasing rail volumes, thereby driving greater economies of scale in the medium term¹⁹², if the price ceiling was set at the lower of:

- the cost at which road transport would be economic
- the stand-alone efficient cost
- the cost that is a 10% margin above the floor price.¹⁹³

¹⁸⁹ Queensland Rail, sub. 14, p. 54.

¹⁹⁰ Queensland Rail, sub. 14, pp. 54-55.

¹⁹¹ Aurizon Coal and Bulk, sub. 2, pp. 27-28.

¹⁹² Glencore (sub. 11, p. 2) considered that Queensland Rail's unwillingness to price rail access in a manner that made it viable for end users to invest in greenfield projects and expansions of operations using the line had now reached the point of creating challenges for the ongoing sustainability of the Mount Isa line.

¹⁹³ Glencore, sub. 11, pp. 5-6.

Centrex considered that there were valid reasons to consider price differentiation for junior mining companies or for those with a lower commodity value requiring bulk and economies of scale in order for them to be competitive. Centrex considered that imposing higher fees acted as a deterrent to innovation, especially for mines that heavily relied on rail transport due to geographic location.¹⁹⁴

We consider that Queensland Rail has an incentive to increase the revenue it recovers from access charges. As outlined by Centrex, if access charges at the ceiling exceed customers' willingness to pay, it is in the interests of all parties to negotiate access charges below the ceiling.¹⁹⁵

However, where access charges are set below the price floor, Queensland Rail does not recover the efficient costs associated with providing that train service access to its network. Implementing pricing policies as part of the regulatory framework to provide for such outcomes may reduce incentives to invest in the network¹⁹⁶ or require other train services to cross-subsidise the provision of access.

Furthermore, policies to enable price differentiation for junior miners have the potential to result in price discrimination between customers that compete in the same end market, which may affect competition in these markets. Aurizon Coal and Bulk considered it was critical that the limits on price differentiation continued to be applied in a way that protected the environment for competition.¹⁹⁷

Further changes to the regulatory arrangements would also be needed to provide for Queensland Rail to revisit access charges as the circumstances of the firm changed.

For these reasons, we consider that measures to encourage more freight from road to rail, which result in subsidised access prices, are better considered and provided through means other than access regulation, within the context of wider transport policy and industry assistance objectives.

5.3.4 Other pricing arrangements

We do not consider it is appropriate to require amendments to the 2025 DAU to require Queensland Rail to offer alternative pricing arrangements besides a two-part tariff, with take-or-pay and relinquishment fee obligations, as part of negotiations with access holders.

Stakeholders submitted that features of Queensland Rail's standard access proposals may deter use of the Mount Isa line by smaller miners. For instance:

• North West Phosphate and Centrex submitted that access charges with a high proportion of fixed charges and take-or-pay obligations introduced significant risk (particularly to smaller mining operations in a start-up situation¹⁹⁸), which undermined new project developments and incentivised short-term contracts.¹⁹⁹

¹⁹⁴ Centrex, sub. 10, pp. 2-3. North West Phosphate (sub. 6, p. 2) considered that further price differentiation would enable smaller, less profitable junior miners to scale production and better access new markets.
¹⁹⁵ Centrex, sub. 10, p. 2.

¹⁹⁶ We do not consider that this will promote economically efficient investment in Queensland Rail's network, nor do we consider this appropriately balances the legitimate business interests of Queensland Rail and access seekers.
¹⁹⁷ Aurizon Coal and Bulk, sub. 2, p. 27.

¹⁹⁸ Centrex considered that these types of operations can encounter irregular production issues, which in turn introduces a risk of filling regular train services.

¹⁹⁹ Centrex, sub. 10, p. 4; North West Phosphate, sub. 6, p. 1. Aurizon Coal and Bulk (sub. 2, p. 21) also considered that the materiality of take-or-pay obligation exceeds the opportunity costs to Queensland Rail of providing the service, given there is no capacity constraint on the Mount Isa line and the associated avoidable costs of services not operating are low.

• Aurizon Coal and Bulk considered that long-term contracts with high path relinquishment fees presented a high risk for freighter services where demand was variable and uncertain (and not underpinned by long-term take-or-pay contracts).²⁰⁰

While these pricing arrangements may allocate the risk of contracted volumes not materialising to access holders, we consider that those access holders are best placed to manage such risk. They are in the best position to forecast the volumes that they require. Furthermore, they are also able to manage these risks by entering into short-term contractual arrangements.

We consider that these pricing arrangements are consistent with promoting the efficient use of the network by access holders. For instance, take-or-pay and relinquishment fee obligations incentivise access holders to only contract capacity that they require. Queensland Rail considered that without effective take-or-pay agreements, less commercially viable traffics could consume substantial capacity at the commercial expense of Queensland Rail.²⁰¹

This is particularly important where network capacity is constrained. Aurizon Coal and Bulk questioned the need for the level of take-or-pay obligations, submitting that it seemed there was more than sufficient capacity available for all users on the Mount Isa line.²⁰² However, Queensland Rail submitted that as traffic grew on the Mount Isa line, take-or-pay would become vital for efficient path utilisation.²⁰³

We also consider that access pricing arrangements that include a two-part tariff, with take-or-pay and relinquishment fee obligations, promote efficient investment in, and operation and use of, the network. These arrangements establish a clear incentive for access seekers and access holders to only enter into contracts for capacity they need, providing a more accurate signal to Queensland Rail as to how much capacity needs to be delivered on its network. This enables Queensland Rail to plan and undertake maintenance and capital activities to efficiently deliver the required capacity.

Moreover, these arrangements provide certainty that Queensland Rail is able to recover the associated fixed costs from those access seekers and access holders.²⁰⁴ Queensland Rail considered it was only fair that parties entering into capacity contracts committed to using that capacity, given Queensland Rail invested in the network to reflect future contracted capacity.²⁰⁵

Therefore, we consider that specific commercial initiatives, rather than changes to the regulatory framework, provide a more targeted option for limiting the extent that Queensland Rail's standard access proposals may deter use of the Mount Isa line by smaller miners. Queensland Rail submitted that it had introduced multiple commercial initiatives for its Mount Isa line customers during the AU2 regulatory period, including:

- providing a variable charge only (for a limited period) for new operations on the Mount Isa line where the short-term demand or train profile was uncertain, with a view to transfer services to a traditional fixed and variable charge when the operation stabilised
- special pricing arrangements for trial shipments of new products (e.g. rock phosphate) so that end customers could test the products in end markets
- an upfront discount for intermodal logistics on the Mount Isa line

²⁰⁰ Aurizon Coal and Bulk, sub. 2, p. 22. Aurizon Coal and Bulk also considered that while the standard relinquishment fee can be reasonable and efficient in circumstances to provide an access provider with certainty that it is able to recover investments required to provide a service, these circumstances do not apply on the Mount Isa line.

²⁰¹ Queensland Rail, sub. 14, p. 39.

 ²⁰² Aurizon Coal and Bulk, sub. 2, p. 21. Glencore (sub. 11, p. 5) also considered that given the Mount Isa line is underutilised there is no purpose of ensuring access seekers pay a fixed take-or-pay charge for the purposes of 'reserving' capacity'.
 ²⁰³ Queensland Rail, sub. 14, pp. 39-40.

²⁰⁴ A significant proportion of the costs incurred by Queensland Rail to provide access are fixed.

²⁰⁵ Queensland Rail, sub. 14, p. 39.

• developing take-or-pay provisions to proportionally recognise credits from secondary agreements, to reduce some of the downside financial risk in contracting additional paths as businesses grew.²⁰⁶

Summary 5.2

It is not appropriate, at this time, to require amendments to the pricing rules to provide for greater price differentiation between access holders.

5.4 Service-specific criteria for the Mount Isa line

Aurizon Coal and Bulk considered that the 2025 DAU should be amended to include servicespecific negotiation criteria for multi-commodity freighter services on the Mount Isa line.²⁰⁷ The elements that Aurizon Coal and Bulk considered should be addressed by specific negotiation criteria for the Mount Isa line are outlined in Table 1.

Aurizon Coal and Bulk considered that these negotiation criteria would provide greater structure around access negotiations, particularly for junior miners or other road contestable freight, where demand was more uncertain. Aurizon Coal and Bulk considered that this would also facilitate a fast and efficient contract renegotiation process at term expiry.²⁰⁸

Mount Isa Line Users supported Aurizon Coal and Bulk's submission that specific negotiation criteria should be developed to apply to multi-commodity freighter services on the Mt Isa line. Mount Isa Line Users considered that Queensland Rail's current approach to negotiations was not well aligned to the needs of its customers, particularly for multi-user freighter type services, which needed to continually attract customers to maximise utilisation.²⁰⁹

For the reasons outlined in Table 1, we do not consider it appropriate to require amendments to the 2025 DAU to oblige Queensland Rail to develop service-specific criteria for the Mount Isa line.

Matters raised by Aurizon Coal and Bulk ²¹⁰	QCA preliminary position
Negotiation criteria should set out circumstances in which the access price differentiation will be applied to aid efficiency, including varying the standard access price to provide incentives for emerging demand or freight that is highly contestable with road.	For the reasons outlined in section 5.3.3, we do not consider it is appropriate to require Queensland Rail to further differentiate access charges for emerging demand, as part of negotiations with access holders.

Table 1: Consideration of matters to be addressed by specific criteria for the Mount Isa line

²⁰⁶ Queensland Rail, sub. 14, p. 34.

²⁰⁷ Aurizon Coal and Bulk, sub. 2, pp. 28-30.

²⁰⁸ Aurizon Coal and Bulk, sub. 2, p. 29.

²⁰⁹ Mount Isa Line Users, sub. 17, pp. 2-3.

 $^{^{\}rm 210}$ Aurizon Coal and Bulk, sub. 2, p. 29.

Matters raised by Aurizon Coal and Bulk²¹⁰

The definition of products that may be carried on train services should be specified broadly to promote an operator's ability to run an efficient train service and attract additional products to its train, and to facilitate price differentiation for different products on those trains where relevant.

Negotiation criteria should address price structure, including the weighting between fixed and variable charges, and consider applying multi-part tariffs where this will aid efficiency.

Take-or-pay and relinquishment fee arrangements should be set to promote efficiency, having regard to issues such as the incremental investment required to provide the required capacity (if any), the level of corridor utilisation, amount of required capacity, and the ability to impose take-or-pay commitments to freight customers.

Negotiation criteria should address quality of rail access service, potentially including performance-based KPIs, to ensure that service quality is maintained. For the reasons outlined in section 5.3.1, we do not consider it is appropriate to amend the 2025 DAU to require Queensland Rail to differentiate access charges for different products on multi-product trains, as part of negotiations with access holders.

QCA preliminary position

For the reasons outlined in section 5.3.4, we do not consider it is appropriate to amend the 2025 DAU to require Queensland Rail to offer alternative pricing arrangements besides a two-part tariff, with take-or-pay and relinquishment fee obligations, as part of negotiations with access holders.

Chapter 7 outlines the reporting obligations that we consider should be incorporated in the 2025 undertaking to provide interested parties with adequate information on how efficiently Queensland Rail has been performing.

Summary 5.3

It is not appropriate to require amendments to the 2025 DAU to oblige Queensland Rail to develop service-specific criteria for the Mount Isa line.

6 Operating requirements

Queensland Rail's 2025 DAU outlines operational requirements for Queensland Rail in providing below-rail services to access holders, as well as obligations for train operators to use the network.

Part 4 of the 2025 DAU establishes the operating requirements that govern how Queensland Rail delivers train service entitlements. These include:

- **network management principles**, which outline how Queensland Rail will, amongst other things, coordinate maintenance and other track restrictions, and schedule and manage train services on the network (Schedule F of the 2025 DAU)
- **an operating requirements manual**, which prescribes rules and procedures for operating trains on the network and addressing matters such as safety and emergency responses (Schedule G of the 2025 DAU).

The proposed operating requirements also include a commitment from Queensland Rail to convene and support regional network user groups – to facilitate discussion and consensus amongst stakeholders, with an aim to promote productivity and operational improvements.²¹¹

Overview of the draft decision

We consider it is appropriate for Queensland Rail to make certain amendments to the operating requirements in the 2025 DAU.

Queensland Rail proposal	Clause	QCA draft decision
Network management principles		
Queensland Rail proposed that the MTP be changed in response to a request from an access holder. ²¹²	Sch. F, cls. 2.1(e) and (f)	It is appropriate to amend the 2025 DAU to provide for Queensland Rail to be more responsive to requests from operators to change the MTP. See section 6.1.
Operating requirements manual		
Queensland Rail proposed amendments to the operating requirements manual ²¹³	cl. 4.3 and Sch. G	It is appropriate to approve Queensland Rail's proposed operating requirements manual. See section 6.2.
User groups		
Queensland Rail proposed obligations regarding productivity and operational improvements. ²¹⁴	cl. 4.4	It is appropriate to approve Queensland Rail's proposed obligations regarding productivity and operational improvements. See section 6.3.

Operating requirements (Part 4) – summary

- ²¹² Queensland Rail, sub. 1, Sch. F, cl. 2.1.
- ²¹³ Queensland Rail, sub. 1, cl. 4.3 and Sch. G.

²¹⁴ Queensland Rail, sub. 1, cl. 4.4.

²¹¹ 2025 DAU, cl. 4.4.

6.1 Network management principles

In general, we consider that Queensland Rail's proposed network management principles appropriately balance the need to provide access holders with certainty about the scheduling of train services with the need to provide sufficient flexibility for Queensland Rail to address network constraints and to be responsive to requests of customers.

The proposed network management principles are unchanged from the principles contained in the 2020 AU, with the exception of proposed amendments to Queensland Rail's obligations following a dispute on a planned possession.

The network management principles in the 2025 DAU establish a predictable and transparent process that outlines how Queensland Rail is to schedule train services and undertake maintenance activities on its network. As part of this process, Queensland Rail is to first develop and publish:

- a MTP, which details the scheduled times as advised by Queensland Rail for all train services and any regular planned possessions where scheduled times are unchanged from week to week²¹⁵
- a supply chain calendar (SCC), which lists upcoming regular and ad hoc planned possessions, urgent possessions (to the extent known) and special events on the network.²¹⁶

Queensland Rail is to then develop the daily train plan (DTP)²¹⁷ to provide the actual expected schedule on the day of operation, which is derived from the MTP and SCC.²¹⁸

The network management principles also provide flexibility for Queensland Rail to manage scheduling of trains services to address network constraints as they arise, and to be responsive to requests of customers. In this regard, Queensland Rail may modify an MTP or schedule an ad hoc possession, but must notify all affected parties at least three months prior to the commencement of a modification.²¹⁹ The MTP may also be changed in response to a request from an access holder, where an access holder gives Queensland Rail notice to consider and implement the requested change. Notice must be given no less than:

- six months prior to a change relating to a passenger service
- three months prior to a change relating to a non-passenger service.²²⁰

Queensland Rail may then only schedule a DTP in variation to a MTP in certain circumstances, including:

- to accommodate operational constraints (see Box 2) and special events, as defined in Part 7 of the 2025 DAU
- to accommodate requests from access holders to change the schedule of its train services or run an ad hoc train service
- where all affected access holders agree to the modification.²²¹

²¹⁵ Queensland Rail may modify an MTP or schedule an ad hoc possession but is to notify all affected parties of any modifications to an MTP.

²¹⁶ 2025 DAU, Sch. F, cl. 2.1(a).

²¹⁷ The DTP details the scheduled times for all train services and regular and ad hoc planned possessions, urgent possessions and emergency possessions for a particular day on a specified part of the network (2025 DAU, Part 7).

²¹⁸ Aurizon Coal and Bulk (sub. 2, pp. 37-38) submitted that in practice the process for scheduling train services in the West Moreton line involves Queensland Rail developing a western corridor alignment calendar and a forecast plan.²¹⁸ While additional deliverables may assist with Queensland Rail's task to efficiently schedule train services and maintenance activities in certain systems, we do not consider it is necessary to require additional obligations be included in the 2025 DAU.

²¹⁹ 2025 DAU, Sch. F, cl. 2.1(d).

²²⁰ 2025 DAU, Sch. F, cl. 2.1(e),(f).

²²¹ 2025 DAU, Sch. F, cl. 2.2(e)–(h).

Box 2: Operational constraints

Operational constraints may include track closures for maintenance and construction activities, or restrictions on train weights or speeds. The four types of operational constraints for maintenance and construction are:

- emergency possessions closures to correct 'dangerous or potentially dangerous' faults or 'severe speed restrictions' within five days after they are detected
- urgent possessions closures to correct 'potentially dangerous' problems less than three months after they are detected (other than emergency possessions)
- regular planned possessions closures that occur at regular intervals that are entered into the MTP and DTP
- ad hoc planned possessions closures (other than emergency and urgent possessions) that are not entered into the MTP because they are not regular planned possessions.

The 2025 DAU requires access holders to make any request to change the schedule of its train services or run an ad hoc train service at least two business days prior to the day of operation and prior to the DTP being scheduled.²²²

Certain stakeholders suggested amendments to the network management principles that they considered would enable Queensland Rail to be more responsive to the requests of customers (see Table 2). For the reasons outlined in Table 2, we consider it is appropriate to amend the 2025 DAU to provide for Queensland Rail to be more responsive to requests from operators to change the MTP.

Table 2: Consideration of suggested amendments to the network management principles

Stakeholder submission	QCA position
Aurizon Coal and Bulk and Pacific National both considered that a requirement to give three months' notice to Queensland Rail for requesting a change to the MTP was too long, particularly where a request had no impact on any other service. To facilitate more responsive outcomes for end customers, Pacific National submitted that it would be reasonable to reduce the MTP modification timeframe to one month for requests that would not impact other operators. ²²³	Changes to scheduling can have implications for end users' logistics and can impose material costs on those users. The 2025 DAU requires Queensland Rail to provide affected parties with 3 months' notice of any modifications to an MTP. ²²⁴ However, where a request to modify an MTP or a planned possession does not impact other access holders, we consider it is appropriate that Queensland Rail reduce the notification timeframes associated with modifying the MTP to one month.

²²² 2025 DAU, Sch. F, cl. 2.2(j).

²²³ Aurizon Coal and Bulk, sub. 2, pp. 36-37; Pacific National, sub. 7, p. 21.

²²⁴ Planned possession are part of a long-term planning schedule. The requirement to give three months' notice was introduced as part of AU1 in response to concerns from certain access holders that the 20-day notice period proposed by Queensland Rail was insufficient.

Stakeholder submission	QCA position
 Aurizon Coal and Bulk submitted that in practice Queensland Rail appeared to adopt a process whereby it revised the MTP periodically and only on fixed dates.²²⁵ Aurizon Coal and Bulk considered that if such a process was adopted by Queensland Rail, then: a process for incorporating new or changed contracted paths in the MTP in between review periods should be included in the network management principles²²⁶ there should be no need for notification and consultation processes with other access holders and other parties unless their services or activities were affected by the new or changed path.²²⁷ 	The 2025 DAU does not prescribe a process whereby Queensland Rail is to periodically revise the MTP on set dates. However, where this approach is applied in practice, it has the potential to unnecessarily prolong the timeframes associated with responding to stakeholder requests. As such, we consider it is appropriate to include a process in the network management principles to incorporate changes to contracted paths in between periodic revisions of the MTP.
Pacific National considered that the requirement for access holders to request an ad hoc service at least two business days in advance could impact a rail operator's ability to flexibly deliver for customers. Pacific National suggested that time frames be capped at a maximum of 24 hours for Queensland Rail to respond to rail operator requests for short-term path variations or an ad hoc path. ²²⁸	We consider that providing access holders with opportunities to request short-term path variations or ad hoc paths needs to be balanced with any potential implications for other end users that may be affected by those changes to the scheduling. The 2025 DAU provides for Queensland Rail to amend the DTP after it has been scheduled, at the request of an access holder – provided the change does not affect another access holder's train service entitlement. ²²⁹
Pacific National submitted that the network management principles should be amended to recognise maximum corridor lengths that trains could run to. ²³⁰ Pacific National considered that this would allow rail operators to be more responsive for customers, as well as aligning with processes on other networks. ²³¹	Amending the network management principles to recognise maximum corridor lengths may affect how Queensland Rail would be able to schedule new train services and possessions on its network. This may have implications for Queensland Rail's ability to, amongst other things, respond to requests from access holders to amend their scheduled paths, and schedule maintenance activities on its network. We encourage stakeholders to discuss this matter, collaborate, and where possible, reach consensus on it. We will have regard to consensus

²²⁵ As an example, Aurizon Coal and Bulk submitted that MTP revisions may be set in some systems to align with summer and winter timetable dates.

²²⁶ The Rail Operator Group (sub. 15, pp. 2, 7-8) supported amending the 2025 DAU to include a mechanism for more ²²⁷ Aurizon Coal and Bulk, sub. 2, pp. 36-37. The Rail Operators Group (sub. 15, pp. 7-8) also considered that it was

reasonable to reduce the MTP modification timeframe to one month or consider whether there was even a need for notification and consultation with other access holders and parties unless their services were affected by the new or changed path.

²²⁸ Pacific National, sub. 7, pp. 21–22.

²²⁹ 2025 DAU, Sch. F, cl. 2.2(j).

²³⁰ When a network corridor is confirmed as having the capacity to run to that longer train length, that length train would be available to any path or service on that track. Requests to run longer trains would be addressed in the DTP. ²³¹ Pacific National, sub. 7, p. 22.

Stakeholder submission	QCA position
	positions reached amongst stakeholders as part of our assessment of Queensland Rail's 2025 DAU.
GrainCorp considered that the inflexibility of the MTP was at odds with the inherent daily variability needs of grain trains. ²³² GrainCorp submitted that Queensland Rail and train operators were instead applying the principles of preserved pathing, making a number of non-coal paths available per week and scheduling grain traffic within the DTP (rather than the MTP).	We consider that the network management principles appropriately balance the need to provide access holders with certainty, with providing sufficient flexibility for Queensland Rail to be responsive to requests of certain customers. Moreover, the 2025 DAU recognises preserved train path obligations where these have been established.
GrainCorp considered that these arrangements failed to protect grain users at times when only a portion of the network capacity was available, as coal trains were programmed in the MTP and Queensland Rail prioritised trains scheduled in the MTP. ²³³	Providing additional flexibility to grain users in the 2025 DAU at times of congestion on the network may impact the certainty and flexibility provided to other access holders, including coal users. While preserved pathing arrangements have the potential to provide grain users with greater certainty and flexibility, such arrangements are established under the <i>Transport Infrastructure Act</i> 1994. ²³⁴

6.1.1 Network control

Network control facilitates the safe running of train services, and the commencement and completion of possessions, as scheduled in the DTPs. To execute this task, the network control principles provide network control officers with some discretion to give a train service priority over other train services if it is reasonably necessary to do so:

- due to, or to avoid, an accident, emergency or incident
- to remedy, mitigate or avoid congestion on the network
- to remedy, mitigate or avoid emergency and urgent possessions
- to ensure the safe operation of any part of the network.²³⁵

In relation to network controllers' ability to give priority to a train service to address congestion on the network, Aurizon Coal and Bulk considered it was unclear in what circumstances Queensland Rail used this provision to vary from the network management principles (and how effective it had been in addressing potential congestion). Aurizon Coal and Bulk said this provision gave Queensland Rail broad discretion, which it could use as an excuse for decisions that contravened other criteria.²³⁶

²³² GrainCorp submitted that grain trains must operate at different times of the day due to the geographically variable origins across the grain supply chain, combined with the impacts of network limitations like summer heat restrictions during peak grain export demand.

²³³ GrainCorp, sub. 4, p. 5.

²³⁴ See section 266A.

²³⁵ 2025 DAU, Sch. F, cl. 3(i)(i).

²³⁶ Aurizon Coal and Bulk, sub. 2, p. 39.

More broadly, Aurizon Coal and Bulk and the Rail Operator Group considered that the 2025 DAU did not always provide transparent guidance on how network controllers were expected to make their decisions.²³⁷

In this regard, the Rail Operator Group submitted that it was not evident how much consideration Queensland Rail was giving freight rail services in managing deviations from the DTP.²³⁸ The Rail Operator Group considered that enhancing Queensland Rail's train control decision matrix would set consistent and objective rules, as well as give the train controllers and rail operators certainty over their operations.²³⁹ Pacific National also considered that the 2025 DAU should establish a priority matrix that set out passenger and freight services at different times of the day and on weekends.²⁴⁰

The train control principles rightly prescribe balanced and transparent decision-making rules and principles that network controllers are to apply to facilitate the safe and efficient running of train services. However, we consider it necessary that train control principles also provide network train controllers with sufficient discretion to exercise judgement in performing this task. Prescribing additional rules or removing the ability of network controllers to exercise discretion may limit the effectiveness of network control to address network constraints as they arise.

We consider that providing network controllers with the ability to give priority to a train service in circumstances that remedy, mitigate or avoid congestion on the network is reasonable. This will help to avoid, or reduce, delays to the running of train services, which is in the interests of access holders. Importantly, in performing the task of network control, we do not consider that Queensland Rail has an incentive to favour any particular freight access holder over another.²⁴¹

We also consider that outcomes of decisions made by network control are sufficiently transparent for access holders. The 2025 DAU requires Queensland Rail to provide an access holder with real-time network control information that indicates actual running of the access holder's train services against the relevant DTP.²⁴² Moreover, Queensland Rail is required to report various measures relating to the operational management of the network (see chapter 7).

In relation to passenger services, processes for allocating capacity to passenger services and for passenger priority obligations are prescribed by the *Transport Infrastructure Act* 1994.²⁴³ The legislation includes an obligation for the railway manager to:

- bring delayed passenger train services back to their scheduled running time²⁴⁴
- allocate rail capacity that is available, or will become available, to meet the requirements for regularly scheduled passenger train services, as set out by the director-general of the Department of Transport and Main Roads (DTMR).²⁴⁵

We consider that prescribing further rules in the 2025 DAU network control principles could adversely affect Queensland Rail's ability to effectively manage its legislative obligations.

Queensland Rail submitted that all of Queensland Rail's passenger train services used either preserved train paths or train paths that were the subject of the passenger train service requirements of the DTMR's director-general. Queensland Rail considered that it was not

²³⁷ Aurizon Coal and Bulk, sub. 2, p. 39; Rail Operator Group, sub. 15, pp. 8-9.

²³⁸ The Rail Operator Group considered that rail operators were significantly impacted by Queensland Rail's operating choices to prioritise passenger operations in pathing decisions.

²³⁹ Rail Operator Group, sub. 15, pp. 8-9.

²⁴⁰ Pacific National, sub. 7, p. 13.

²⁴¹ Queensland Rail is not vertically integrated into above-rail freight operations and therefore cannot favour a related party. ²⁴² 2025 DAU, Sch. F, cl. 3(e).

²⁴³ Queensland Rail, sub. 14, p. 51.

²⁴⁴ Transport Infrastructure Act, s. 265.

²⁴⁵ Transport Infrastructure Act, s. 266A.

responsible for identifying passenger service requirements, nor for the effects of the preserved train path obligations.²⁴⁶

6.1.2 Obligations following a dispute on a planned possession

Queensland Rail proposed not to include in the 2025 DAU a requirement that any planned possession subject to a dispute raised by an access holder is to be delayed until the dispute is resolved. The 2020 AU includes that requirement.²⁴⁷

Aurizon Coal and Bulk, Glencore, GrainCorp, New Hope, Pacific National, the Rail Operator Group and Yancoal opposed Queensland Rail's proposal to not include such a provision in the 2025 DAU.²⁴⁸ A number of stakeholders said they had experienced an increase in possessions in recent years.²⁴⁹

We do not consider it necessary for the 2025 DAU to include a requirement for Queensland Rail to delay implementing any planned possession that is subject to a dispute.

Including such a requirement has the potential to lead to inefficiencies and disruptions to the running of the network in circumstances where an access holder raises a dispute just before the start of a planned possession.

Queensland Rail considered that the 2025 DAU had a dispute mechanism that applied equally to all relevant matters covered by an approved undertaking, and it is this mechanism that should apply to prevent overregulation.²⁵⁰

In contrast, Aurizon Coal and Bulk and the Rail Operator Group considered that delaying a possession until a related dispute was resolved assisted in promoting disciplined operation of the network, and accountability on Queensland Rail to comply with its obligations to the access holder. Aurizon Coal and Bulk also considered that the requirement to delay any disputed possession ensured there was no incentive for Queensland Rail to delay resolution of a dispute in order to allow the possession to proceed.²⁵¹ Pacific National and the Rail Operators Group considered that the removal of this requirement would appear to be an erosion of access holders' rights when a bona fide dispute took place.²⁵²

Importantly, the 2025 DAU does not prevent access holders from raising a dispute in relation to Queensland Rail complying with the network management principles. As outlined in section 3.4, we consider that the proposed dispute resolution mechanism in the 2025 DAU is sufficient to hold Queensland Rail accountable for its conduct under the access undertaking.

In scheduling train services, the 2025 DAU requires Queensland Rail to use reasonable endeavours to minimise any material adverse effects on train services caused from scheduling an ad hoc possession, or any modification or variation to an MTP.²⁵³ Queensland Rail must also use its

²⁴⁶ Queensland Rail, sub. 14, pp. 51-53.

²⁴⁷ 2020 AU, sch. F, cl. 2.4.

²⁴⁸ Aurizon Coal and Bulk, sub. 2, p. 38; Glencore, sub. 11, p. 7; GrainCorp, sub. 4, p. 3; New Hope, sub. 5, pp. 28-29; Pacific National, sub. 7, pp. 23-24; Rail Operator Group, sub. 15, p. 2; Yancoal, sub. 9, pp. 24-25.

²⁴⁹ Pacific National also submitted that it had experienced an increase in possessions and full system closures on the North Coast line, as well as longer closure durations (Pacific National, sub. 7, pp. 23–24). New Hope and Yancoal submitted that the West Moreton system had been beset by possessions in recent times (New Hope, sub. 5, pp. 28-29; Yancoal, sub. 9, pp. 24-25).

²⁵⁰ Queensland Rail, sub. 1, p. 64. Queensland Rail submitted that no other rail infrastructure provider in Australia was subject to this requirement.

²⁵¹ Aurizon Coal and Bulk, sub. 2, p. 38; Rail Operators Group, sub. 15, pp. 9-10.

²⁵² Pacific National, sub. 7, p. 23; Rail Operators Group, sub. 15, pp. 9-10.

²⁵³ 2025 DAU, Sch. F, cl. 2.3(a).

reasonable endeavours to offer any access holder that is affected by a modification or variation an alternative schedule time.²⁵⁴

To date, there have been no disputes in relation to the way that Queensland Rail has scheduled planned possessions. We consider that the provisions in the 2025 DAU will provide sufficient incentives for Queensland Rail to comply with its obligations outlined in the network management principles.²⁵⁵

However, we will continue to monitor whether the regulatory arrangements affect Queensland Rail's performance in implementing planned possessions throughout the AU3 regulatory period. Should Queensland Rail not comply with its obligations outlined in the network management principles and this has implications for access holders, further amendments to the regulatory arrangements may be necessary in future. As such, we consider it important that the reporting obligations in the 2025 DAU provide sufficient transparency in relation to Queensland Rail's performance in implementing planned possessions. The appropriateness of Queensland Rail's proposed reporting obligations is discussed in chapter 7.

In scheduling planned possessions, Queensland Rail should be responsive to the needs of customers where possible.²⁵⁶ We consider that effective engagement with access holders prior to scheduling a planned possession will promote the efficient operation of the Queensland Rail's network.

Pacific National considered that the 2025 DAU would reduce Queensland Rail's consultation with access holders and operators. Pacific National considered that there should be more transparency and incentives around consultative procedures before the network owner takes possession of the railway.²⁵⁷

The 2025 DAU requires Queensland Rail to consult with affected parties where Queensland Rail schedules a DTP in variation to a MTP or makes a modification to a scheduled DTP.²⁵⁸ This provides for cooperation between parties, where possible, to realise the most suitable and efficient scheduling outcomes for those affected.

The 2025 DAU also establishes processes that provide for Queensland Rail to effectively engage with the relevant parties as part of the planning process. In this regard, Queensland Rail must convene regional network user groups for the West Moreton system, North Coast line and Mount Isa line (see section 6.3).²⁵⁹ It is not clear that not including a requirement to delay a planned possession subject to a dispute will affect these consultation processes. However, as outlined, we will continue to monitor this matter throughout the AU3 regulatory period.

²⁵⁴ 2025 DAU, Sch. F, cl. 2.3(c).

²⁵⁵ In our view, the dispute resolution mechanism in the 2025 DAU holds parties accountable for their conduct under the access undertaking.

²⁵⁶ A planned possession has the potential to significantly disrupt train services and thereby negatively affect access holders and operators. In this regard, Pacific National (sub. 7, p. 23) submitted that where the MTP is changed without agreement, the financial and operational impact to a rail operator can be substantial and can cause significant impact to the end customer.

²⁵⁷ Pacific National, sub. 7, p. 23. Pacific Rail considered that rail operators needed a level of protection and a consultation process with Queensland Rail to influence closures and suggest alternative timeframes to undertake works, particularly where they could be aligned with closures on other networks. Specifically, strong consultation was needed to ensure industry provided advice on seasonality to ensure possessions were not imposed during the sugar season and freight peaks.

²⁵⁸ 2025 DAU, cls. 2.2(e)-(h).

²⁵⁹ Subject to the continual support of the respective access holders and operators (2025 DAU, cl. 4.4(a)).

Summary 6.1

It is not appropriate to approve Queensland Rail's proposed network management principles (cls. 4.1 and 4.2, and Schedule F of the 2025 DAU).

It is appropriate for Queensland Rail to amend the 2025 DAU to provide for Queensland Rail to be more responsive to requests from operators to change the MTP by:

- reducing the timeframe within which an access holder must notify Queensland Rail of a request to change the MTP to one month, where a request to modify an MTP, or a planned possession, does not impact other access holders
- including a process in the network management principles to incorporate changes to contracted paths in between periodic revisions of the MTP, where Queensland Rail only periodically revises the MTP on set dates.

6.2 Operating requirements manual

The proposed operating requirements manual sets out the rules and procedures for use of the network by train operators to enable Queensland Rail to manage its network in a way that it considers safe and efficient.

The operating requirements manual includes practices, standards, systems, protocols and rules relating to network control and access to, and use of, the network by train operators.²⁶⁰ The operation requirements outlined in Schedule G of the 2025 DAU are common across the network and are not subject to individual variation between different access agreements. They address, amongst other things:

- interface risk management, including environmental risk management
- safe working procedures and safety standards
- incident and emergency response procedures
- various technical requirements for train control and network planning.

In general, Queensland Rail's proposed operating requirements manual reflects the operating requirements manual approved as part of its 2020 access undertaking. Queensland Rail has proposed a number of amendments that we consider are administrative in nature, including updated references to legislation, standards, terminology, external documents and contact details. Queensland Rail also proposed minor amendments to the process for investigating Category A and Category B rail safety incidents.

We consider that Queensland Rail's proposed amendments do not have material adverse implications for access holders.

We did not receive any stakeholder submissions in relation to the operating requirements manual.

²⁶⁰ GrainCorp (sub. 4, p. 2) submitted that it was concerned that the 2025 DAU did not recognise access seekers that were not accredited operators under the Rail Safety National Law. We consider it appropriate for rail operators that use the network to be accredited rail transport operators (as defined in the Rail Safety National Law). This means that rail operators are capable of performing their functions and obligations under a SAA, which provides for the safe and efficient operation of the network.

Queensland Rail's 2025 Draft Access Undertaking

Summary 6.2

It is appropriate to approve Queensland Rail's proposed operating requirements manual (cl. 4.3 and Schedule G of the 2025 DAU).

6.3 User groups

The 2025 DAU requires that Queensland Rail convene regional network user groups for the West Moreton system, North Coast line and Mount Isa line, subject to the support of the respective access holders and operators.²⁶¹

The regional user groups are to be cooperative groups, with emphasis on analysis of data, open and impartial discussion and consensus decision-making to improve the operation of the supply chain.²⁶² We consider that regional user groups provide for Queensland Rail to:

- effectively engage with access holders and operators to collectively identify and introduce improvements in system and supply chain performance
- be more responsive to requests of customers in scheduling train services, as well as investing in and undertaking maintenance activities on its network.

Furthermore, we consider that regional user groups have the potential to identify and implement productivity and operational improvements across the network. As such, we consider that a requirement to convene and support regional user groups is in the interests of access holders, access seekers and Queensland Rail.

A number of stakeholders supported arrangements that provided further consultation and collaboration with access holders and operators to identify opportunities for further improvement. Qube considered that Queensland Rail needed to work with rail operators to understand the opportunities for broader industry improvement.²⁶³ Specifically, Centrex and Mount Isa Line Users supported establishing user groups to drive operational and capital improvements.²⁶⁴

6.3.1 Specific measures to promote productivity

Certain stakeholders raised concerns that the 2025 DAU did not include specific measures to promote productivity and operational improvements.²⁶⁵

Moreover, a number of stakeholders identified specific measures that they considered should be included in the 2025 DAU in order to promote productivity and operational improvements.

• Pacific National considered that the 2025 DAU should include a commitment to publish an updated North Coast Rail Line Capacity Improvement plan and an agreement to consult on

²⁶¹ 2025 DAU, cl. 4.4(a).

²⁶² 2025 DAU, cl. 4.4(b).

²⁶³ Qube, sub. 8, p. 6.

²⁶⁴ Centrex, sub. 10, pp. 3-4; Mount Isa Line Users, sub. 17, p. 3.

²⁶⁵ Pacific National (sub. 7, p. 3) considered that the 2025 DAU was locking in a system of complacency, rather than embracing the continual improvement and commitment needed to address change and critical issues of resilience and efficiency. North West Phosphate (sub. 6, p. 1) considered that limited innovation and improvement on the Mount Isa line translated to poor quality of service and performance. GrainCorp (sub. 4, p. 3) submitted that investment in, reliability and productivity of the rail infrastructure needed to be brought to the fore more prominently than it currently was.

investment plans with access holders. In relation to the North Coast line, Pacific National and the Rail Operator Group submitted that Queensland Rail needed an action plan that took a long-term view on investment, demand, capacity, and resilience.²⁶⁶

- Pacific National considered that the 2025 DAU should incorporate mechanisms to drive data accuracy, efficiency and improved transparency, including providing accurate, real-time data.²⁶⁷
- GrainCorp considered that the 2025 DAU could provide a stronger incentive for Queensland Rail to make its track detection equipment used to enforce compliance with asset protection standards fit for purpose and appropriately accurate.²⁶⁸ GrainCorp submitted that users significantly underloaded all grain train services to avoid the delays and costs caused by false positive detections, given installing load-site weigh systems at every grain loading site was economically unviable. Furthermore, GrainCorp considered that there was no urgency or requirement for Queensland Rail to install an appropriately precise and accurate weighbridge.²⁶⁹
- Aurizon Coal and Bulk considered that the 2025 DAU should enable access seekers to register their interest in acquiring a new or varied train path if it became available as a result of path rescheduling or resumption. Aurizon Coal and Bulk considered that the 2025 DAU should also require Queensland Rail to provide interested access seekers with capacity information to allow the access seeker to assess opportunities for path resumption or rescheduling.²⁷⁰

We encourage identifying and implementing measures that will realise productivity and operational improvements across Queensland Rail's network.

However, it is important to consider and understand the costs and benefits of any proposed investments or operational improvements to provide for the efficient operation of, and investment in, the network.

Queensland Rail will incur costs from investing in operational improvements, or from having additional planning, reporting or monitoring obligations because of measures that are introduced within the regulatory framework. We consider it is in Queensland Rail's legitimate business interests to generate expected revenue to meet the efficient costs of providing access to its network.

System-specific issues, such as developing an action plan for the North Coast line, require systemspecific solutions rather than overarching obligations in an approved access undertaking. Relevantly, the network and customer characteristics for each system vary significantly.

While it is not clear to what extent the productivity and operational improvements proposed by certain stakeholders will benefit other access holders across the network, such measures have potential to impose additional costs on these access holders.²⁷¹ We also note that many of stakeholders raised concerns about the affordability of access to Queensland Rail's network.

²⁶⁶ Pacific National, sub. 8, pp. 8-9; Rail Operator Group, sub. 15, pp. 1, 4-5. The Rail Operator Group supported including commitments to developing long-term investment strategies for key freight and resource corridors, in consultation with users.

²⁶⁷ Pacific National, sub. 13, p. 12.

²⁶⁸ GrainCorp submitted that Queensland Rail used a number of devices to monitor train weights that were not designed or certified for accurate weighing of wagons.

²⁶⁹ GrainCorp, sub. 4, p. 5.

²⁷⁰ Aurizon Coal and Bulk, sub. 2, pp. 50-51.

²⁷¹ For example, New Hope (sub. 12, pp. 3-4) submitted that it already installed sufficiently accurate weighing devices at its own loading facility and was not in a position to support contributing costs to additional mainline devices. New Hope also considered that caution should be exercised when considering the installation and maintenance of more accurate mainline weighing devices, as the associated track outages could be significant.

We therefore consider that effective consultation with affected parties is an important part of the process in identifying, assessing and implementing efficient operational improvements to the network. We consider that regional user groups are well placed to provide for this.

To support regional user groups, the 2025 DAU requires Queensland Rail to, amongst other things:

- provide analysis of the root causes of ongoing or systematic issues being experienced
- identify resolutions to such issues and other productivity or efficiency initiatives
- provide evaluation and modelling of the outcomes of potential supply chain operational changes that the regional network user group supports investigating.²⁷²

Aurizon Coal and Bulk submitted that the 2025 DAU did not create any accountability on Queensland Rail's compliance with its commitment to support regional network user groups.²⁷³ New Hope considered that from its observation, Queensland Rail was not providing adequate resources for data collection and analysis to meet the needs of the South West User Group.²⁷⁴

We consider the dispute resolution mechanism in the 2025 DAU holds parties accountable for their conduct under the access undertaking.

Furthermore, now that the regional user groups have been established in the various systems, we will continue to monitor whether Queensland Rail is fulfilling its obligations to support these groups. Should regional user groups not be effective in identifying and implementing measures that realise productivity and operational improvements across Queensland Rail's network, further amendments to the regulatory arrangements may be necessary in future. However, we note that the effective functioning of the regional user groups is dependent on the support of all relevant parties.

6.3.2 Further obligations to respond to operational requests

Pacific National considered that the 2025 DAU should include obligations for Queensland Rail to respond to rail operators in a more timely way. Pacific National submitted that there were no mandated timeframes in the 2025 DAU for Queensland Rail to respond to rail operator requests designed to improve efficiency.²⁷⁵

Pacific National submitted that it had a number of high impact operational requests outstanding with Queensland Rail as of March 2024, most of which required Queensland Rail engineering review. Pacific National provided two examples of requests that it considered had not been addressed in a timely manner by Queensland Rail:

- In July 2023, Pacific National requested to use Queensland Rail-owned equipment, which it offered to pay for. It considered that would significantly increase the efficiency of its operations. Pacific National submitted that this request had still not been implemented, despite Queensland Rail indicating that it was willing to enter into an agreement to facilitate use of the equipment.
- In September 2023, Pacific National approached Queensland Rail's engineering team to approve the use of wagons (currently in storage) for a specific shuttle service between Pacific National's Townsville terminal and the Townsville jetty. Pacific National submitted that

²⁷² 2025 DAU, cl. 4.4(c).

²⁷³ Aurizon Coal and Bulk, sub. 2, p. 20.

²⁷⁴ New Hope, sub. 12, p. 3. New Hope submitted that it supports Queensland Rail's revenue allowances providing sufficient resources for this critical function.

²⁷⁵ Pacific National, sub. 13, pp. 5-6.

Queensland Rail has been unable to engage on this issue in a meaningful way until very recently due to resourcing constraints.²⁷⁶

In contrast, Queensland Rail submitted that it strove to work with access seekers and end users to advance the use of the network.²⁷⁷

From the information available, it is not clear that the 2025 DAU should include further obligations that prescribe the time in which Queensland Rail should respond to rail operators in order to improve service and responsiveness for their end customers.

Importantly, Queensland Rail's regulatory framework provides for access seekers to obtain access to Queensland Rail's network on reasonable terms and conditions. Moreover, we consider that the negotiation framework in the 2025 DAU facilitates access seekers and Queensland Rail to negotiate the terms and conditions for access to Queensland Rail's network (see chapter 3).

Within this regulatory framework, we consider that Queensland Rail is incentivised to negotiate mutually beneficial terms of access. In this regard, Queensland Rail considered that it was incentivised to assist access seekers and users, as additional usage of the network contributed to cost recovery of largely fixed capital costs of the network. Queensland Rail provided a number of case studies to demonstrate its commitment to work with access seekers and users to achieve mutually beneficial outcomes.²⁷⁸

Imposing broad obligations in the 2025 DAU requiring Queensland Rail to respond to requests from operators may be problematic, because efficient response times depend on the specific request. Obligations to respond to operators' requests could also present resourcing challenges for Queensland Rail, given the infrequent nature of requests. Requests from operators my vary significantly, as noted in the examples provided by Pacific National. Furthermore, it is not clear that all requests are related to Queensland Rail's operation of the below-rail infrastructure.

In any event, it is not evident to us that there are systemic problems with Queensland Rail's existing response to specific requests.

Summary 6.3

It is appropriate to approve Queensland Rail's proposed obligations to provide for productivity and operational improvements (cl. 4.4 of the 2025 DAU).

 ²⁷⁶ Pacific National, sub. 7, pp. 5-6.
 ²⁷⁷ Queensland Rail, sub. 14, pp. 27-29.
 ²⁷⁸ Queensland Rail, sub. 14, pp. 27-29.

7 Reporting requirements

Queensland Rail's 2025 DAU sets out how Queensland Rail is to report its performance in managing the network.

Overview of the draft decision

We consider it is appropriate for Queensland Rail to make some amendments to the reporting requirements in the 2025 DAU.

Reporting requirements (Part 5) – summary

Queensland Rail proposal	Clause	QCA draft decision
Performance reporting		
Queensland Rail proposed not to include some reporting, including on deviations from the DTP, and some performance indicators.	Part 5	Appropriate to approve. See section 7.2
Reporting on ad hoc planned possessio	ns	
Queensland Rail proposed not requiring it to report on aspects of ad hoc planned possessions.	cl. 5.1	It is appropriate to require quarterly reporting on ad hoc planned possessions. See section 7.3.

7.1 Overview of reporting obligations in the 2025 DAU

Part 5 of the 2025 DAU outlines Queensland Rail's reporting responsibilities. Queensland Rail proposed reporting arrangements that require it to prepare quarterly and annual performance reports to inform parties about its performance in operating the network and negotiating access, as well as the costs associated with providing access to parts of the network (see Figure 6). Queensland Rail is also required to prepare annual financial statements, publishing the accounting records for its below-rail services.²⁷⁹

Part 5 also provides rules for auditing Queensland Rail's compliance with the provisions in the 2025 DAU. Acting reasonably, we may require an audit of compliance with any aspect of the undertaking, or request Queensland Rail to provide information required for the purpose of complying with the undertaking.²⁸⁰

²⁷⁹ 2025 DAU, cl. 5.3.
²⁸⁰ 2025 DAU, cl. 5.4.

Figure 6: Queensland Rail's quarterly and annual network performance reports

Quarterly network performance report	Annual network performance report
Reports various measures relating to Queensland Rail's operational management of the network, including:	Reports various measures across the access negotiation process, including:
 on-time running of train services 	capacity information requestsaccess applications
transit time delays	• indicative access proposals
train cancellations	negotiation cessation notices
possessions – planned, urgent and emergencytrack under temporary speed restrictions	 access agreements executed and timeframes for negotiation
overall track condition index (OTCI)major reportable safety incidents	 disputes referred to the dispute resolution process
 verified written complaints by access holders. 	 maintenance costs, operating expenditure and capital investment for each regional network
	• volumes of train services for each regional network.

Source: 2025 DAU, cls. 5.1 and 5.2.

7.2 Performance reporting

Queensland Rail's proposed reporting obligations provide interested parties with information on how efficiently Queensland Rail has been performing and confidence that it is complying with an approved access undertaking. As such, we consider that these reporting obligations will assist to promote the economically efficient operation of, use of and investment in the network.

The proposed quarterly reporting obligations in the 2025 DAU reflect the type of information produced in the quarterly reports during the AU2 regulatory period.²⁸¹ However, Queensland Rail's proposed minor changes to the quarterly reporting obligations in the 2020 AU. In this regard, the 2025 DAU does not require Queensland Rail to report:

- the number of times that the network control officer made a decision to deviate from a DTP to remedy, mitigate or avoid network congestion²⁸²
- aggregate information on temporary speed restrictions, overall track condition and possessions by train type for the various systems of the network²⁸³
- ad hoc planned possession start and end times, as well as the number and duration of these possessions for the quarter²⁸⁴ (see section 7.3).

Aurizon Coal and Bulk, Glencore, New Hope, Pacific National and Yancoal all opposed Queensland Rail's proposal to remove the above reporting requirements. In particular, these stakeholders

²⁸¹ The 2025 DAU requires the quarterly report be published within 6 weeks of the end of the quarter, instead of within 4 weeks. Queensland Rail considered that a 4-week timeframe is not sufficient given the complexity of the current quarterly reporting requirements (Queensland Rail, sub. 1, pp. 60-62.). New Hope, Yancoal and Pacific National did not object to Queensland Rail's proposed amendments to the timeframes for delivering the quarterly report (New Hope, sub. 5, p. 27; Yancoal, sub. 9, p. 23; Pacific National, sub. 7, p. 15-16).

²⁸² Queensland Rail submitted that its Vizirail system did not have the ability to record this KPI. See Queensland Rail, sub. 1, pp. 62-63.

²⁸³ Queensland Rail (sub. 1, p. 63) submitted that this information could not be reported by train type, and this requirement was an error in the 2020 AU.

²⁸⁴ Queensland Rail (sub. 1, p. 63) submitted that its systems did not have the capacity to report on ad hoc possessions – and its Vizirail system would require expensive enhancements to record start and end times for these possessions. Queensland Rail also considered that ad hoc possessions only had a minor effect on delays.
considered that further accountability and transparency was important in relation to Queensland Rail's decision to implement ad hoc planned possessions and to deviate from a DTP.²⁸⁵

Pacific National considered there was a disappointing theme of information asymmetry and reduced transparency for rail operators.²⁸⁶ GrainCorp was concerned there had been a backwards step in Queensland Rail's proposed standard of reporting.²⁸⁷

In contrast, Queensland Rail considered the existing reporting arrangements were robust and characterised by a notably high level of openness and transparency.²⁸⁸

Effective reporting and compliance monitoring underpin the integrity of the access regime, as they place accountability on Queensland Rail's performance and provide for greater levels of transparency. However, we consider the benefits obtained from reporting and compliance monitoring need to be balanced with the regulatory burden that reporting obligations may impose on Queensland Rail.

Queensland Rail submitted that further regulatory requirements to provide additional information may be an administrative burden on Queensland Rail and impede operational efficiency.²⁸⁹

While the proposed reporting obligations will provide interested parties with information to assess Queensland Rail's performance in delivering train services, we recognise that certain access holders may benefit from the reporting of additional information. In this regard, a number of stakeholders submitted that requiring Queensland Rail to provide the following information would enable further analysis of system performance or inform long-term network planning:

- the number of times that the network control officer deviated from a DTP²⁹⁰
- information on ad hoc planned possessions²⁹¹
- modelling on future passenger service growth and the likely impacts for rail freight.²⁹²

A greater understanding of system performance may assist access holders to identify potential areas for investment or operational improvement, particularly as systems become more congested. However, at this time, we consider that providing further information to better analyse system performance or to inform long-term planning should be further considered at the respective regional user groups.

²⁸⁵ Aurizon Coal and Bulk, sub. 2, p. 43-44; New Hope, sub. 5, p. 27; Glencore, sub. 11, p. 7; Yancoal, sub. 9, pp. 23-24; Pacific National, sub. 7, p. 17-19.

²⁸⁶ Pacific National, sub. 7, p. 3.

²⁸⁷ GrainCorp, sub. 4, p. 7.

²⁸⁸ Queensland Rail, sub. 14, pp. 48-50. Queensland Rail also submitted that regular reports on service performance and network operations were routinely disseminated to stakeholders, ensuring transparency and accountability. Moreover, Queensland Rail stated that it was also currently bound by obligations to provide detailed information on proposed access charges within the negotiation process and was mandated to disclose commercial access pricing to other operators. ²⁸⁹ Queensland Rail, sub. 14, pp. 48-50.

²⁹⁰ Pacific National (sub. 7, pp. 17-18) submitted that operators needed visibility of whole-of-network delays and cancellations in order to interrogate root causes and understand what it meant for network resilience and efficient use of the network. Moreover, Pacific National considered that with greater congestion on the network, deviations from the DTP were likely to increase. Pacific National submitted that removing this reporting requirement would increase the information asymmetry between rail operators and Queensland Rail.

²⁹¹ Pacific National (sub. 7, pp. 18-19) considered that such information assisted it to assess and understand whether Queensland Rail's maintenance planning process was efficient. Pacific National considered that any trend towards increased ad hoc planned possessions raised concerns about Queensland Rail's maintenance planning process. Pacific National also considered that this information enabled it to reconcile where a train had been cancelled due to ad hoc possessions or related impacts.

²⁹² Pacific National (sub. 7, p. 13) and the Rail Operator Group (sub. 15, p. 1) considered that the 2025 DAU should include a requirement for Queensland Rail to provide modelling on future passenger service growth and the likely impacts for rail freight, to inform long-term network planning. In this regard, Pacific National submitted that both the Queensland freight task and demand for passenger rail services were expected to increase significantly. Pacific National considered that increased transparency about how Queensland Rail would make capacity decisions would be required.

Queensland Rail will incur additional costs where additional reporting obligations are incorporated in the regulatory framework. For instance, Queensland Rail submitted that its Vizirail system did not have the ability to record the number of times that the network control officer made a decision to deviate from a DTP. Moreover, Queensland Rail considered that it was not appropriate to impose this administrative burden on train controllers, as they needed to be fully focused on the task of network control.²⁹³ While there may be potential solutions for obtaining and reporting this information²⁹⁴, obligations to report this information will impose costs on Queensland Rail.

Moreover, the extent that end users value additional performance information may vary, especially because network and customer characteristics for each system vary significantly. For instance, not all systems will suffer from increasing congestion during the upcoming regulatory period. We consider that regional user groups are better placed to identify and address system-specific issues than overarching obligations in an access undertaking.

This information was also not reported during the AU2 regulatory period. We consider that the reporting obligations in the 2025 DAU will provide interested parties with adequate information on how efficiently Queensland Rail has been performing, while balancing Queensland Rail's legitimate concerns about additional compliance costs.

New Hope submitted that its preference was to work with the regional user group so that the type of information needed to understand system performance and drive improvements was made available, rather than prescribe further reporting requirements in the 2025 DAU.²⁹⁵

7.3 Further reporting on ad hoc planned possessions

We do not consider it is necessary for the 2025 DAU to include a requirement for Queensland Rail to delay implementing any planned possession that is subject to a dispute (see section 6.1 of this draft decision). However, it is important that the reporting obligations in the 2025 DAU provide sufficient transparency in relation to Queensland Rail's performance in implementing planned possessions. This is necessary so to hold Queensland Rail accountable for its performance in scheduling possessions and to promote disciplined operation of the network.²⁹⁶

The proposed reporting obligations in the 2025 DAU do not require Queensland Rail to report ad hoc planned possessions. We consider it is appropriate for the 2025 DAU to require that Queensland Rail report on ad hoc planned possessions as part of its quarterly report.

A number of stakeholders considered that ad hoc planned possessions had implications for rail operators.²⁹⁷ Aurizon Coal and Bulk considered that if Queensland Rail wanted to retain the right to

²⁹⁶ Scheduling possessions for the purpose of undertaking regular maintenance and capital works should be planned in advance to provide access holders with certainty as to the upcoming scheduling of train services.

²⁹³ Queensland Rail, sub. 1, pp. 62-63.

²⁹⁴ For instance, Pacific National (sub. 7, pp. 17-18) considered that Queensland Rail could seek a solution that automates most of this work for train controllers; or allocate dedicated time at the end of each shift so train controllers could document reasons for any deviation.

²⁹⁵ New Hope, sub. 12, pp. 2-3.

²⁹⁷ Aurizon Coal and Bulk (sub. 2, p. 42), New Hope (sub. 5, p. 27), Pacific National (sub. 7, p. 18) and Yancoal (sub. 9, p. 24) all questioned Queensland Rail's assertion that ad hoc planned possessions only had a minor effect on delays for rail operators. Pacific National submitted that the downstream impacts of any delay could be significant in terms of supply chain management, and ultimately the end customer suffered the consequences of such repercussions. Pacific National considered that such delays could also have labour impacts and could potentially cause a cancellation in subsequent services in instances where the rail operator had exhausted all available crew. GrainCorp (sub. 4, p. 3) considered there could be a significantly increased reliance on ad hoc planned possessions, which would further erode train paths for grain in the DTP.

apply ad hoc planned possessions, it was important that it was transparent and accountable in the way it used this right.²⁹⁸

Queensland Rail submitted that its Vizirail system would require expensive enhancements to record start and end times for ad hoc planned possessions.²⁹⁹

We acknowledge that imposing further reporting obligations may result in additional costs for Queensland Rail. We encourage stakeholders to discuss, collaborate on and, where possible, reach consensus on feasible options for reporting ad hoc planned possessions during the regulatory period. In this regard, Aurizon Coal and Bulk submitted that adopting a simplified approach to reporting on compliance with possession timeframes may be satisfactory.³⁰⁰

If it is not feasible to report information on ad hoc planned possessions, then alternative amendments may need to be made to the regulatory framework. For instance, it may be appropriate for the 2025 DAU to include a requirement for Queensland Rail to delay implementing any planned possession that was subject to a dispute, in order to hold Queensland Rail accountable for its performance in scheduling possessions (see section 6.1).

We will have regard to consensus positions reached amongst stakeholders as part of our assessment of Queensland Rail's 2025 DAU.

Summary 7.1

It is not appropriate to approve Queensland Rail's proposed reporting requirements (Part 5 of the 2025 DAU). It is appropriate for the 2025 DAU to require that Queensland Rail to report on ad hoc possessions as part of its quarterly report.

 ²⁹⁸ Aurizon Coal and Bulk, sub. 2, pp. 42-43.
 ²⁹⁹ Queensland Rail, sub. 1, pp. 62-63.
 ³⁰⁰ Aurizon Coal and Bulk, sub. 2, pp. 42-43.

Queensland Rail's 2025 Draft Access Undertaking

8 West Moreton tariff approach

The 2025 DAU covers all of Queensland Rail's declared services but only includes a proposed reference tariff for coal haulage on the West Moreton route service (the West Moreton reference tariff). The two systems that comprise the declared route service (West Moreton and Metropolitan systems) connect mines in southern Queensland with the coal export terminal at the Port of Brisbane.

In the 2025 DAU, Queensland Rail proposed a West Moreton coal reference tariff of \$32.63 per thousand gross tonne kilometres ('000 gtk), based on forecast volumes of 9.6 million tonnes a year (mtpa).

Overview of the draft decision

Our draft decision is that it is not appropriate to approve Queensland Rail's proposed reference tariff in Schedule D of the 2025 DAU. We are proposing some amendments that we consider may be appropriate to approve.

West Moreton tariff approach – summary

Queensland Rail proposal	Clause	QCA draft position
Reference tariff		
Queensland Rail has proposed a 24% tariff increase to \$32.63 per '000 gtk. ³⁰¹	Schedule D	Not appropriate to approve. See sections 8.1 and 8.8.
Assessing the proposed reference tariff		
Queensland Rail said it aimed to meet the requirements of access holders. ³⁰²		The best basis for an efficient tariff is a reasoned and prudent strategy for West Moreton that has preferably been agreed with customers. See section 8.2.
Capital spending: indicator and approve	al	
Queensland Rail has proposed a capital expenditure review and approval process that is unchanged from the process in the 2020 undertaking. ³⁰³	Schedule E	It may be appropriate to amend the capital expenditure approval process to provide for more engagement with customers. See section 8.3.
Capacity assessment		
Queensland Rail said the West Moreton system had 97 weekly paths available to contract for coal-carrying services. ³⁰⁴		It may be appropriate to amend the 2025 DAU to include an independent capacity review process. See section 8.4.

³⁰¹ Queensland Rail, sub. 1, p. 12. The increase is compared with the price customers have been paying under the 2020 undertaking. The 2020 undertaking also included a higher ceiling price that was used to calculate loss capitalisation amounts.

³⁰² Queensland Rail, sub. 1, pp. 9-10.

³⁰³ Queensland Rail, sub. 1, p. 23.

³⁰⁴ Queensland Rail, sub. 1, p. 7.

Risk mitigation		
Queensland Rail proposed accelerated depreciation for West Moreton assets. ³⁰⁵		It may be appropriate to approve accelerated depreciation as part of a balanced risk-sharing approach. See sections 8.5.1 and 8.5.5.
Queensland Rail proposed investment triggers based on contract renewals. ³⁰⁶	Sch. D, cl. 3.2	It may be appropriate to approve investment triggers as part of a balanced risk-sharing approach. See sections 8.5.2 and 8.5.5.
Stakeholders proposed capital expenditure reconciliation. ³⁰⁷		It may be appropriate to approve capital expenditure reconciliation as part of a balanced risk-sharing approach. See sections 8.5.3 and 8.5.5.
Stakeholders proposed renewal rights. ³⁰⁸		It may be appropriate to approve renewal rights as part of a balanced risk-sharing approach. See sections 8.5.4 and 8.5.5.
Tariff structure		
Queensland Rail proposed a two-part tariff, split between a train path charge and a weight and volume-based charge.	Schedule. D, cl. 3.1	Appropriate to approve. See section 8.6.
New Hope said its private balloon loop should be covered by a rebate mechanism. ³⁰⁹ Yancoal disagreed. ³¹⁰		It is appropriate to amend the 2025 DAU so that treatment of single-user infrastructure is equitable. See section 8.6.
Loss capitalisation and the 2020 underta	aking	
Queensland Rail said it expected to detail its proposed loss capitalisation treatment in a DAAU to amend the 2020 undertaking tariff. ³¹¹	Sch. D, cl. 8	The mechanism for addressing capitalised losses is best negotiated between Queensland Rail and its customers. See section 8.7.
Appropriate reference tariff		
Queensland Rail proposed a reference tariff based on annual volumes rising to 9.6 million tonnes.		The proposed reference tariff is not appropriate to approve. See section 8.8.

³⁰⁵ Queensland Rail, sub. 1, pp. 12, 30.
³⁰⁶ Queensland Rail, sub. 1, pp. 12, 30.
³⁰⁷ New Hope, sub. 5, pp. 26, 36-38; Yancoal, sub. 9, pp. 15, 28-29 and sub. 16, p. 7.
³⁰⁸ New Hope, sub. 5, p. 26; Yancoal, sub. 9, pp. 20-21.
³⁰⁹ New Hope, sub. 5, p. 13.
³¹⁰ Yancoal, sub. 16, pp. 5-6.
³¹¹ Queensland Rail, sub. 1, p. 55.

8.1 Queensland Rail's proposal

Queensland Rail proposed a West Moreton reference tariff of \$32.63/ '000 gtk, or \$12.63 per net tonne³¹², which is 24% higher than the price of \$26.42/ '000 gtk (estimated July 2025 dollars) that customers have been paying under the 2020 undertaking.³¹³ The proposed tariff is based on:

- annual volumes rising more than fourfold, from 2.1 million tonnes to 9.6 million tonnes³¹⁴
- forecast capital spending (the 'capital indicator') more than doubling, to \$346.9 million over the five-year undertaking period
- a weighted average cost of capital (WACC) of 7.39%, up from 5.46%
- a 61% increase in annual maintenance spending, to an average of \$34 million a year, and a near doubling of operating spending
- accelerated depreciation over 14 years for new assets and 19 years for existing assets, based on Queensland Rail's forecast of mine lives
- a trigger mechanism that enables Queensland Rail to revise capital spending if customers do not contract for forecast volumes.

Stakeholders said the proposed West Moreton reference tariff was not affordable. They said:

- Queensland Rail had not consulted sufficiently on the tariff proposal³¹⁵
- Queensland Rail had not substantiated cost forecasts including capital, maintenance and operating spending, or other aspects of the price proposal, including accelerated depreciation and WACC³¹⁶
- the DAU should include a capital prudency process that gives users a vote on which projects are built³¹⁷
- the DAU should include an independent mechanism to assess the current capacity of the network and determine whether the proposed spending will deliver the promised capacity.³¹⁸

8.2 Developing the proposed reference tariff

Queensland Rail has proposed a step change in its capital, maintenance and operating expenditure for the West Moreton system, to address record forecast volumes. The 2025 DAU also introduces a range of measures to protect Queensland Rail's interests, including accelerated depreciation and 'triggers' that enable it to opt out of proposed capital expenditure. But the representations stakeholders made to us and the evidence they provided to us suggest that Queensland Rail's customers have had limited involvement in developing such spending plans.³¹⁹

In this section, we set out our expectations for a reasoned and prudent strategy, explore the benefits of agreed outcomes, and discuss our role in the negotiation and assessment process.

³¹³ All West Moreton tariff figures have been converted to 2025 dollars, using actual and forecast CPI. The 2020 undertaking also included a ceiling price, calculated based on forecast volumes of 2.1 million tonnes, that will have escalated to \$44.82/ 000 gtk in 2025 dollars by the end of the 2020 undertaking period.

³¹² The proposed cost per net tonne is an average – the actual proposed cost per net tonne for each mine varies depending on its distance from the port. The total average cost per tonne, including traversing the Metropolitan system, is \$15.71,

³¹⁴ Queensland Rail's forecast volumes ramp up to 9.6 million tonnes during the 2025 undertaking period.

³¹⁵ New Hope, sub. 5, p. 8; sub. 12, pp. 5-6; Aurizon Coal and Bulk, sub. 2, pp. 72-75.

³¹⁶ New Hope, sub. 5, pp. 4, 8-12; Yancoal, sub. 9, pp 13-17.

³¹⁷ New Hope, sub. 5, pp. 5, 9-10; Yancoal, sub. 9, pp. 3, 13-14.

³¹⁸ New Hope, sub. 5, p. 24; Yancoal, sub. 9, pp. 3, 19-20.

³¹⁹ Yancoal, sub. 16, p. 3; New Hope, sub. 12, pp. 4–5.

8.2.1 Reasoned and prudent strategy

Spending is most likely to be efficient if it is based on a reasoned and prudent medium- to longterm strategy such as an asset management plan and/or a long term operational and development plan setting out possible future investment. The strategy should preferably be agreed on a consensus basis with customers and reflect their service requirements and cost and risk preferences. Key elements should include:

- appropriate targets, such as an expected capacity, at an agreed level of reliability
- a framework for achieving those goals.

This reasoned and prudent strategy should clearly articulate the approach for capital, maintenance and operating programs, and be capable of being assessed by us as we consider whether Queensland Rail's proposal is appropriate to approve with regard to the section 138(2) factors and the pricing principles in the QCA Act. It should reflect the legitimate interests of both users and Queensland Rail. As users and Queensland Rail are reliant on each other, it is likely that their interests will align in many circumstances. Where their objectives diverge, it is highly preferable that they negotiate trade-offs rather than leave us to determine an outcome.

The reasoned and prudent strategy will drive detailed spending plans, supported by business cases that explain how the proposed specific investments align with the strategic goals (see section 8.3.2, including Figure 7).³²⁰

We identified the need for coherent strategies in our recent guideline on climate change related spending³²¹ but were clear that the same general principles applied for all spending proposed by a regulated business. We emphasised our preference for agreed approaches as the best means for achieving outcomes that are appropriate to approve.³²²

Consulting with customers is a key part of developing a coherent strategy under the negotiate-arbitrate and similar frameworks ... Open and transparent stakeholder consultation is consistent with good business practice and demonstrates that a business is committed to aligning its strategy with the needs of its customers. Consultation is particularly important for strategies that are expected to drive larger expenditures.³²³

Queensland Rail is planning larger expenditures for the 2025 undertaking period – its proposed capital program of \$346.9 million over the five years is almost two-thirds of its estimated regulatory asset base at the start of the period, and more than double its capital indicator for the 2020 undertaking.³²⁴ And this higher spending is without any increase in the previously stated capacity of the system.³²⁵

Queensland Rail maintained that its West Moreton capital and maintenance expenditure proposal would reduce operational risk and increase confidence that the supply chain could deliver record coal railing demand. It said:

Queensland Rail's capital and maintenance programs for DAU3 aim to meet the requirements of access holders by reasonably limiting the number of speed

³²⁰ We refer to the long-term broader plan as a 'reasoned and prudent strategy' and the related asset-specific plan as a 'business case' to reflect the differences in the nature of the two types of plans, and to avoid confusion.

³²¹ QCA, *Climate change related spending*, guideline, September 2023.

³²² In the case of services, like Queensland Rail's West Moreton route service, that are declared for access under Part 5 of the QCA Act, the approach or strategy would need to meet and balance the criteria in section 138(2) of the QCA Act.
³²³ QCA, Climate change related spending, guideline, September 2023, p. 5.

³²⁴ \$346.9 million of forecast capital expenditure, which is 64.8% of the proposed opening regulatory asset base (before allocation) of \$535.2 million.

³²⁵ QCA, *Queensland Rail 2020 draft access undertaking*, decision, February 2020, pp. 57-59.

restrictions and section closures and therefore increase reliability with the aim of an associated throughput improvement \dots^{326}

However, stakeholders have represented to us that they received insufficient detail from Queensland Rail to form a view on its spending plans. Yancoal said there were:

doubts as to the prudency of the high costs of capital expenditure and operation and maintenance costs, given the limited disclosure provided and high-level nature of the justifications provided³²⁷

New Hope said:

At this stage, NHG considers there is a real question as to whether, given the level of information provided, stakeholders have been provided with procedural fairness in respect of QR's various cost proposals.³²⁸

New Hope added that it had received a high-level overview of Queensland Rail's planned DAU in April 2023 and a further briefing on revenue allowances and tariffs in October 2023, shortly before the DAU was lodged. It said:

A key impediment to consultation with QR appears to be that a reluctance to openly discuss proposals prior to receiving approval by the QR board is followed by a reluctance to alter proposals which have obtained board approval.

We remain hopeful that meaningful consultation will commence in the near term.³²⁹

In relation to West Moreton expenditure, the 2025 DAU most likely does not reflect an informed view from its customers. This makes it difficult for us to determine whether Queensland Rail's West Moreton tariff proposal provides for efficient investment in and use of the rail infrastructure to support an agreed standard of service or, in the absence of agreement with customers, a level of service reflecting customers' projected requirements.

Queensland Rail and its customers already have substantial sunk costs in rail and mining infrastructure. Queensland Rail has proposed a large investment program, and some of the West Moreton customers have major investments planned or underway. It is difficult to see how Queensland Rail could contract for its full 9.6 million tonne demand forecast if it became clear it would be unable to deliver that volume on a sustainable basis.³³⁰

The volume and price considerations are intertwined. If the price is too high, then the customers may not contract for the forecast capacity. And if the price is too low, Queensland Rail may not have sufficient incentive to invest to support the volume as it may not receive a return commensurate with the regulatory and commercial risks. That balance is best addressed, at least in the first instance, by exchange of information and good-faith negotiation between the parties.

If Queensland Rail were proposing to assume the investment risk at prices that were sustainable for customers, and to be liable for making its contracted capacity available, it might be appropriate for it to proceed without a reasoned and prudent strategy for investing in and operating the West Moreton system, preferably developed on a consensus basis. However, to the extent that customers are expected to share, or assume, a substantial proportion of the investment and performance risk for West Moreton, it is reasonable for them to expect to be kept informed, and to share in the decision-making process.

³²⁶ Queensland Rail, sub. 1, pp. 9-10.

³²⁷ Yancoal, sub. 9, p. 3.

³²⁸ New Hope, sub. 5, p. 8.

³²⁹ New Hope, sub. 12, p. 5.

³³⁰ For a discussion of system capacity, see section 9.2.2, and Arcadis, Review of West Moreton System Costs and Other Technical Matters in Queensland Rail's 2025 DAU, May 2024, pp. 5, 19-21. that is published with this draft decision.

It is also important that all options be considered, including measures that are alternatives to investment by Queensland Rail. It may be that there are above-rail options or different stockpiling or scheduling approaches that will be more efficient ways to operate the supply chain. A below-rail investment that is done in isolation runs a real risk of not being the most efficient strategy. These alternative approaches should be explored with the West Moreton stakeholders, including the end users and the above-rail operators.

Queensland Rail has not proposed changes to the 2020 undertaking approval processes for capital and maintenance expenditure (see section 8.3.2). However, Queensland Rail and its customers may wish to consider an approach similar to the one Aurizon Network and its customers in central Queensland adopted under the 2017 Aurizon Network access undertaking (UT5), where maintenance and replacement capital expenditure are reviewed jointly. As we said in our 2021 guideline on pricing for Aurizon Network's GAPE and Newlands systems:

UT5 was developed through extensive negotiation between Aurizon Network and stakeholders and reflected a step towards increased collaboration between all parties. It provides a range of industry-led solutions and processes that create a more customer-centred approach to access regulation. Such an approach, if effectively utilised, may avoid regulatory error, reduces the regulatory burden on all parties and creates opportunities for parties to explore solutions to diverse and complex issues. In this context, we consider Aurizon Network, affected parties and other users should seek to uphold this type of approach to access regulation and engage productively, with the aim of presenting an agreed solution, while making compromises that are mutually agreeable.³³¹

Not just Aurizon Network and its customers provide an example for Queensland Rail and its West Moreton customers to emulate. DBI and its customers negotiated the most recent tariff for DBCT without seeking a determination from us. Moreover, Queensland Rail negotiates prices and other terms for its freight traffic on almost all its route services, the exception being West Moreton.³³²

We have sought to address the need for an informed view by providing Queensland Rail's customers with more information, both in this draft decision (see chapter 9) and in an accompanying report from our technical advisor, Arcadis. We encourage Queensland Rail to expand on this information to enable constructive discussions as it engages with customers on agreed outcomes.

8.2.2 Agreed outcomes

The reference tariff is part of a package of service standards, obligations, costs and risks, which reflects customers' preferences, the legitimate business interests of Queensland Rail, and other factors. Given Queensland Rail and its customers are best placed to assess their respective needs and preferences, we are keen for them to work together to find the best outcome. As we said in our recent guideline on climate-related spending:

Our first preference is that regulated businesses and their customers reach consensus on strategies and spending approaches that suit all parties. Where there is an agreed capital investment or operating expenditure proposal, our role is likely to be light-handed.³³³

We also emphasised the value of commercially agreed outcomes under the Queensland access regime in our guide on access disputes under the QCA Act.³³⁴ The guide explains that the access

 ³³¹ QCA. Pricing of shared infrastructure for the GAPE and Newlands systems, guidance paper, September 2021, pp. 2-3.
 ³³² Queensland Rail, sub. 14, pp. 23-25.

³³³ QCA, *Climate change related spending*, guideline, September 2023, p. 8.

³³⁴ QCA, Access disputes under the QCA Act, summary guide, June 2019.

regime 'is based on a negotiate-arbitrate framework, which envisages that, in the first instance, access to a declared service should be procured on the basis of terms and conditions that are commercially agreed between the access seeker and the provider of the declared service'.³³⁵ The introductory chapter of this draft decision also discusses the benefits of agreed outcomes (see section 1.3).

Even if Queensland Rail and its customers are unable to agree on all aspects of a reference tariff, they should have a minimum goal of an agreed reasoned and prudent capital, maintenance and operating strategy for West Moreton, which establishes clear performance targets and sets out how to achieve them. That will assist us in assessing the efficient levels of capital indicator and maintenance and operating allowance to be used in determining the reference tariff, should it be necessary for us to do so. We also set out in section 8.3.2 how that reasoned and prudent strategy should, in turn, drive specific spending proposals.

8.2.3 Our role

We wrote to Queensland Rail in 2022, saying we were keen to accommodate its preference to submit a voluntary DAU and welcoming its commitment to consulting extensively with stakeholders in preparing the DAU. We also said we wanted to leave sufficient time after our draft decision for Queensland Rail and other stakeholders to collaborate on matters that remained outstanding.³³⁶

Our role is to assess Queensland Rail's DAU, including the reference tariffs and related matters in Schedules D and E, having regard to the approval criteria in section 138(2) of the QCA Act. If we find they are not appropriate to approve having regard to those criteria, we must suggest what Queensland Rail might propose that would be appropriate to approve.

We consider the objectives of the QCA Act are best achieved by Queensland Rail and its customers working towards agreed approaches on all aspects of the access undertaking. In respect of matters that cannot be agreed, we will assess the appropriate outcome against the criteria in section 138(2) of the QCA Act. In respect of matters agreed, we still, to discharge our statutory role, do need to apply an independent review to confirm that the agreed matters comply with section 138(2) – which includes, among other things, having regard to the interests of future access seekers. In practice, though, we expect that an outcome agreed by the parties will significantly assist in satisfying us that the criteria in section 138(2) have been appropriately addressed.

In assessing whether a pricing proposal is appropriate, having regard to the criteria in the QCA Act, we look, among other things, for evidence that it provides sufficient returns to compensate the access provider for the regulatory and commercial risks. But equally, we need to form a view on whether the proposal promotes efficient investment in, use of and operation of the infrastructure.

Some of the best evidence that it is appropriate to approve a tariff proposal is that the directly affected parties – Queensland Rail and its customers – have negotiated a price and associated terms that suit all parties. That should include the standard of service and feasible volumes, as well as mechanisms for engaging on matters such as capital spending and maintenance, where risks and outcomes are shared.

³³⁵ QCA, Access disputes under the QCA Act, summary guide, June 2019, p. 1.

³³⁶ QCA, *Queensland Rail access undertaking timeline*, letter to Queensland Rail, 21 September 2022.

8.3 Capital spending: indicator and approval

The treatment of capital expenditure in the rail reference tariffs we regulate has for many years involved two steps:

- preparing a forecast the capital indicator that is used in the building blocks when the tariff is approved in an access undertaking
- conducting a subsequent review of the spending, once the projects are complete, before they are approved to be included in the regulatory asset base.

8.3.1 Capital indicator

Queensland Rail's tariff proposal includes a capital indicator, which enables its revenue to reflect expected investment during the term of the undertaking. The indicator is a forecast of capital spending. However, the capital projects built during the regulatory period are not included in the regulatory asset base until the next undertaking, to allow for them to be reviewed for prudency and efficiency.³³⁷

This approach was introduced to encourage investment by the regulated party, by:

- providing a return on expected efficient investment (the capital indicator) during the regulatory period
- having a pathway for capital expenditure to be included in the regulated asset base in the subsequent undertaking period, provided a decision-making and spending process that is clearly set out is followed.

But for the avoidance of doubt, the capital indicator is a budgetary and cashflow measure, not a mechanism for preapproval of proposed capital investments to be included in the asset base. In other words, while the specific projects proposed by Queensland Rail in its DAU material are relevant, our assessment of the efficiency of the capital indicator will be based more on the strategy underlying the investment plans (see section 8.2.1).

Approval of the capital indictor, as part of assessing whether it is appropriate to approve the reference tariff, should not be treated by Queensland Rail or its customers as an implied or explicit approval of the prudency of any of the projects included in Queensland Rail's DAU material. The prudency assessment will be done through the capital expenditure approval process in Schedule E of the 2025 DAU, as under previous undertakings.

8.3.2 Capital expenditure approval process

Queensland Rail has proposed a capital expenditure review and approval process in Schedule E of the 2025 DAU that is unchanged from that in the 2020 undertaking.

This provides for annual review of commissioned capital projects that are only included in the regulatory asset base once they have been found prudent in scope, standard and cost.

In our guideline on climate change-related spending, we set out a four-part approach to developing a spending proposal and business case. The approach focused on demonstrated need, consultation, consideration of options, and efficient cost. Once there is a reasoned and prudent strategy for West Moreton, the specific spending proposals arising from that strategy should be

³³⁷ In fact, capital projects completed in the last year (or sometimes two years) of the regulatory period are still included as 'capital indicator' amounts in the building blocks for the subsequent undertaking, as the prudency review cannot be completed before the new undertaking takes effect. See sections 8.3.2 and 9.3.1.

prepared under a similar need, consultation, options and efficiency framework, before they are submitted to us for assessment under the processes in Schedule E (see Figure 7).

Figure 7 : The four	key elements of	f a robust business case
---------------------	-----------------	--------------------------

Demonstrated need	Any spending proposal should solve an identified problem or fulfil a demonstrated need, whether it be for increasing or sustaining the service potential of the facility. This need should be demonstrated with reference to a reasoned and prudent long-term strategy and should not be ad hoc (see section 8.2.1). This might be established through a quantitative approach or, where that is not possible or reasonable, through a qualitative analysis.
Consultation with customers	The proposal should have regard to customers' views, including their risk preferences. Consultation should include potential customers, where this is possible. What have customers said about their preferred approach to investment in the rail infrastructure? Have customers been provided with robust and transparent information? How have their views been taken into account when choosing the proposed approach? This consultation could be demonstrated through customer letters of support or, as is done for some regulated businesses, a customer vote process. The consultation on an individual investment may be less important than consultation on a reasoned and prudent overall strategy (see section 8.2.1).
Demonstrated consideration of options	The business should show it has considered a range of alternative ways to address the identified problem. What options have been considered in assessing both the scope and standard of the planned spending? What are the pros and cons of those options? Are the options consistent with any hierarchy of options identified in the long-term strategy? The business case should explain how and why the proposed approach has been selected over the alternatives.
Efficient cost	The efficient cost should reflect value for money, rather than a simplistic choice of lowest upfront cost. Least cost should be considered over the life of an asset. Efficient cost also reflects externalities, in addition to the costs directly incurred by the regulated business and, ultimately, its customers.
	The upfront cost could be established through an appropriate process, such as a competitive tender.

Source: Adapted from Table 1 in QCA, Climate change related spending, guideline, September 2023, p. 7.

New Hope and Yancoal submitted that Schedule E should be amended to require Queensland Rail to:

- provide details of capital expenditure projects to customers
- consult with customers (including a vote process) before committing to significant projects
- prepare business cases that demonstrated the need for the projects and how a particular scope had been selected
- seek preapproval from us where customers did not support a proposed project, as rejection by customers would need to be relevant, but not determinative.³³⁸

The two mining companies provided the same suggested drafting of amendments to Schedule E.³³⁹

³³⁸ New Hope, sub. 5, p. 26; Yancoal, sub. 9, p. 20.

³³⁹ New Hope, sub. 5, pp. 34-36; Yancoal, sub. 9, pp. 26-28

Aurizon Coal and Bulk endorsed this capital expenditure approval proposal and said Queensland Rail should 'investigate interest in forming an industry group to evaluate and approve the scope, benchmarks and costs of QR's maintenance and capex plans on a regular basis'. It said such a group would be similar to the Rail Infrastructure Group established under Aurizon Network's UT5, and the Rail Capacity Group established under ARTC's Hunter Valley Access Undertaking.³⁴⁰

Queensland Rail said it was contractually committed to providing the capacity and was responsible for the safe operation of the network.³⁴¹ It said:

Queensland Rail believes that the proposed amendments to DAU3 suggested by Yancoal and New Hope, where individual projects are voted on by users will introduce uncertainty and timing risk to the delivery of the overall program and reduce the effective control the Queensland Rail has over its network. This will put at risk system capacity, operational efficiency and safety.³⁴²

We would need more evidence to form a view that a consultative approach to managing capital and maintenance spending on West Moreton would put capacity, efficiency or safety at risk. However, it may be that some mechanism more iterative and cooperative than a customer vote is the appropriate way forward. One option might be to expand the scope of the system user group that Queensland Rail has established for West Moreton.

Much of the success of a cooperative approach to supply chain management would depend on the 'buy-in' of all parties. This is more likely to be achieved with a negotiated system approach than with something proposed only by some of the intended participants or developed by us. A framework for cooperation that was implemented without the backing of all parties – particularly the operator of the system – would probably be less effective than one that had broader support.

We share the view that a consultative mechanism like the Rail Infrastructure Group that has been operating successfully in central Queensland for several years has merit.³⁴³ We have a role in that central Queensland process, but it is largely driven by Aurizon Network and its customers – and was negotiated and agreed by them.

Queensland Rail, too, has opted in the past to obtain support from its customers for proposed West Moreton capital spending. For example, it provided letters in favour of its Toowoomba Range stabilisation project as part of its 2018 application for preapproval of the investment under Schedule E of the 2016 undertaking.³⁴⁴

We consider the mechanism proposed by New Hope and Yancoal may be the appropriate way to promote efficient investment in and use and operation of the West Moreton system, and be in the legitimate interests of both Queensland Rail and its customers (ss. 138(2)(a), (b), (e) and (h)). However, there may be other ways for Queensland Rail and its customers to apply the need, consultation, options and efficiency approach set out in Figure 7.

We look forward to the outcome of discussions between Queensland Rail and its customers.

³⁴⁰ Aurizon Coal and Bulk, sub. 2, pp. 75 and 76.

³⁴¹ Queensland Rail, sub. 14, pp. 11-12.

³⁴² Queensland Rail, sub. 14, p. 12.

³⁴³ See Aurizon Coal and Bulk, sub. 2, p. 75.

³⁴⁴ More information on the Toowoomba Range Stabilisation project, including Queensland Rail's submission, stakeholder comments, expert reports and our decision, is available on the QCA website <u>at *Capital expenditure preapproval*</u>.

Summary 8.1

It may be appropriate for Queensland Rail to amend the provisions for review of its capital spending in the 2025 DAU (Schedule E) to provide for more involvement of its customers in the assessment and approval process. The mechanism for greater customer involvement is best negotiated between Queensland Rail and the relevant customers.

8.4 Capacity assessment

Queensland Rail has based its proposed West Moreton coal tariff on annual volume forecasts of 9.6 million tonnes for coal.^{345,346} Queensland Rail said the coal volume estimate was based on its acceptance of advice from the three West Moreton coal customers of volumes they wished to contract and/or renew.³⁴⁷ It said 'the coal mines and their operators can contract up to 97 return train paths per week across the range' and could also run ad hoc train services for as many as 16 more paths.³⁴⁸ But Queensland Rail also said it needed to undertake \$346.9 million of capital expenditure and \$172.5 million of maintenance over the five-year undertaking period to enable the infrastructure to deliver the proposed volumes.³⁴⁹ This compares with Queensland Rail's proposed opening asset value of \$535.2 million.

Stakeholders requested more information on the current capacity of the network, including the key assumptions behind Queensland Rail's network capacity analysis, and how the proposed capital spending would increase West Moreton capacity to the 9.6 mtpa forecast. New Hope and Yancoal suggested an independent assessment of capacity on the West Moreton system was necessary. They said the process used in the central Queensland coal network (CQCN), whereby capacity was assessed by an independent expert, provided a suitable model.³⁵⁰

Australian Coal and Bulk provided results of its own analysis of West Moreton capacity and said that Queensland Rail needed to publish the assumptions on which it had based its capacity assessment.³⁵¹

Queensland Rail did not address the stakeholders' proposal for an independent capacity review in its responsive submission.

Engineering firm Arcadis has provided views on the potential capacity of the West Moreton system as part of its technical assessment of the information provided by Queensland Rail to accompany the 2025 DAU. This assessment can be found in section 3.1.3 of its report.³⁵² Our views on specific

³⁴⁵ A West Moreton coal train carries about 2,000 tonnes of coal. So, a weekly path accounts for about 100,000 tonnes of annual coal capacity, assuming 50 paths a year. On that basis, 97 weekly paths would be sufficient for approximately 9.7 million tonnes of annual coal haulage.

³⁴⁶ Queensland Rail did not provide a forecast for grain volumes.

³⁴⁷ Queensland Rail, sub. 1, p. 12.

³⁴⁸ Queensland Rail, sub. 1, p.7.

³⁴⁹ Queensland Rail, sub. 1, pp. 30, 38.

³⁵⁰ New Hope, sub. 5, p. 24; Yancoal, sub. 9, pp. 19-20 and sub. 16, p. 7.

³⁵¹ Aurizon Coal and Bulk, sub. 2, pp. 61-66. The assumptions that Aurizon Coal and Bulk suggested Queensland Rail should publish are listed on pp. 63-64.

³⁵² Arcadis, Review of West Moreton System Costs and Other Technical Matters in Queensland Rail's 2025 DAU, May 2024, pp. 5, 19-21.

drivers of West Moreton capacity, including the Arcadis advice, are discussed in section 9.2.2 of this draft decision.

Our preliminary position is that greater clarity on the capacity of the network will promote efficient investment in and use of West Moreton rail infrastructure (s. 138(2)(a) of the QCA Act). As such, an independent review of capacity has merit. It will be in the interests of customers to have confidence that contracted volumes can be delivered, particularly given the scale of investment and maintenance spending that Queensland Rail has proposed (ss. 138(2)(e) and (h)).

We encourage Queensland Rail and its customers to address independent capacity assessment as part of their discussions on overall West Moreton tariff matters and work toward an agreed approach that can be presented to us to be included in our final decision.

Summary 8.2

It may be appropriate for Queensland Rail to amend the 2025 DAU to include an independent capacity review process for West Moreton.

8.5 Risk mitigation

Queensland Rail has proposed various measures to reduce its risks in investing in and maintaining the infrastructure that provides the West Moreton route service. These include:

- accelerated depreciation that, among other things, helps mitigate its asset stranding risk (see section 8.5.1)
- 'triggers' that let it opt out of proposed capital investment that has been included in the building blocks used to assess the reference tariff (section 8.5.2).

Customers also proposed risk mitigation measures. These include:

- capital expenditure reconciliation, with tariff adjustments where investment is materially different from forecasts (section 8.5.3)
- priority rights to renewal of access for existing West Moreton access holders (section 8.5.4).

8.5.1 Accelerated depreciation and capital underwriting

Our consistent position on risks is that they are best borne by those with the greatest ability to manage them. We therefore consider that accelerated depreciation or some other mechanism agreed between Queensland Rail and its customers is appropriate to address any difference between the expected technical and economic lives of Queensland Rail's assets.

The 2025 DAU proposes an accelerated depreciation profile based on forecast mine lives estimated by AME in a report for Queensland Rail. It provides for existing assets to be depreciated over 19 years, and capital expenditure during the 2025 undertaking period to be depreciated over 14 years or less.³⁵³ These proposed asset lives are significantly shorter than those that have been applied to

³⁵³ These asset lives are considered in more detail in section 9.6.

Queensland Rail's 2025 Draft Access Undertaking

West Moreton system assets when assessing the West Moreton tariffs for the 2016 and 2020 undertakings (see Table 10 in section 9.6).

Queensland Rail said the accelerated depreciation would ensure that spending during the term of the 2025 undertaking was recoverable over the mine lives of all users, while longer asset depreciation would result in the remaining users paying for the assets they continued to use.³⁵⁴ Queensland Rail said accelerated depreciation was equitable for users:

This approach results in an equitable distribution of costs to each West Moreton System miner. The shorter accelerated depreciation period is equitable as it will ensure that all mines pay for the new assets required for the record 9.6Mtpa tonnage forecast ... post 2034 the remaining mines will continue to pay for the service they are still using.

Queensland Rail is proposing an equitable relationship of the costs attributed to the individual mines which protects the legitimate business interests of Queensland Rail's customers.³⁵⁵

Queensland Rail justified its accelerated depreciation methods based on its analysis of structural viability of coal mines informed by advice from consultants AME³⁵⁶ and HoustonKemp.³⁵⁷ Queensland Rail also noted our position on accelerated depreciation as a measure to address asset stranding risk in our decision on the 2020 DAU. In that decision, we said:

Our decision is to not implement accelerated depreciation as part of this DAU process; however, we would be amenable to accepting an appropriate accelerated depreciation profile, should Queensland Rail propose it as part of a DAAU. We consider that an appropriate accelerated depreciation profile would likely be sufficient to address the longer-term stranding risks that West Moreton coal faces.³⁵⁸

Stakeholders said they were concerned about the proposal, particularly its impact on affordability.³⁵⁹ Yancoal said the increase in reference tariff could potentially result in mines ceasing production earlier than assumed.³⁶⁰

In principle, we consider it may be appropriate for Queensland Rail to undertake depreciation with schedules that reflect prudent and efficient management of its asset base and align with expected economic lives (s. 138(2)(b)).³⁶¹

However, the question of whether accelerated depreciation would act to mitigate asset stranding risk is not straightforward and cannot be considered in a vacuum. Queensland Rail's proposal would significantly increase the annual depreciation costs recovered from users and increase the reference tariff. This may, in turn, bring forward the date at which assets cease to be used.

There are other ways to mitigate asset stranding risk, apart from accelerated depreciation. Longerterm contracts or user underwriting of capital expenditure are just two measures that could act to

³⁵⁴ We note that HoustonKemp analysis supplied by Queensland Rail did not strongly support the different depreciation treatment of capital expenditure and existing assets. HoustonKemp said 'we recommend that new capital investment should be recovered over the estimated weighted average remaining mine life' It recommended these lives be reviewed periodically. See Queensland Rail, sub. 1, attachment 5, p. 17.

³⁵⁵ Queensland Rail, sub. 1, p. 12.

³⁵⁶ Queensland Rail, sub. 1, attachment 4, pp. 2-5.

³⁵⁷ Queensland Rail, sub. 1, attachment 5, pp. 8-11.

³⁵⁸ QCA, *Queensland Rail draft access undertaking*, decision, February 2020, p. 50.

³⁵⁹ New Hope, sub. 5, p. 12; Yancoal, sub. 16, p. 4.

³⁶⁰ Yancoal, sub. 16, p. 4.

³⁶¹ We consider it appropriate to periodically revise depreciation schedules to reestablish the efficient use of the assets. Where technical lives are used, it is reasonable to assume that operational characteristics that established the original technical lives of the asset would also be periodically reassessed to their currency in light of the ongoing use of the asset. Where volumes are expected to exceeded prior volume forecasts it is reasonable to question whether the technical life of the asset is still appropriate to the operational wear and tear faced. We do not consider that technical life is a set-andforget depreciation methodology.

reduce the likelihood that Queensland Rail would be unable to recover its infrastructure investments.

Queensland Rail and its users are best placed to understand the full range of financial and contractual options available to mitigate stranding risk as well as the potentially varying impacts on their respective commercial operations and identify the most efficient method(s) to address it. As discussed in section 8.5.5, we consider the first step to finding the most economically efficient outcome will be negotiation between Queensland Rail and its users that takes into account asset lives along with other risks and considerations.

8.5.2 Volume triggers (tariff reset)

The 2025 DAU includes 'volume triggers' that provide for Queensland Rail to lodge a DAAU to reset the reference tariff each time a West Moreton contract is up for renewal if it is not renewed. Queensland Rail proposed that the QCA would have the same powers to review the DAAU as it would if it had issued an initial undertaking notice (Schedule D, cl. 3.2).

Queensland Rail said this would give it 'an opportunity to reset its capital and maintenance program to reduce the capital expenditure that would otherwise be brought forward to meet capacity requirements and protects our [Queensland Rail's] customers from paying for capacity that is not required.'³⁶²

New Hope and Yancoal said Queensland Rail's proposed drafting of clause 3.2 of schedule D would 'immunise' Queensland Rail from any volume risk and, in the case of one of the three mines shutting down, transfer that risk to the other mines. They said that, if it was implemented, it should be symmetric, so that there would be a further review if volumes recovered.³⁶³

We consider the volume trigger as proposed by Queensland Rail may be in the legitimate business interest of Queensland Rail, as it might protect its revenue during the term of the 2025 undertaking (s. 138(2)(b)). However, there are other ways to achieve similar objectives that might be more appropriate to approve.

These include, for example, having the triggers work from the bottom up, rather than the top down, with capital spending only added to the capital indicator once volumes are confirmed, or customers agree to underwrite the capital expenditure. This might give the customers an incentive to contract for access, as that would give them and Queensland Rail more certainty to support investment decisions.

The driver of Queensland Rail's trigger proposal seems to be that its forecast volumes are not contracted for the full term of the undertaking, so some investments made by Queensland Rail will therefore turn out to have been unnecessary. This concern overlaps, at least in part, with the miners' proposed capital expenditure reconciliation, which would address under-spending of the capital indicator amount.

³⁶² Queensland Rail, sub. 1, p. 30.³⁶³ Yancoal, sub. 9, p. 3.

8.5.3 Capital expenditure reconciliation (true-up)

Yancoal and New Hope submitted that there should be an annual reconciliation or true-up to address any under- or over-spend of the capital indicator, with tariffs adjusted during the term of the undertaking, rather than only in the subsequent term.³⁶⁴

Both mining companies provided the same suggested drafting of amendments to replace clause 7 of Schedule E.³⁶⁵

Yancoal said that without reconciliation provisions:

there is real potential for users to be paying an inflated tariff during the entirety of the DAU3 period, with the affordability and therefore volume risks that creates, with relief for the lower than proposed capital expenditure only being experienced in the next term.³⁶⁶

New Hope said that an earlier true-up (reconciliation) would prevent cashflow impacts on either Queensland Rail or customers and avoid 'creating winners and losers within the customer group as relative volumes change'.³⁶⁷

In its responsive submission, Queensland Rail quoted Yancoal's comment about capital expenditure reconciliation but did not address the matter.³⁶⁸

We consider an earlier reconciliation during the term of the undertaking may promote efficient investment in, and use of, West Moreton infrastructure. It may also be in the legitimate business interests of both Queensland Rail and its customers, if it applies to both upward and downward variances in actual spending, compared with the capital indicator (ss. 138(2)(a), (b), (e) and (h)). However, such a measure is best considered as part of an overall package of measures for sharing and mitigating risks, that is negotiated by the parties.

8.5.4 Renewal rights

Queensland Rail proposed the West Moreton miners be subject to the same one-time renewal right that is provided for when access arrangements are prioritised (see section 3.2 of this draft decision, and clause 2.10 of the 2025 DAU).

However, Yancoal and New Hope said miners that were underwriting Queensland Rail's West Moreton infrastructure spending should have preferential renewal rights. They said this was because they will assume the burden of paying the 2020 undertaking capitalised losses, as well as providing Queensland Rail with the 'vast majority' of its return on capital if accelerated depreciation is implemented.³⁶⁹ Yancoal said:

It is difficult to see why a future user, who will not have underwritten the existence and continued maintenance and operation of the West Moreton network, should have the same renewal rights as those who have.³⁷⁰

The appropriate level of certainty about renewal will depend significantly on the degree to which the access holders are required to address Queensland Rail's investment risk (see the other risk-

³⁶⁴ The capital indicator is discussed in more detail in sections 8.3.1 and 9.5.2 of this draft decision. The capital expenditure carry-over account, which governs the treatment of the under- or-overspend of the capital indicator through adjustments in the subsequent undertaking period, is considered in section 9.3.1.

³⁶⁵ New Hope, sub. 5, pp. 36–38; Yancoal, sub. 9, pp. 28–29.

³⁶⁶ Yancoal, sub. 9, p. 20.

³⁶⁷ New Hope, sub. 5, p. 26.

³⁶⁸ Queensland Rail, sub. 14, p. 11.

³⁶⁹ Yancoal, sub. 9, pp. 20-21; New Hope, sub. 5, p. 26.

³⁷⁰ Yancoal, sub. 9, p. 21.

related matters discussed in this section 8.5). But as a broad principle, the more the risk is transferred to the customers, the more certainty over security of access they should receive.

Our preliminary position therefore is that it may be appropriate to amend Queensland Rail's DAU to include renewal rights that apply to West Moreton access holders, particularly if Queensland Rail and the relevant customers agree that would be their preferred way to implement those rights, as part of an overall allocation of risks. However, we also consider that it may not be appropriate to have a blanket removal of clauses 2.10(c)(i) and (iii), as that would apply to all of Queensland Rail's customers, not just those that might provide capital underwriting and other risk mitigation measures for Queensland Rail for West Moreton.

It may also be appropriate that renewal rights for West Moreton access form part of the contractual agreements between Queensland Rail and its customers that emerge from their negotiations on the reference tariff and other terms. The 2025 DAU retains existing provisions that nothing in the undertaking prevents an access holder exercising contractual renewal rights, and that exercising such rights does not require an access application (clause 2.10(d)).

We encourage further submissions, preferably on a consensus basis, on how and whether to implement West Moreton renewal rights and look forward to learning the outcome of discussions between the parties.

8.5.5 Conclusion: negotiated risk-sharing and security

The complicated web of risk-sharing measures and security of access for West Moreton is best negotiated between Queensland Rail and its customers. Each of the four measures discussed above – depreciation, triggers, capital reconciliation and renewal rights – should be 'on the table' along with other risk-related matters in discussions between the parties.

For example, measures such as accelerated depreciation or capital underwriting that help mitigate Queensland Rail's asset stranding risk may be balanced against renewal rights for the customers who provide that assurance to Queensland Rail. Equally, Queensland Rail's volume triggers and the miners' capital expenditure reconciliation seem like two sides of the same coin and should be discussed as part of the parties' overall negotiation about how to address volume uncertainty.

It may also be that different customers will want different approaches to risk sharing. In that case, it may be best that measures like capital underwriting and security of access be negotiated on a bilateral basis between Queensland Rail and its customers, just as they negotiate matters like contract term.

Summary 8.3

It may be appropriate for Queensland Rail to amend the 2025 DAU to include a balanced risk-sharing approach for West Moreton, that has regard to the risk preferences and commercial requirements of all parties.

8.6 Tariff structure

We consider having a two-part tariff and associated distance taper will continue to strike an appropriate balance between the 'user pays' principle, revenue adequacy for Queensland Rail and fostering economic development along the West Moreton system.³⁷¹

The 2025 DAU provides for a two-part tariff that recovers the annual revenue requirement on a train path basis and on a weight and distance basis (i.e. per gtk) (sch. D, cl. 3.1).

The two-part tariff was introduced in 2010 to address the potential for above-rail investments to increase volumes, and therefore below-rail revenues. The tariff structure splits the gains from any increase in capacity per train – Queensland Rail increases its revenue, while customers benefit from lower unit costs.³⁷²

The tariff structure has the effect of creating a 'distance taper' – a tariff outcome that lessens the disincentive for developing mines further from ports.³⁷³ It also recognises that mines closer to the Port of Brisbane (e.g. New Hope's New Acland mine) do not use infrastructure west of their haulage point – but consume capacity so that fewer paths are available to access seekers further west (e.g. Yancoal's Cameby Downs mine). This means supplying a train path with an origin closer to the port carries an inherent opportunity cost to Queensland Rail (which requires sufficient revenue to cover access to the entire system) and the distance taper provides for users closer to the port to pay a portion of that cost.

New Hope said New Acland mine would be paying around 40% more per thousand gtk than Cameby Downs paid, and this was a cross-subsidy to mines that were further from the port. New Hope asked that the QCA confirm the Cameby Downs and Wilkie Creek mines would be paying sufficient revenue to cover at least the incremental cost of their services.³⁷⁴

Yancoal said we had previously considered the two-part tariff structure and considered it appropriate across multiple undertakings. It said the recent reopening of the Wilkie Creek mine demonstrated the distance taper had encouraged economic development.³⁷⁵

Our draft decision is that it is appropriate to retain the two-part tariff and distance taper approach in the pricing structure. The distance taper helps balance the competing objectives of cost reflectivity and revenue adequacy by:

- having miners closer to the port pay less for access than those further away, which is consistent with the user pays principle (ss. 138(2)(a), (e) and (h))
- encouraging economic development by mitigating some of the cost disadvantage faced by mines further from the port (ss. 138(2)(d) and (h))
- addressing in part the opportunity cost to Queensland Rail of selling a shorter path, which it might otherwise have been able to use for a more distant mine that provided more revenue (s. 138(2)(b)).

³⁷³ The distance taper has been a feature of the COCN tariffs since the first OCA-approved access undertaking in 2001.

- ³⁷⁴ New Hope, sub. 5, p. 13.
- ³⁷⁵ Yancoal, sub. 16, pp. 5-6.

³⁷¹ QCA, *Queensland Rail's Draft Access Undertaking*, decision, June 2016, p. 202.

³⁷² QCA, *QR Network 2009 Draft Access Undertaking*, draft decision, December 2009, p. 93.

Summary 8.4

It is appropriate to approve the two-part West Moreton tariff structure in the 2025 DAU.

Private and user-funded infrastructure

We consider the treatment of single-user infrastructure should be equitable between different users, so that the impact on other users of the system is similar, regardless of whether the asset is owned by Queensland Rail or an access holder.

New Hope said the balloon loop for the Cameby Downs mine at Columboola was included in the West Moreton RAB, while Queensland Rail had proposed that the spur and balloon loop it was building for New Acland would be private infrastructure and not included in the RAB. It said tariffs paid by New Hope were increased by having the Cameby Downs loop in the RAB.³⁷⁶

Yancoal said the loop for Cameby Downs was funded via Yancoal's payment obligations under an access facilitation deed (AFD) with Queensland Rail. It said changing the treatment now would create windfall gains and losses, and result in New Hope being cross-subsidised by Yancoal.³⁷⁷

As a matter of principle, we do not consider it appropriate to reopen the historical treatment of assets in these circumstances, particularly where the distributional impacts are unclear.

That said, in considering the appropriate treatment of private infrastructure going forward, as opposed to infrastructure underwritten by an end user, but owned by Queensland Rail, we have had regard to the approach in similar circumstances for Aurizon Network. It was common for single-user assets to be funded via AFDs, before the 2010 separation of the central Queensland coal transport assets that are now owned by Aurizon Network from the rest of the state's below-rail assets, that are owned by Queensland Rail.³⁷⁸ AFDs were used to enable capital underwriting by customers.

Customers under AFD arrangements made a payment to Queensland Rail or Aurizon Network (when their network businesses were both part of QR Network), that was rebated over the life of the asset, through a reduction in access charges to that customer. The asset was included in the RAB, but the payments to the customer matched the cashflows (return on and of capital) that the network provider received for having the asset in its asset base.

Aurizon Network, which has many single-user assets underwritten with AFDs, has an equivalent approach for privately funded assets. In effect, trains running across private infrastructure receive a discount off the reference tariff which is equivalent to the cashflows the network provider would have received if the mine-specific asset was in the network's asset base. In both cases, the cashflows are revenue neutral for Aurizon Network, the mainline infrastructure owner, with the discount or rebate covered by the tariffs paid by all users on the system.

³⁷⁶ New Hope, sub. 5, pp. 13-14.

³⁷⁷ Yancoal, sub. 16, p. 6.

³⁷⁸ Aurizon Network is a subsidiary of Aurizon Holdings. Aurizon Holdings initially traded as QR National Ltd., and Aurizon Network as QR Network, before the group was renamed in the early years after privatisation.

The New Acland balloon loop will be the first private coal infrastructure on Queensland Rail's West Moreton system, so there has not been a need previously to establish how such assets should be treated when assessing the West Moreton tariff.

We consider that an equitable arrangement is appropriate, such that the effect on system reference tariffs is the same regardless of infrastructure ownership. The Aurizon Network approach in central Queensland provides a good model for an equitable approach on the West Moreton system. Therefore, we consider that:

- the New Acland spur and loop should be taken into account when considering the West Moreton tariff
- there should be an equivalent rebate or discount arrangement with New Hope that matches the revenue Queensland Rail would have received if the asset was in its regulated asset base in the tariff calculation
- the 2025 DAU should be amended so that it has provisions for us to assess the prudency of scope, standard and cost of private infrastructure, similar to those in clause 6.3.2 of Aurizon Network's 2017 access undertaking (UT5).

We note that New Hope's balloon loop will allow for more efficient use of above-rail assets, by loading and turning around trains quickly, and in a way that does not negatively affect other users of the main line. This will benefit all West Moreton users and promote efficient investment in, use and operation of the rail assets (s. 138(2)(a)). An equitable funding treatment for private infrastructure will be in the interests of current users, and of future access seekers who may also require private infrastructure (ss. 138(2)(e) and (h)).

Summary 8.5

The West Moreton tariff in the 2025 DAU should be amended so that private infrastructure is included in the tariff calculation, subject to there being a rebate arrangement, and a process for us to assess the prudency of the assets.

8.7 Loss capitalisation and 2020 undertaking

We consider the approach to the losses capitalised during the term of the 2020 undertaking is best agreed between Queensland Rail and its access holders before the new undertaking period begins.

The loss capitalisation mechanism was introduced in the 2020 undertaking amid expectations that West Moreton volumes would fall to 2.1 million tonnes a year – a level that made it unlikely Queensland Rail would be able to charge a price that recovered the full efficient cost of providing access. The capitalised losses reflect the difference between Queensland Rail's total revenue requirement based on efficient costs, and its actual revenue based on the 'affordable' price in the 2020 undertaking, which was set below the ceiling price.

The exact mechanism for recovering the capitalised amounts, should demand rise to a level that makes that feasible, is left to be settled once volume expectations are clearer.³⁷⁹ The 2020 undertaking also includes a requirement that Queensland Rail submit a revised reference tariff if it

³⁷⁹ QCA, *Queensland Rail 2020 draft access undertaking*, decision, February 2020, p. 20.

expects annual contracted volumes will exceed 4.1 million tonnes during the term of the undertaking.³⁸⁰

Queensland Rail said in its material accompanying the 2025 DAU that it expected West Moreton volumes to exceed 4.1 million tonnes in 2023-24 and therefore intended to lodge a DAAU to reset the reference tariff. This would have included a proposed treatment of the loss capitalisation account.³⁸¹

However, given the Wilkie Creek mine is in receivership and is subject to a sale process, it is no longer clear that the 4.1 million tonne threshold will be reached, or that Queensland Rail will lodge a DAAU.

It is also not clear what the amount of capitalised losses will be by the start of the new undertaking period. Queensland Rail said \$22.8 million of losses had accumulated in the three years ended 30 June 2023. That total capitalised amount may still rise or fall during the remainder of the 2020 undertaking period. Experience so far has been that the 'break-even' at the current tariff is about 4 million tonnes a year (above that volume, there is a surplus, rather than a capitalised loss). It is possible, if volumes rise substantially over the next 18 months, that the accumulated capitalised losses will decline substantially or be eliminated by 30 June 2025.

Both New Hope and Yancoal said they would provide substantive submissions on loss capitalisation once Queensland Rail had provided its proposed approach under AU2 (the 2020 undertaking). However, they said that Queensland Rail's proposed partial changes to the loss capitalisation provisions in the 2025 DAU were not appropriate, as the clauses from the 2020 undertaking should either:

- be entirely deleted if the capitalised losses had been repaid by the start of the new undertaking, or
- provide a recovery methodology if there was a remaining balance to be recovered.³⁸²

As with the building blocks assessment of the tariff, any decision on the appropriate treatment of the capitalised losses will depend on clarity about volumes. At this time, the potential amounts of capitalised losses at the end of the 2020 undertaking period range from nil or immaterial, up to a substantial proportion of a year's allowable revenue.

Our preliminary position is that it may be appropriate to continue escalating the 2020 undertaking tariff at CPI for the remainder of the undertaking period. Any outstanding balance in the loss capitalisation account would then be rolled over into the 2025 undertaking tariff calculation, similar to the treatment of the capital carry-over account.

However, it will be best if Queensland Rail and its customers find an agreed approach that results in the capitalised losses being fully addressed during the term of the 2020 undertaking. We look forward to the outcome of their discussions.

³⁸⁰ QCA, Queensland Rail 2020 draft access undertaking, decision, February 2020, pp. 22-23.

³⁸¹ Queensland Rail, sub. 1, p. 55.

³⁸² Yancoal, sub. 9, pp. 17-18; New Hope, sub. 5, pp. 24-25.

Summary 8.6

The mechanism for addressing the accumulated capitalised losses is best negotiated between Queensland Rail and its customers.

8.8 Appropriate reference tariff

We are required to determine whether it is appropriate to approve the reference tariff and associated provisions proposed by Queensland Rail in Schedules D and E of the 2025 DAU, having regard to the criteria in section 138(2) of the QCA Act (s. 136(4)). We also have regard to comments from stakeholders. If we find it is not appropriate to approve the proposal, we must explain why and suggest changing it in a way that would be appropriate (s. 136(5)).

Queensland Rail has proposed a reference tariff based on forecast volumes, and an assumption that it will be able to deliver those volumes. Having regard to the approval criteria in section 138(2) of the QCA Act, including the interests of Queensland Rail and its customers, we are not currently satisfied, based on the information provided to date, that the tariff proposal satisfies the approval criteria. This is because of:

- a lack of an agreed reasoned and prudent strategy for the West Moreton route service, which means investment in and use and operation of the infrastructure is likely not to be efficient because it is not clear the investment is required to meet customer demand, and it is thus not aligned with the object of Part 5 of the QCA Act (s. 138(2)(a))
- the uncertainty about volumes, given the status of customers' plans and contract renewals, which means an assessment cannot be made of whether the proposed investment in and operation of the infrastructure is efficient and thus aligned with the object of Part 5 (s. 138(2)(a))
- the uncertainty about whether the proposed volumes can actually be delivered, even with the proposed spending, which casts further doubt on the efficiency of the investment and indicates the spending is not in the interests of access seekers and access holders (ss. 138(2)(e) and (h))
- the associated uncertainty about the efficient level of costs and, consequently, whether the proposal provides Queensland Rail with expected revenue that is at least enough to recover those costs and provide a return on investment commensurate with the regulatory and commercial risks involved (ss. 138(2)(b) and (g) and s. 168A(a)).

All of these missing elements create significant obstacles for us in seeking to use a building blocks approach to estimate an efficient price. For example, we can secure expert technical advice on whether a proposed strategy and program of works are likely to support a particular volume and standard of service. But this technical analysis becomes difficult (if not impossible) to provide in a manner that will lead to decisions consistent with the QCA Act without having confidence in the forecast volume or knowing the standard against which to assess the proposal.

It is therefore our draft decision not to approve Queensland Rail's proposed reference tariff because Queensland Rail has not demonstrated it is appropriate to do so having regard to the criteria in section 138(2) of the Act. Amongst other things, insufficient information has been provided to us to enable us to form a view the reference tariff meets the criteria in the pricing principles in section 168A of the Act or to enable us to assess the remaining section 138(2) criteria.

In the absence of any firm foundation on which to calculate a building-blocks-based reference tariff, we have sought an alternative approach that has regard to the approval criteria in section 138(2) of the QCA Act, including the interests of Queensland Rail and its customers.

Our preliminary position is that one appropriate approach may be for Queensland Rail to amend the 2025 DAU to provide for CPI escalation of the current (AU2) West Moreton reference tariff. Our understanding is that the 'break-even' volume for coal services at the 2020 undertaking tariff is about 4 mtpa, with Queensland Rail able to generate a surplus to recoup capitalised losses above that volume (see section 8.7). We also note that the West Moreton system has previously been able to support annual volumes of 7.8 million tonnes, with infrastructure at or below the current standard (see section 9.2).

This indicates that CPI escalation may be appropriate to approve with regard to the criteria in section 138(2) of the QCA Act, assuming annual volumes are sufficient that Queensland Rail recovers its efficient costs based on the current standard of the network, and below a level at which a step change in spending is likely to be required.

While the outlook remains uncertain, and we will review all our assumptions before making a final decision, there are a number of plausible scenarios that would give annual coal volumes in a range of 4 to 8 million tonnes. Given this, a price escalated by CPI may be in the interests of all parties, pending effective negotiations on an agreed reference tariff based on a reasoned and prudent strategy for capital expansion and maintenance, that reflects customer requirements. A price escalated by CPI may:

- be in the interests of customers and Queensland Rail, as it is a price that access seekers and access holders have been willing to pay, and at which Queensland Rail has been prepared to provide access during the current regulatory period (ss. 138(2)(b), (e) and (h))
- be likely to provide Queensland Rail with sufficient revenue to at least cover its incremental costs, based on expected volumes in the short term (ss. 138(2)(b) and (g) and s. 168A(a) see sections 8.7 and 9.2.1)
- be in the interests of Queensland Rail, as it is likely to recover its efficient costs at reasonably expected volumes, and there are mechanisms, including draft amending access undertakings (DAAUs), for it to seek to amend Schedule D to recover additional efficient costs if volumes rise significantly (ss. 138(2)(b) and s. 168A(a))
- provide time to gain greater certainty about volumes (and costs), which will enable more effective assessment of efficient investment in and use of the rail infrastructure (s. 138(2)(a))
- provide time for a negotiated outcome, including a reasoned and prudent strategy, which would promote efficient investment in, use of and operation of the West Moreton system for coal services (s. 138(2)(a)).

Barring a change in circumstances, and subject to consideration of further submissions and other evidence, it may be appropriate to include a CPI-escalated tariff or some alternative proposed by Queensland Rail or stakeholders in the final decision. In considering this, we will have regard to whether the price provides sufficient revenue to support continued safe and effective operation of the rail infrastructure, which is in the interests of Queensland Rail, its customers and the public (ss. 138(2)(b), (e), (d), (g) and (h)). A negotiated building blocks-based reference tariff and related measures could then be submitted for approval later in a DAAU under section 142 of the QCA Act.

However, we encourage negotiations between Queensland Rail and its customers on a new building-blocks-based reference tariff that is appropriate to approve from the start of the new

undertaking. To assist with those negotiations, we provide information in chapter 9. We observe that both sides have an incentive to negotiate and are better informed as to their respective interests, needs, constraints, risk profiles and willingness to pay.

Queensland Rail may want to consider the customers' interest in, among other things:

- infrastructure investment required to enable a reliable service that will deliver their coal to the port
- more information and greater participation in decision-making about investment and other supply chain issues
- a price that enables them to compete with miners on other rail systems.

The customers may want to consider Queensland Rail's interest in, among other things:

- comfort it will recover the cost of its investments, even where their economic life is shorter than their technical life
- a return on investment commensurate with the regulatory and commercial risks
- rail operations that comply with its safety obligations.

Both sides may wish to consider whole-of-system approaches to achieving service goals, including alternatives such as above-rail investment, and operational changes.

In considering the DAU under section 138(2), we will have regard to the extent to which Queensland Rail and its customers have provided information to us and to each other in a timely manner and the extent to which all parties have engaged in consultation and negotiation in good faith. We consider this a relevant factor for the purposes of section 138(2)(h) in a circumstance where a major capital expansion is planned. We will therefore welcome detailed information on the progress and process of negotiations, as part of submissions after this draft decision, and in response to any subsequent discussion papers we may publish.

We look forward to seeing the outcome of discussions between the parties.

Summary 8.7

Queensland Rail's proposed West Moreton reference tariff is not appropriate to approve.

An appropriate way for Queensland Rail to amend the 2025 DAU may be to escalate the 2020 undertaking reference tariff by CPI, if Queensland Rail and its customers are unable to agree on key aspects of a building blocks approach, including volumes, service standards and a reasoned and prudent strategy for capital spending and maintenance.

9 Tariff building blocks

As discussed in chapter 8, given the present lack of a reasoned and prudent strategy for investment in and maintenance of the network, and the uncertainty about both contracted volumes and available capacity, we do not consider it appropriate to approve the reference tariff proposed by Queensland Rail for access to the West Moreton system by coal services.

Therefore, we do not seek in this chapter to provide an assessment of the efficiency of each of the building blocks to form a view on the reference tariff, as has been customary in draft decisions on previous DAUs.

We instead provide contextual and technical information, where possible, that may assist Queensland Rail and its customers in working towards agreed positions for West Moreton, including a capital and maintenance strategy. This information includes engineering advice from our consultant, Arcadis, on which we have not yet formed a view.

For clarity, the information in this chapter does not represent our considered views. Rather, it should be viewed as information that we consider may be relevant for stakeholders to reach a consensus position on the building blocks, tariffs and related matters in schedules D and E of the 2025 DAU.

We will have regard to submissions, the advice from Arcadis and our own analysis, in assessing the efficient levels of each of the building blocks, should it be necessary for us to do so.

Overview

Queensland Rail proposal	Clause	QCA comment
Coal volumes		
Forecast volumes of 9.6 mtpa, based on expected contracted tonnages. ³⁸³	Schedule D	Queensland Rail should agree with users appropriate volume forecasts. See section 9.2.
Opening regulatory asset base		
Opening West Moreton RAB ³⁸⁴ of \$535.2m, ³⁸⁵ of which \$446.2m should be allocated to coal services. ³⁸⁶	Schedule D	It may be appropriate to approve the opening RAB, once it is adjusted to reflect updated AU2 capital expenditure, and changes in system capacity. See section 9.3.
WACC		
An indicative WACC of 7.39%. ³⁸⁷	Schedule D	We are minded to approve a rate of return based on Queensland Rail's proposal with an updated risk-free rate and cost of debt. See section 9.4.

Building blocks and tariff (Schedule D and E) – summary

³⁸³ Queensland Rail, sub. 1, p. 9.

³⁸⁴ As at 1 July 2025.

³⁸⁵ Unless stated otherwise, all figures in this chapter are expressed in \$2025-26 and exclude interest during construction where applicable.

³⁸⁶ Queensland Rail, sub. 1. p. 12.

³⁸⁷ Queensland Rail, sub. 1, p. 10.

Capital expenditure		
Capital expenditure of \$346.9m meets the four-fold increase in demand. ³⁸⁸	Schedule D	We would be minded to approve a capital indicator that is agreed between Queensland Rail and its users. See section 9.5.
Asset lives		
New assets to be depreciated over 14 years, and existing assets over 19 years. ³⁸⁹	Schedule D	Queensland Rail's proposed asset lives may be appropriate to approve. See section 9.6.
Appreciation		
The RAB should be escalated by inflation, forecast to be 3%, reducing to 2.5% for the final 2 years of the AU3 period. ³⁹⁰	Schedule D	It may be appropriate to approve the proposed escalation of the RAB. See section 9.7.
Maintenance expenditure		
Maintenance expenditure of \$172.5m over the AU3 period. ³⁹¹	Schedule D	It may be appropriate to approve maintenance expenditure that is agreed between Queensland Rail and its users. See section 9.8.
Operating expenditure		
Operating expenditure of \$85.3m over the AU3 period. ³⁹²	Schedule D	It may be appropriate to approve operating expenditure that is agreed between Queensland Rail and its users. See section 9.9.
West Moreton system reference tar	iff	
Queensland Rail proposed a headline reference tariff of \$32.63/'000 gtk. ³⁹³	Schedule D	Queensland Rail and its customers should agree on key aspects of a building blocks approach, including volumes, service standards and a reasoned and prudent strategy for capital spending and maintenance. See section 9.10.
Metropolitan system reference tari	if	
Escalate the existing Metropolitan system reference tariff by CPI. ³⁹⁴	Schedule D	It may be appropriate to approve the proposed metropolitan system tariff. See section 9.11.

9.1 Building blocks approach to regulatory pricing

While the analysis in this chapter is provided for information, our general approach to regulatory pricing for the West Moreton coal reference tariff has been to use a building blocks approach, which was used by Queensland Rail to develop its proposed total revenue requirement. The total revenue requirement is usually calculated to recover building blocks including:

[•] a return on assets (WACC) from a regulatory asset base (RAB)

³⁸⁸ Queensland Rail, sub. 1, p. 10.

³⁸⁹ Queensland Rail, sub. 1, pp. 10-11.

³⁹⁰ Queensland Rail, sub. 1, p. 11.

 ³⁹¹ Queensland Rail, sub. 1, p. 11.
 ³⁹² Queensland Rail, sub. 1, p. 11.

³⁹³ Queensland Rail, sub. 1, p. 11.

³⁹⁴ Queensland Rail, sub. 1, p. 59.

- a return of assets from the RAB (depreciation)
- allowances for:
 - maintenance
 - operating expenses
 - taxation.

The reference tariff is then assessed as recovering Queensland Rail's costs of providing access for West Moreton coal services, and is split into two parts:

- a weight and distance-based component (AT1), charged per gtk
- a fixed component (AT2), charged per train path.

The Metropolitan tariff was assessed previously using a 'proxy approach' that relies on prices derived for the coal services that use the West Moreton system (see section 9.10). It is also a two-part tariff.

9.2 Coal volumes

West Moreton is a mixed system serving a variety of freight, passenger and coal traffics. The bulk of services and tonnage using the West Moreton system route service are coal. Despite significant engineering challenges, the West Moreton system has been able to support volumes from three coal mines operating on the line, transporting a peak of 7.8 mtpa in 2011-12 (see Figure 8).



Figure 8: Historical volumes on the West Moreton system

Source: Queensland Government, <u>Coal industry review statistical tables</u>, Coal production data by mine, coal type and financial year, Open Data Portal, accessed 2 April 2024. Figures from financial year end 2011-2015 were accessed as part of the DAU2 decision and archived.

Queensland Rail projected that coal volumes will ramp up to 9.6 mtpa during the DAU3 period.³⁹⁵ This is much higher than the volume the West Moreton system currently carries, and more than the peak volume the system has carried historically. We have considered Queensland Rail's volume forecasts for AU3 from the perspective of:

- forecast demand during AU3 (section 9.2.1)
- the capacity of the West Moreton system (section 9.2.2).

9.2.1 Forecast demand

Queensland Rail has proposed a four-fold increase in coal volumes in the AU3 period, compared with the volumes used to assess the 2020 undertaking tariffs. It forecasts demand will ramp up to 9.6 mtpa in 2027-28 and continue at this level for the remainder of the 2025 undertaking period (Table 3).³⁹⁶ Current demand on the West Moreton system is significantly below Queensland Rail's forecast of 8.2 mtpa in the first year of AU3 (2025-26). While Queensland Rail indicated in its explanatory document that it expected contracted tonnages to exceed 4.1 mtpa in the later years of the AU2 period, at the time of writing it is not clear that this has happened.³⁹⁷

Table 3: Queensland Rail's forecast annual volumes (mtpa)

2025-26	2026-27	2027-28	2028-29	2029-30
8.2	9.5	9.6	9.6	9.6

Source: Queensland Rail, sub. 1, p. 12.

While we understand these demand forecasts reflect information provided by users, the volume forecast must be considered in light of the potential uncertainties that could affect the operations of users and their level of demand during this period. We address these demand-side uncertainties in three categories:

- mine-specific factors
- coal industry uncertainty
- the potential impact of the reference tariff.

Mine-specific factors

Queensland Rail has forecast tonnages based on expected volumes from the three mines operating on the West Moreton system, namely:

- Cameby Downs
- New Acland Stage 3
- Wilkie Creek.

The mine operators have contracts to transport varying volumes of product to port using Queensland Rail's West Moreton route service (including traversing the Metropolitan system). These contracted volumes, along with advice from miners as to their future operations, are the basis of the expected demand underpinning Queensland Rail's coal volume forecasts. However, none of the contracts currently in place covers the entire five-year 2025 undertaking period, as evidenced by

³⁹⁵ Queensland Rail, sub. 1, p. 12.

³⁹⁶ Queensland Rail, sub. 1, p. 12.

³⁹⁷ AU2 includes a recovery mechanism (tariff reset) requiring Queensland Rail to submit a draft amending access undertaking if contracted volumes and corresponding railings exceed 4.1 mtpa. As of the time of writing, no DAAU has been submitted. Queensland Rail, sub. 1, p. 55.

Queensland Rail's proposal to have triggers to revise work programs and tariffs should contracts not be renewed.³⁹⁸

Another complication is that the Wilkie Creek mine was in receivership as we prepared this draft decision, although we understand it had resumed railing, albeit at reduced volumes.³⁹⁹ Yancoal stated that due to this development since the 2025 DAU was lodged we should:

keep a watching brief on the prospects of New Wilkie continuing as the DAU3 process progresses before making a decision on the appropriate volume forecast.⁴⁰⁰

In addition, the New Acland mine is still in the development phase and is subject to a legal challenge over its water licence.⁴⁰¹ Aurizon Coal and Bulk, which operates the coal trains on the West Moreton system, said that given the uncertainty surrounding the Wilkie Creek and New Acland mines, 'forecast coal volumes will remain unreliable until these two matters are settled.'⁴⁰²

For Queensland Rail's demand forecasts to be achieved, the New Acland and Wilkie Creek mines will have to resolve their respective legal and financial situations and significantly ramp up production. Until this happens, it is not certain what West Moreton demand will be during the AU3 period, or how quickly demand will ramp up.

Coal industry uncertainty

Beyond mine-specific factors, changes in the overall thermal coal market have the potential to affect all the mines using the West Moreton system. We have previously observed that Queensland Rail is exposed to significant volume risk due to its dependence on thermal coal miners.⁴⁰³ In our recent guideline on climate change-related spending we said:

The long-term outlook for Queensland thermal coal may be ... problematic. In our final decision on Queensland Rail's 2020 draft access undertaking we specifically noted that Queensland Rail's West Moreton line coal customers (who produce thermal coal) are likely to be vulnerable to sustained economic shocks. But in 2022, prices for Queensland thermal coal achieved all-time record levels in real terms.⁴⁰⁴

Figure 9 charts the volatility of historical and forecast thermal coal prices.

³⁹⁸ See sections 8.5.2 and 8.5.5.

³⁹⁹ S Thompson, K Sood and E Rapaport, '<u>Houlihan Lokey starts New Wilkie clean-up job; flyers out</u>', *Australian Financial Review*, 28 February 2024.

⁴⁰⁰ Yancoal, sub. 9, p. 12.

⁴⁰¹ Queensland Rail, sub. 1, pp. 12-13.

⁴⁰² Aurizon Coal and Bulk, sub. 2, p. 70.

⁴⁰³ The AU2 decision was made in the context of only one coal miner having certainty of operating.

⁴⁰⁴ QCA, <u>Approach to climate change related expenditure</u>, final position paper, September 2023.

Figure 9: Average Australian thermal coal price forecast



Note: Figures for FY 2017-FY 2023 are actuals and for FY 2024-FY 2029 are forecast. Source: Department of Industry, Science and Resources, <u>Resources and Energy Quarterly</u>, Australian Government, March 2024.

Coal prices have declined significantly from their 2022 peak, and forecasters have diverging views on where they are headed. HoustonKemp, on behalf of Queensland Rail, considered that coal prices might rise during the energy transition,⁴⁰⁵ while Aurizon Coal and Bulk⁴⁰⁶ and Yancoal⁴⁰⁷ considered prices were drifting lower to a long-term pricing level. Yancoal highlighted that its operations had recently been dependent on unanticipated high coal prices and that a return to long-term levels could impact affordability.⁴⁰⁸

While it is unclear what coal price assumptions underpinned the volumes nominated by miners for the AU3 period, there is evidence that the coal industry is less profitable now than in the last couple of years.⁴⁰⁹

Stakeholders' submissions present contrasting views on whether Queensland Rail's volume assumptions are soundly based, in light of the proposed West Moreton reference tariff. Yancoal said:

thermal coal volumes are forecast to remain stable over the medium term (which Yancoal submits suggests a lower assessment of volume risk than QR has sought to assert).⁴¹⁰

Potential impact of the reference tariff

An additional factor that may impact the viability of customers' mines and, by extension, the level of user demand, is the level of the West Moreton reference tariff itself. Yancoal said that during the AU2 period:

Cameby Downs ... was effectively saved by unanticipated high thermal coal prices as a result of unforeseen trade issues and geopolitical events.⁴¹¹

⁴⁰⁵ Queensland Rail, sub. 1, attachment 4, p. 2.

⁴⁰⁶ Aurizon Coal and Bulk, sub. 2, p. 60.

⁴⁰⁷ Yancoal, sub. 9, p. 5.

⁴⁰⁸ Yancoal, sub. 16, p. 4.

⁴⁰⁹ In addition to pricing factors, national and international carbon dioxide reduction and net zero carbon emission policies will likely have an impact on the length of time coal mining operations remain viable in the longer term. We note that HoustonKemp considers that all mines on the West Moreton system will cease operation by 2050, with the first mine's reserves depleted by 2034. These issues are discussed further in section 8.5, which addresses asset stranding risk.

⁴¹⁰ Yancoal, sub. 16, p. 3.

⁴¹¹ Yancoal, sub. 16, pp. 4-5.

Yancoal noted that production costs had remained consistent across AU2 and leading into AU3,⁴¹²and that:

at the pricing levels QR is now proposing, there are material risks of mines ceasing production at the next expiry of their access agreement (when they can exit without triggering a material take or pay liability), compounding the affordability issues for any producers that remain ...⁴¹³

9.2.2 System capacity

It is in every stakeholder's interest, including Queensland Rail's, to make sure users have an accurate picture of the capacity that will be available to them during the AU3 period, as discussed in section 8.4. In this section, we provide information on what the actual capacity of West Moreton might be, both now and in the future.

Analysis from Arcadis indicates carrying Queensland Rail's forecast of 9.6 mtpa of coal on the West Moreton system would require approximately 93 loaded train paths per week.⁴¹⁴ While this is less than the maximum 97 train paths Queensland Rail states can be provided for in the MTP for coal traffic, this volume represents a significant increase compared to current operations. For example, the current MTP only provides for 43 train paths per week for coal traffic, giving volumes of approximately 4.4 mtpa.⁴¹⁵ Queensland Rail's current Western system information pack suggests the allowable gross tonnes for the West Moreton system are between 4.5 and 7 mtpa.^{416,417} Achieving the forecast tonnages for the 2025 undertaking period would require significant increases in these figures. This may prove challenging from an engineering perspective and represent a significant increase in noise impact for the community compared to current operations.

Stakeholders were concerned the forecast system capacity might not be achieved and asked for greater transparency around Queensland Rail's proposal.⁴¹⁸ Aurizon Coal and Bulk said its own analysis showed it was possible to schedule sufficient trains for 9.6 mtpa. However, it noted that this analysis did not include any allowance for system losses such as availability of rolling stock, and mine and port scheduling. It applied a system loss rate of 15% and suggested the capacity of the network might be 8 mtpa.⁴¹⁹ As noted in section 9.2, this figure is in line with the maximum annual volumes achieved on the West Moreton system in recent history.

Analysis by Arcadis shows that Queensland Rail faces a number of challenges to deliver the realworld availability of 97 return paths on an ongoing basis.⁴²⁰ Key drivers of the potential capacity include:

- engineering factors
- environmental factors
- scheduling factors
- ad hoc capacity.

⁴¹² Yancoal, sub. 9, p. 4.

⁴¹³ Yancoal, sub. 16, pp. 4-5.

⁴¹⁴ Assuming 2,000 tonnes of product per train.

⁴¹⁵ This excludes ad hoc capacity, which is discussed below.

⁴¹⁶ The Rosewood to Toowoomba section is currently rated at 7 mtpa, and the section from Toowoomba to Dalby is currently rated at 4.5 mtpa.

⁴¹⁷ Queensland Rail, <u>Western System Information Pack</u>, August 2016.

⁴¹⁸ Aurizon Coal and Bulk, sub. 2, pp. 6-7; New Hope, sub. 5, p. 8; Yancoal, sub. 9, p. 19 and sub. 16, p. 7.

⁴¹⁹ Aurizon Coal and Bulk, sub. 2, pp. 61-63.

⁴²⁰ Arcadis, Review of West Moreton System Costs and Other Technical Matters in Queensland Rail's 2025 DAU, May 2024, pp. 17-22.

Engineering factors

Sections of the West Moreton system, particularly those using 41kg/m rail, and the Toowoomba Range, require significant maintenance. With the relatively low utilisation of the system at present, this maintenance requirement can be fulfilled using spare train paths with minimal impact on overall rail traffic from these possessions, as half or fewer of the 97 weekly train paths assigned to coal are being used.

However, the existing strategy cannot be followed where the number of paths used increases to meet the forecast throughput of 9.6 mtpa, as that amount of traffic places far tighter limits on the number of paths available for maintenance. It appears that Queensland Rail's record proposed capital program is, in some respects, aimed at increasing system resilience to reduce the impact on the system of train traffic, thereby reducing the number of train paths required for maintenance.

While this strategy may be successful in delivering a system that can provide the required number of paths to transport 9.6 mtpa, analysis by Arcadis raises some doubts over whether this level is sustainable.⁴²¹ Even with the capital works proposed (see section 9.5) the increased coal traffic will require higher levels of maintenance. Given the system has not operated at this capacity previously, there is also the possibility that unanticipated new maintenance needs will emerge.⁴²² Completing the required level of maintenance in an environment where the number of available maintenance possessions is effectively halved without affecting the number of paths available to be used by access holders will be challenging for Queensland Rail.⁴²³

Environmental factors

The two principal environmental factors that affect the reliability of the network and the number of paths available to coal traffic are heat restrictions and rainfall events.

Some sections of the West Moreton system are subject to speed restrictions where the ambient temperature exceeds 32 degrees Celsius, and closure where temperatures exceed 35 degrees Celsius. These restrictions are necessary due to the increased risk to the network and traffic related to both weakening, and expansion-related movement, of rail at high temperatures. These sections are generally where 41 kg/m rail is used in the network. Queensland Rail proposes to replace the rail in many of these sections to reduce the need for restrictions on operation. However, it is not possible for a railway to eliminate all heat-related restrictions. General engineering standards provide for system shutdown at 40 degrees ambient temperature. It follows that while rail replacement may enable Queensland Rail to reduce heat-related speed restrictions by imposing them only at higher temperatures, it is not possible to eliminate them.

In addition to heat-related environmental factors, high rainfall events are cause for shutdown on the Toowoomba Range. Due to the topography of the range, high rainfall events can lead to rock or tree falls that could cause derailments. Where cumulative rainfall exceeds 30 mm in a 24-hour period, the range crossing must be closed for an inspection to check the range crossing is safe for rail traffic. Even where no issues are identified that need rectification works, these inspections take a minimum of between 3 and 8 hours, during which trains cannot operate.

⁴²¹ Arcadis, Review of West Moreton System Costs and Other Technical Matters in Queensland Rail's 2025 DAU, May 2024, pp. 21-22.

⁴²² Arcadis, Review of West Moreton System Costs and Other Technical Matters in Queensland Rail's 2025 DAU, May 2024, pp. 21-22.

⁴²³ While it is a factor beyond the scope of the DAU3 decision, we also note that Arcadis predicts greater difficulties beyond 2030 in maintaining this capacity when the five-year 'honeymoon period' of lower maintenance from the capital expenditure ends.

Proactive maintenance and expenditure on improving drainage and slope stabilisation, as well as removing potential hazards such as trees and unstable rock formations can reduce the likelihood of remediation works after rainfall events, but the need for inspections will remain.

Scheduling factors

Coal traffic on the West Moreton system must be scheduled with traffic carrying other types of freight, including that from the south-west system and from Toowoomba. The trains must also pass through the Metropolitan system, which has scheduling windows at peak times dedicated to passenger transport.

Stakeholders said they were concerned about capacity issues in the Metropolitan system arising from closures related to Cross River Rail,⁴²⁴ as well as the potential for increased passenger rail traffic from Cross River Rail to reduce paths available for coal traffic.⁴²⁵

The scheduling issues affecting operational capacity are further complicated by the environmental factors outlined above. For example, applying heat-related speed restrictions not only increases transit times – it also affects the subsequent scheduling of the train through the Toowoomba Range crossing and the Metropolitan system. Given the range crossing is a bottleneck for traffic across multiple cargo types and multiple systems, closing the range for necessary safety inspections after high rainfall events affects all traffic using the range and how trains are scheduled through the Metropolitan system.

Further complicating scheduling is an above-rail factor – that is, the limited number of locomotives available for running with a 15.75 tonne axle load on the narrow gauge Queensland Rail network, which restricts scheduling flexibility. Arcadis advised that due to the limited market for narrow gauge locomotives it can take years to procure them and would potentially require construction of additional maintenance and storage facilities, meaning that high tonnages would be required over the long term to provide a return on investment.⁴²⁶

Ad hoc capacity

The volume forecast by Queensland Rail relates to contracted capacity in the MTP, which it says is a theoretical maximum of 97 train paths per week. In addition to the contracted capacity for coal in the MTP, there are 14 return paths which are reserved for non-coal traffic, such as grain, livestock and passenger trains.

Where these preserved paths are not used,⁴²⁷ they are available to coal traffic in the DTP on an ad hoc basis, or to provide more scope for maintenance activities. As such, ad hoc capacity may act as a buffer in certain periods of the year to conduct greater maintenance or allow for catch-up of any paths lost in the MTP. However, given it can only be provided on an ad hoc basis, this potential capacity provides less certainty to coal customers, which may affect their willingness to underwrite the cost of providing it.

⁴²⁴ Aurizon Coal and Bulk, sub. 2, pp. 67-70.

⁴²⁵ Yancoal, sub. 16, p. 5.

⁴²⁶ Arcadis, Review of West Moreton System Costs and Other Technical Matters in Queensland Rail's 2025 DAU, May 2024, p. 19. ⁴²⁷ Agricultural traffic, such as grain, is highly seasonal.

9.2.3 Forecast coal volumes

As we have outlined, there are multiple factors that present significant challenges for Queensland Rail in consistently delivering the tonnages it has forecast in DAU3. We note the concerns from stakeholders around the forecast volumes, particularly regarding their real-world achievability.⁴²⁸ We also note the advice from Arcadis in relation to Queensland Rail's forecast coal volumes:

While our engineers believe that the scenario is theoretically achievable, the demanding nature of this rail network, with high tonnage and operational utilisation, will present significant challenges despite the proposed expenditure.⁴²⁹

If the forecast coal volumes are contracted but cannot be achieved, it will be a problem for both Queensland Rail and its customers. The volume forecasts drive both Queensland Rail's work programs and its users' capital investments in their mines.

West Moreton system capacity is a potential limiting factor in miners' ability to deliver to their customers and achieve forecast cashflows for their projects. Investments that are made by miners to support production that exceeds the total system capacity actually available may impact the viability of their operations, which in turn may increase the risk of asset stranding for Queensland Rail. System capacity shortfalls may also slow the rate at which the resource is depleted, potentially affecting the mine life predictions that underpin Queensland Rail's depreciation proposal.⁴³⁰

And, given wear and tear on the rail infrastructure is closely correlated with tonnage carried, it is likely not to be efficient to invest to serve a volume that cannot be achieved, or is not needed, when a more realistic volume forecast would require lower investment.

The most efficient outcome is likely to be found where both Queensland Rail and users, who are best placed to understand all the uncertainties and their own risk and service preferences, work toward agreed volume and capacity estimates that can be presented to us to be used in preparing our final decision.

Summary 9.1

It may not be appropriate to approve Queensland Rail's proposed coal volumes unless it is clear they will be contracted and there is capacity to deliver them. This is a matter best resolved by negotiation between Queensland Rail and its customers.

9.3 Asset base

Queensland Rail's asset base reflects historical regulatory treatment of long-lived capital investments, and the asset base methodology has remained little changed since the AU1 decision. Queensland Rail's DAU proposed an opening unallocated system regulatory asset base (RAB) of \$535.2 million, further capital spending of \$346.9 million during the five-year regulatory period (see

⁴²⁸ Aurizon Coal and Bulk, sub. 2, p. 61.

⁴²⁹ Arcadis, Review of West Moreton System Costs and Other Technical Matters in Queensland Rail's 2025 DAU, May 2024, p. 5.

⁴³⁰ Queensland Rail, sub. 1, attachment 4, p. 3.
section 9.5), and maintenance spending of \$182.6 million (see section 9.8). This section considers the treatment of the RAB, including:

- the opening regulatory asset value (section 9.3.1)
- RAB optimisation (section 9.3.2)
- asset allocation (section 9.3.3).

9.3.1 Opening regulatory asset base

Queensland Rail has proposed an opening coal allocated RAB of \$446.2 million for the 2025-26 financial year. This is derived from a common network RAB of \$535.2 million, with the assumed 97 path coal allocation from a total of 113 weekly coal paths available for the West Moreton system (85.8%).⁴³¹

Queensland Rail has derived the 2025-26 RAB from the current AU2 RAB. Its estimate includes the capital expenditure for 2020-21 and 2021-22 period that we approved as well as capital indictor amounts for the remaining AU2 period.⁴³² Queensland Rail has applied historical CPI figures from the Australian Bureau of Statistics alongside our inflation forecasting approach (section 9.7).

Table 4: West Moreton common network asset base roll-forward proposed by Queensland
Rail (\$000's)

	2020-21	2021-22	2022-23	2023-24	2024-25	2025-26
Opening asset value	388,912ª	419,083	469,185	506,476	521,000	535,230
Capital expenditure	23,582	33,348	23,595	16,937	17,242	
Inflationary gain	19,746	31,783	30,450	15,446	15,887	
Less depreciation	13,157	15,029	16,754	17,860	18,899	
Closing asset value	419,083	469,185	506,476	521,000	535,230	

a The opening asset value includes coal-only infrastructure valued at \$14.5 million. Source: Queensland Rail, sub. 1, p. 17.

Source: Queensiand Rail, sub. 1, p. 17

Since it lodged the 2025 DAU, Queensland Rail has submitted its capital expenditure claim for 2022-23, to the amount of \$44.9 million,⁴³³ which significantly exceeds the \$23.6 million used to calculate the opening 2025-26 RAB. As such, the opening RAB will need to be re-estimated, taking into account our final decision on the latest capital claim, as well as any further changes during the AU2 period. Queensland Rail also said it would propose additional capital spending in the final years of the 2020 undertaking period, above amounts projected in the AU2 capital indicator. This can be expected to increase the capital carry-over account, which will carry forward any differences between actual capital expenditure and the spending included in the AU2 capital indicator into the opening RAB for AU3.

More broadly, the capital works programs proposed in the 2025 DAU (discussed in section 9.5) will build on the foundation of works completed during the AU2 period. As such, these works will have

⁴³¹ Allocation method for pre-1995 assets is 97 paths out of 137 (84%) and for coal only assets is 1 path out of 1 (100%). The overall effective allocation to coal for 2025-26 is 83.37%. For more on how the allocations are derived, see QCA, Queensland Rail's Draft Access Undertaking, decision, June 2016, pp. 119-146.

 ⁴³² Queensland Rail has confirmed that this opening regulatory asset base amount does not include the proposed additional capital expenditure planned for the AU2 period over and above that included in the AU2 capital indicator.
 ⁴³³ Including interest during construction.

flow-on effects on the capital, maintenance and operational programs required during the AU3 period.

9.3.2 RAB optimisation

Queensland Rail has proposed a significant increase in capital expenditure during the AU3 period, compared with AU2 (see section 9.5). Queensland Rail said this was needed to meet the significant increase in forecast coal volumes (see section 9.2). Queensland Rail states that:

For the DAU3 period, Queensland Rail has proposed efficient capital costs for the West Moreton System having regard to the age and condition of the network, and the volumes proposed to be hauled over a network that was not originally designed for this purpose.⁴³⁴

Aurizon Coal and Bulk⁴³⁵ and Yancoal⁴³⁶ (with support from New Hope⁴³⁷) said the historical claimed capacity of the West Moreton system might have been overstated, and asset values might reflect past capacity estimates that were no longer justified. Yancoal said:

[Queensland Rail's] very significant capital expenditure spend, justified largely based on the inadequate state of the current rail infrastructure, should give rise to serious questions about whether the existing regulatory asset base should be materially optimised downwards ...⁴³⁸

HoustonKemp, in a report for Queensland Rail, said optimising assets would create an undesirable regulatory precedent. In its opinion:

the write down of assets on the West Moreton system is inappropriate, inconsistent with objectives of the QCA Act, and could have material and significant consequences for investment in sectors subject to economic regulation throughout Queensland and Australia.⁴³⁹

We think this is unlikely but would be interested in stakeholder views on which regimes would most likely be affected and why.

West Moreton users raised asset optimisation in the context of achieving a reference tariff that they considered affordable. While asset optimisation could act to lower reference tariff levels, as New Hope said,⁴⁴⁰ there are multiple factors that affect the level of the reference tariff that Queensland Rail and users are able to negotiate on to enable the efficient use of the network. For example, in AU2, Queensland Rail charged an affordable tariff, along with a loss capitalisation mechanism to support the continued use of the network.⁴⁴¹

Some agreed adjustment to the RAB would be an approach that, if it chose to, Queensland Rail could propose with regard to the interests of users.

⁴³⁴ Queensland Rail, sub. 1, p. 26.

⁴³⁵ Aurizon Coal and Bulk, sub. 2, p. 71.

⁴³⁶ Yancoal, sub. 9, pp. 14-15.

⁴³⁷ New Hope, sub. 2, p. 1.

⁴³⁸ Yancoal, sub. 9, p. 14.

⁴³⁹ Queensland Rail, sub. 14, attachment 1, p. ii.

⁴⁴⁰ New Hope, sub. 12, p. 1.

⁴⁴¹ We note that in the context of accelerated depreciation HoustonKemp said that '[the] existing RAB represents sunk costs. As such, promoting efficient investment in the West Moreton system is not a relevant consideration' (Queensland Rail, sub. 1, attachment 5, p.17). It could equally be argued that accelerating the return of assets on effective life-expired assets is economically inefficient because users are paying for assets that have no practical economic utility, especially where this leads to an unaffordable reference tariff.

9.3.3 Asset allocation

We are minded to approve an opening RAB that has been updated to reflect differences in capital expenditure during the AU2 period. While we are also minded to approve maintaining the AU2 allocation of coal assets based on the availability of 97 train paths, we invite stakeholder views on whether allocating of 97 out of 113 train paths remains appropriate, given the information emerging about what the actual capacity of the system may be.

Summary 9.2

It may be appropriate to approve an opening RAB that has been updated to reflect differences in capital expenditure during the AU2 period and uses a methodology of allocation between coal and non-coal users that reflects any changes in system capacity.

9.4 Rate of return

We consider that Queensland Rail's proposed indicative rate of return⁴⁴² of 7.39% is reasonable for determining the reference tariffs to apply to the coal handling services operating on the West Moreton and Metropolitan systems. We would be minded to approve a rate of return based on Queensland Rail's proposal, with updates to reflect:

- an updated risk-free rate, calculated using an averaging period nominated in advance by Queensland Rail
- an updated cost of debt, using the average of 12-monthly observations from April to March in advance of the upcoming regulatory period.

The methodologies that we consider appropriate for updating these two time-variant parameters are specified below.

We consider that this will provide for a rate of return that is commensurate with the regulatory and commercial risks faced by Queensland Rail in providing access to coal handling services on the West Moreton and Metropolitan systems.

In assessing the reasonableness of Queensland Rail's proposed rate of return for the AU3 regulatory period, we have:

- undertaken a bottom-up WACC estimate, based on our preferred methodology for calculating the WACC and each of the relevant individual parameters.
- considered the overall reasonableness of Queensland Rail's proposed rate of return.

Our approach to assessing Queensland Rail's proposed rate of return, including the methodology applied to calculate our bottom-up WACC estimate, is consistent with the approach outlined in our rate of return review.⁴⁴³ New Hope and Yancoal supported estimating market parameters using the methods identified in our rate of return review. However, both New Hope and Yancoal considered

⁴⁴² The rate of return compensates the investor for the time value of money and risk that they face in providing the assets that deliver the services that are subject to the regulatory regime.

⁴⁴³ QCA, <u>Rate of return review</u>, final report, version 3, February 2024.

that firm-specific parameters required a reassessment from the parameters applied to calculate the WACC in AU2, in light of changes in the risk environment for the AU3 regulatory period.⁴⁴⁴

In making this draft decision, we have considered Queensland Rail's exposure to risks afresh as part of our bottom-up WACC estimate and our overall consideration of the reasonableness of Queensland Rail's proposed rate of return (see section 9.4.2). In doing so, we have had regard to the extent that Queensland Rail's proposed regulatory arrangements mitigate risk or efficiently allocate risk to another party (see Box 3).

While we have calculated an indicative bottom-up WACC of 6.70% for Queensland Rail for the AU3 regulatory period, we consider that Queensland Rail's rate of return proposal provides Queensland Rail with a reasonable rate of return for the AU3 regulatory period.

Queensland Rail's approach for calculating its proposed rate of return is consistent with that applied to calculate the approved rate of return for the 2020 access undertaking. Differences between the Queensland Rail's proposed rate of return to the rate of return approved as part of the 2020 access undertaking largely reflect timing differences in the calculation of specific time-variant WACC parameters. In updating these time variant parameters for AU3, Queensland Rail has applied the relevant methods and values set out in our rate of return review.

Furthermore, our comparison with other regulatory decisions supports Queensland Rail's proposed WACC being reasonable, and sufficient to promote efficient investment in the infrastructure required to provide the West Moreton route service.

Accordingly, our draft decision is that Queensland Rail's proposed WACC, adjusted for updated time-variant numbers in 2025, is appropriate to approve, on the basis that it will provide a return commensurate with the regulatory and commercial risks of providing access. We consider it is also in the interests of customers to provide Queensland Rail with a return that supports investment in the network to provide the access they require.

⁴⁴⁴ New Hope, sub. 5, pp. 14, 20; Yancoal, sub. 9, pp. 7–8. More specifically, Yancoal submitted that where the systematic risks faced by Queensland Rail in proving access to the West Moreton users had materially changed, the asset beta should also undergo a corresponding change. New Hope submitted that changes to the customer and contract profile on the West Moreton system meant that Queensland Rail would face a lower degree of systematic risk.

Box 3: Mitigation and allocation of risk under Queensland Rail's 2025 DAU proposal

Queensland Rail's regulatory regime for the West Moreton route service includes a number of measures that mitigate its risk. A key feature is 100% take-or-pay on contracted volumes. These take-or-pay contracts provide a revenue floor for Queensland Rail and mitigate much of its short-run volume risk.

Queensland Rail also benefits from a building blocks-based reference tariff, that is assessed for, among other things, whether Queensland Rail receives a return that meets the efficient costs of providing access and includes a return on investment commensurate with the regulatory and commercial risks. The reference tariff is not affected by fluctuations in the coal price received by Queensland Rail's customers.

Queensland Rail's take-or-pay contracts and reference tariffs are only two of the features of the regulatory framework that materially buffer its cash flows from cyclical movements in the economy. A number of other mechanisms, which feature in the QCA Act or have been proposed as part of the 2025 DAU process, contribute to insulating Queensland Rail's revenue stream from external shocks, including:

- the ability of Queensland Rail to recover capital expenditure where it might differ from forecast levels, should we approve it
- security arrangements for Queensland Rail, with access holders required to pay relinquishment fees (80% of the remaining access charges) in the event that they surrender their contracted volumes
- limited asset optimization, restricting the scope for Queensland Rail's RAB to be lowered
- a capacity investment framework that provides Queensland Rail with the ability to secure capital underwriting
- accelerated depreciation, which is proposed to align the recovery of Queensland Rail's investments with its forecasts of remaining mine lives
- the right of Queensland Rail to submit a draft amending access undertaking under the QCA Act.

9.4.1 Bottom-up WACC analysis

We have undertaken a bottom-up WACC analysis to evaluate Queensland Rail's proposal.

We have calculated an indicative bottom-up WACC of 6.70% for Queensland Rail for the AU3 regulatory period (see Table 5).

To calculate an indicative bottom-up WACC for Queensland Rail, we have used a nominal, post-tax WACC⁴⁴⁵ based on our estimates of individual WACC parameters. Our assessment of the individual parameters used to generate our bottom-up estimate is outlined below.

⁴⁴⁵ Our approach uses the Officer WACC3 model and estimates the WACC for a benchmark firm, rather than the regulated firm's actual WACC.

Table 5: Parameters used to calculate our indicative bottom-up WACC estimate

WACC parameter	QCA preliminary estimate
Risk-free rate [#]	3.37%
Market risk premium	6.3%
Asset beta	0.48
Equity beta	0.71
Gearing	40%
Cost of debt [#]	4.95%
Gamma	0.484
Indicative bottom-up WACC estimate	6.70%

Placeholder values have been adopted for the risk-free rate and the cost of debt to estimate an indicative bottomup WACC estimate for Queensland Rail's AU3 regulatory period. These placeholder values will be updated prior to our final decision.

Risk-free rate

We consider it reasonable to adopt Queensland Rail's proposed risk-free rate⁴⁴⁶ of 3.37% as a placeholder to determine our bottom-up WACC estimate for Queensland Rail's AU3.

Our approach to estimating the risk-free rate for Queensland Rail is to:

- use 10-year Australian Government (nominal) bond yields as the proxy for calculating Queensland Rail's risk-free rate.⁴⁴⁷ These bonds have very low default risk and are also highly liquid. We consider using long-term Australian Government bonds reflects the requirements of investors and lenders who, in relation to long-lived infrastructure assets, will deploy equity over the entire life of the asset, rather than over any given regulatory period
- average the yields over a period nominated in advance by the regulated entity that is between 20 and 60 business days in length, ending as close as reasonably possible to the commencement of the regulatory period. Averaging the daily risk-free rate over a short period will manage the risk of unanticipated volatility from one-off shocks.

Queensland Rail proposed an indicative risk-free rate of 3.37%, applying this methodology and using a 20-business-day averaging period ending April 2023.⁴⁴⁸

We consider it appropriate for the risk-free rate to be updated using this methodology prior to the start of the AU3 regulatory period, using an averaging period nominated by Queensland Rail. Queensland Rail is to propose the timing and length of its nominated averaging period in advance of the averaging period commencement date. The nominated averaging period should:

- be between 20 and 60 business days to provide a rate that reflects current conditions, but smooths the effects of temporary shocks
- commence as close as reasonably practical to the start of the regulatory period (ending before commencement of the period) to capture current rates.

⁴⁴⁶ The risk-free rate is the rate of return an investor would expect to receive on an asset with zero default risk. It compensates an investor for the time value of money.

⁴⁴⁷ Our approach is to use daily Australian Government bond rates published by the Reserve Bank of Australia (RBA) (F2 table) to estimate the risk-free rate, converting the daily yields into an effective annual rate using the conversion method outlined in Appendix F of our rate or return review. See QCA, <u>Rate of return review</u>, final report, version 3, February 2024.
⁴⁴⁸ Queensland Rail, sub. 1, p. 82.

Market risk premium

We consider it reasonable to apply a market risk premium⁴⁴⁹ of 6.3% to determine our bottom-up WACC estimate for Queensland Rail's AU3.

Our approach to estimating the market risk premium for Queensland Rail is based on the Ibbotson method. The market risk premium is not directly observable. We consider that the Ibbotson method is likely to provide a plausible indication of the average market risk premium investors can expect to receive in normal conditions. The Ibbotson method assumes that investors use historical excess returns data to inform their expectations of achievable future returns.

In applying the lbbotson method, we consider it appropriate to use a sampling period from 1958⁴⁵⁰ to February 2024. We consider this provides for a data series that contains both a relatively large number of observations and consists of high-quality data.

Queensland Rail proposed a market risk premium of 6.5% for the AU3 regulatory period, which was estimated using our preferred methodology with data to December 2022.⁴⁵¹

Beta

We consider it reasonable to apply an asset beta⁴⁵² of 0.48 and an equity beta⁴⁵³ of 0.71 to determine our bottom-up WACC estimate for Queensland Rail's AU3.

An appropriate beta estimate will reasonably compensate Queensland Rail for the systematic risk that it faces in providing access to coal handling services on the West Moreton system. To determine a reasonable beta estimate for Queensland Rail, we have had regard to Queensland Rail' s business and operating environment, as well as the features of the proposed regulatory framework that reduce or increase its exposure to systematic risk.

Box 4 outlines a number of key characteristics that we consider will influence Queensland Rail's exposure to systematic risk.

⁴⁴⁹ The market risk premium is the additional return that an equity investor requires, to be compensated for the risk of investing in a fully diversified portfolio of risky assets, relative to purchasing a risk-free asset.

⁴⁵⁰ 1958 is the first year for which the Sydney All Ordinary Shares price index was calculated on a daily, rather than a retrospective basis, and 1958 is also (approximately) the first year for which marketable short-term government securities (e.g. Treasury notes) were issued. See T Brailsford, J Handley and K Maheswaran, 'Re-examination of the historical equity risk premium in Australia', *Accounting & Finance*, vol. 48, 2008, pp. 85-86.

⁴⁵¹ Queensland Rail, sub. 1, p. 83.

⁴⁵² The asset beta (or unlevered equity beta) of an entity is a measure of the volatility of returns from a firm's assets relative to the volatility of returns to the market as a whole – often referred to as systematic (or non-diversifiable) risk.

⁴⁵³ The equity beta (or levered asset beta) measures the movement of equity return of a business with the market return. It captures both the underlying systematic risk of Queensland Rail (relative to the risk of the market) and the debt funding to equity holders.

Box 4: Key characteristics influencing Queensland Rail's exposure to systematic risk

We consider that Queensland Rail's market position as the sole below-rail service provider for the coal-handling users that access the West Moreton system and the resulting regulatory framework afford a high level of revenue insulation. In this regard, we consider that various features of Queensland Rail's proposed regulatory arrangements will significantly dampen Queensland Rail's exposure to systematic risk (see Box 3).

However, Queensland Rail serves a limited number of coal handling customers on the West Moreton system. A smaller customer base increases the risk that where a customer's volume forecasts are not achieved, that lost revenue cannot be diversified across the pool of other customers. Queensland Rail submitted that the West Moreton system would have 3 coal-handling customers at most during the AU3 period.

Fundamentally, the competitiveness of coal producers accessing the West Moreton system will be a key determinant of Queensland Rail's exposure to volume risks in the longer term. Below-rail coal services will be sustained as long as end customers have a sustained demand for the output of their mines. Thermal coal export volumes from Australia are forecast to remain steady over the regulatory period.

Supporting this forecast for stable demand over the medium term, Aurizon Coal and Bulk submitted that the demand for Australian thermal coal was more influenced by Asian trade, which represented 83% of global seaborne import volume. Moreover, Aurizon Coal and Bulk noted that the average age of coal-fired generation capacity in Asia was only 14 years. Aurizon Coal and Bulk considered that countries with younger coal-fired generators were less likely to transition toward renewable power until existing assets were closer to expiry.

Importantly, the mines using the West Moreton route service produce thermal coal that is of relatively high energy content and relatively low ash content. Aurizon Coal and Bulk also noted that out of the 10 largest coal-producing countries, Australia had the third-lowest emissions intensity for producing and transporting coal.

Overall, we consider that Queensland Rail's coal handling customers on the West Moreton system have strong incentives to maintain production at a consistent level throughout the economic cycle. High shut-down and start-up costs give mines an incentive to continue to produce coal even in a low coal price environment, where market circumstances are expected to be temporary.

The characteristics of West Moreton system's customer base make demand for the service less responsive to fluctuating coal prices and economic cycles.

We have sought to estimate a beta reference point, to help guide our views on the appropriate equity beta for Queensland Rail. In doing so, we have considered observed betas obtained from a sample of sufficiently comparable listed firms as a benchmark to inform our estimate of a reasonable beta for Queensland Rail.

We consider that beta estimates obtained from a sample of regulated energy and water businesses and a sample of toll roads businesses provide useful reference points to help guide our decision on an appropriate beta for Queensland Rail. We consider that the systematic risk associated with providing access to coal handling services on the West Moreton system is:

- greater than the systematic risk of regulated energy and water businesses (see Box 5)
- less than the systematic risk of toll road businesses (see Box 6).

Box 5: Regulated energy and water businesses' exposure to systematic risk

We consider that regulated energy and water businesses have similar attributes to Queensland Rail's provision of services on the West Moreton system, including:

- services that are subject to cost-based regulatory regimes
- a customer base that has no alternative service options, which exhibits a resilient demand for the service through economic cycles.

New Hope considered that Queensland Rail had relatively limited exposure to fluctuations in domestic economic conditions, due to the availability of revenue protection mechanisms and the nature of customer demand. New Hope submitted that, amongst other things:

- demand for Queensland Rail below-rail services and regulated energy and water businesses were similarly inelastic to changes in domestic economic conditions
- revenue protection mechanisms applied to Queensland Rail and energy/water network businesses.

Despite there being some similarities between Queensland Rail and regulated energy and water businesses, we consider that the customer base of regulated energy and water businesses provides for more resilient demand, compared to coal handling services on the West Moreton system. Queensland Rail has a smaller and less diverse coal customer base than that of regulated energy and water businesses – making it more exposed to counterparty risk should a customer temporarily stop railing.

Relative to coal handling services on the West Moreton system, regulated energy and water businesses are typically not as reliant on risk-mitigating mechanisms such as long-term, take-or-pay-contracts, to insulate them from volume risk. Queensland Rail will be exposed to a higher level of volume risk in instances when customers have not contracted to high levels.

Box 6: Toll roads businesses' exposure to systematic risk comparator sample

Toll road businesses are not subject to a cost-based regulatory framework with fixed periodic reviews. While these businesses are often provided with a CPI-linked price cap, cash flow volatility may arise from market shocks and revenue might deviate from the costs of providing the service. Both New Hope and Yancoal submitted that toll road businesses typically did not have the same revenue protection mechanisms.⁴⁵⁴

Additionally, toll road businesses do not have contracting arrangements with customers and face competitive pressure from existing alternatives to toll road services. In contrast, Queensland Rail's coal handling customers on the West Moreton system have no feasible alternative transport options, and long-term contractual arrangements are a feature of Queensland Rail's regulatory compact. As such, Yancoal considered that toll road business did not have the same types of volume risk protections as Queensland Rail.⁴⁵⁵

We consider that Queensland Rail's coal handling customers on the West Moreton system have strong incentives to maintain their production at a consistent level throughout the economic cycle. In this regard, New Hope considered that toll roads businesses were likely to be much more elastic to changes in economic conditions, compared to demand on Queensland Rail's West Moreton system.⁴⁵⁶ However, toll roads have larger and more diversified customer bases than Queensland Rail which will to an extent mitigate exposure to economic downturns.

New Hope and Yancoal considered that Queensland Rail now had a risk profile that was more closely aligned to regulated energy and water businesses. In this regard, New Hope and Yancoal considered that Queensland Rail's exposure to risk had changed since the AU2 regulatory period, reflecting:

- changes in Queensland Rail's customer profile⁴⁵⁷
- Queensland Rail's proposed changes to the regulatory arrangements.^{458, 459}

Yancoal submitted that Queensland Rail was now closer to being relatively immune from volume risk and therefore more closely akin to the inelastic demand and guaranteed revenue outcomes that regulated electricity and water businesses faced.

In contrast, HoustonKemp (in a report commissioned by Queensland Rail) considered that Queensland Rail faced greater exposure to volume risk than regulated energy and water

⁴⁵⁸ New Hope and Yancoal noted that Queensland Rail had proposed an accelerated depreciation profile for AU3 based on an estimated weighted average mine life. Yancoal also referred to Queensland Rail's proposal to introduce a volume-based trigger for re-opening reference tariffs during the AU3 regulatory period.

Queensland Rail's 2025 Draft Access Undertaking

⁴⁵⁴ New Hope, sub. 5, p. 19; Yancoal, sub. 9, pp. 8-9.

⁴⁵⁵ Yancoal, sub. 9, pp. 8-9.

⁴⁵⁶ New Hope, sub. 5, p. 19.

⁴⁵⁷ New Hope and Yancoal noted that Queensland Rail was forecasting much higher contracted volumes. Yancoal also submitted that the New Acland Stage 3 mining lease had been approved with production volumes ramping up and the New Wilkie mine had reopened.

⁴⁵⁹ New Hope, sub. 5, pp. 14-17; Yancoal, sub. 9, pp. 8-9.

businesses. HoustonKemp stated that the risk to Queensland Rail if volume forecasts did not materialise or a mine did not come online remained significant. HoustonKemp considered that there was no evidence to suggest that the level of systematic risk faced due to exposure to the thermal coal export sector was materially different to that in the AU2 period, particularly given the increasing and uncertain shift towards decarbonisation.⁴⁶⁰

We do not consider that Queensland Rail's proposed amendments to the regulatory framework result in it having a risk profile that aligns with that of a regulated energy and water business.

Accelerated depreciation will mitigate Queensland Rail's exposure to the risk of volumes not being sustained by access holders, or reallocated to new coal-handling access seekers, over the long-term. However, we consider that Queensland Rail remains exposed to the risk of a customer's volume forecasts not materialising. This is particularly the case as none of its forecast volume is contracted for the full term of AU3, although this is offset by the fact that all three customers have substantial sunk costs, which they are unable to recoup without securing access from Queensland Rail. We consider that Queensland Rail will be exposed to a higher level of volume risk than regulated energy and water businesses, noting:

- regulated energy and water businesses are able to diversify the risk of a customer's volume forecast not materialising across a large pool of customers
- Queensland Rail is reliant on mechanisms in the regulatory framework, such as loss capitalisation, to address instances of forecast volumes not materialising.

In this regard, HoustonKemp stated that Queensland Rail's accelerated depreciation proposal did not protect Queensland Rail against broader revenue risks, as the depreciation allowance was based on a specific volume forecast, which might or might not materialise.⁴⁶¹

While the current volume forecasts for certain coal-handling customers on the West Moreton system have increased since the rate of return was assessed for the AU2 regulatory period, volume forecasts change over time. Queensland Rail remains exposed to the risk that the current volume forecast fails to materialise.

Aurizon Coal and Bulk submitted that two of the three mines using the West Moreton system faced challenges to their operation, noting:

- New Acland Stage 3 was facing a challenge from the Oakey Coal Action Alliance relating to the Queensland Government's decision to grant an associated water licence to New Acland, which could hinder its future operations
- New Wilkie Energy entered receivership in early 2024 and the New Wilkie mine was (at the time of Aurizon Coal and Bulk's submission) in care and maintenance while alternatives for its future operation were considered.⁴⁶²

We have therefore calculated the asset betas for our sample of regulated energy and water businesses and a sample of toll roads businesses to inform our estimate Queensland Rail's beta (see Table 6).⁴⁶³

⁴⁶⁰ Queensland Rail, sub. 14, p. 70.

⁴⁶¹ Queensland Rail, sub. 14, p. 71.

⁴⁶² Aurizon Coal and Bulk, sub. 2, pp. 60–61.

⁴⁶³ The businesses included in each of these samples, as well as the approach applied to estimate the asset betas, are outlined in Appendix C.

Table 6: QCA asset beta estimates using 10-year weekly data

Sample	Observations	Mean	Median
Regulated energy and water businesses	37	0.39	0.38
Toll roads businesses	3	0.63	0.61

Source: QCA analysis.

We consider that the small size of the toll roads businesses sample limits the extent the resulting beta estimate can be considered a reliable upper bound estimate from which to establish a reasonable asset beta for Queensland Rail.⁴⁶⁴

Nevertheless, we consider that Queensland Rail's proposed asset beta of 0.48 and equity beta of 0.71⁴⁶⁵ to be reasonable.

We consider that an asset beta of 0.48 is supported by our beta reference points. In this regard, an asset beta of 0.48 sits above our asset beta for the regulated energy and water business, and 0.07 below the lowest asset beta observed in our toll roads sample.

We also consider that an equity beta of 0.71 provides for regulatory certainty, as an equity beta was applied to determine Queensland Rail's bottom-up WACC estimate for AU2. HoustonKemp considered that a firm's systematic risk was not expected to change significantly over time, unless broad market factors such as regulatory frameworks or market conditions changed.⁴⁶⁶

Benchmark capital structure

We consider it reasonable to apply a 40% debt and 60% equity (40% gearing) benchmark capital structure to determine our bottom-up WACC estimate for Queensland Rail's AU3.

This is consistent with the gearing approved to estimate Queensland Rail's WACC associated with providing access to coal handling services on the West Moreton route service for its AU2 regulatory period.

We consider it reasonable to maintain the current regulatory benchmark gearing for the AU3 regulatory period, given that Queensland Rail's established risk profile has not varied significantly from the previous regulatory period. As such, we consider the benchmark gearing for Queensland Rail should remain reasonably stable over this time.

Queensland Rail proposed to retain a 40% gearing ratio for the AU3 regulatory period.

Cost of debt

We consider it reasonable to adopt Queensland Rail's proposed cost of debt of 4.95% as a placeholder to determine our bottom-up WACC estimate for Queensland Rail's AU3.⁴⁶⁷

We consider it appropriate to estimate Queensland Rail's cost of debt using:

⁴⁶⁴ New Hope (sub. 5, pp. 17-18) submitted that, if considered, each of the available toll road businesses should be given no more weight than each of the regulated energy and water businesses, noting the regulated energy and water businesses sample was roughly 10 times larger than that of toll roads businesses. We have not assigned weights to our asset beta estimates obtained from the two samples. Rather, we have used these estimates as reference points to help guide our decision on an appropriate beta for Queensland Rail.

⁴⁶⁵ Using the Brealey-Myers levering formula (using a value of 0.12 for the debt beta) for de-levering and re-levering beta. ⁴⁶⁶ Queensland Rail, sub. 14, p. 69.

⁴⁶⁷ We consider that Queensland Rail's cost of debt should be estimated with reference to the provision of the declared service, and not for a standalone benchmark entity supplying rail transportation services in the West Moreton. See Aurizon Network, sub. 3, p. 9.

- a 10-year term cost of debt⁴⁶⁸
- a BBB benchmark credit rating⁴⁶⁹
- an unweighted trailing average approach,⁴⁷⁰ with annual debt tranche refinancing.

Queensland Rail proposed an indicative cost of debt of 4.95%, applying this methodology and using a 20-business day averaging period ended 30 April 2023.⁴⁷¹ Queensland Rail produced 10 annual cost of debt estimates, using 10-year corporate bond yields reported by the RBA (see Table 7). Queensland Rail also applied a 10-basis-point uplift to its cost of debt estimate for debt raising/refinancing costs.⁴⁷²

Annual averaging period	Cost of debt estimate
Ending 30 April 2014	7.18%
Ending 30 April 2015	5.09%
Ending 30 April 2016	5.27%
Ending 30 April 2017	4.69%
Ending 30 April 2018	4.48%
Ending 30 April 2019	4.65%
Ending 30 April 2020	3.35%
Ending 30 April 2021	2.78%
Ending 30 April 2022	4.03%
Ending 30 April 2023	6.95%
10-year trailing average	4.85%

Table 7: Queensland Rail's annual cost of debt estimates

Source: Queensland Rail, sub. 1, p. 87.

We consider it appropriate for the cost of debt to be updated using this methodology prior to the start of AU3. Consistent with our rate of return guidelines, Queensland Rail is proposing to update its cost of debt calculation to the average of 12-monthly observations from April to March in advance of the upcoming regulatory period.⁴⁷³

In doing so, we consider that the cost of debt data source should reflect 10-year corporate bond yields reported by the RBA. Furthermore, we consider it appropriate to linearly extrapolate Reserve Bank of Australia (RBA) 10-year bond yields to 10 years.⁴⁷⁴

We have published a supporting workbook on our website (*Cost of debt estimation*, 2024) that provides an example of this approach for estimating the cost of debt.

 ⁴⁶⁸ This is consistent with the efficient debt financing practices of regulated infrastructure entities with long-lived assets.
 ⁴⁶⁹ This reflects the benchmark credit rating approved to estimate Queensland Rail's WACC for the AU2 regulatory period. No justification has been provided to warrant a departure from this benchmark credit rating.

⁴⁷⁰ This reflects that it may be efficient for capital-intensive infrastructure firms to manage refinancing risk by staggering debt financing, rather than refinancing the entire debt portfolio over a relatively short window of time.

⁴⁷¹ Queensland Rail, sub. 1, p. 82.

⁴⁷² Queensland Rail, sub. 1, p. 87.

⁴⁷³ Queensland Rail, sub. 14, p. 77.

⁴⁷⁴ The RBA determines its 10-year bond yields by aggregating relevant bonds with a residual maturity close to the target 10year tenor, but the aggregated tenor of its 10-year bonds has tended to be marginally less than 10 years.

We also consider it appropriate to allow debt raising costs of 10 basis points within the cost of debt for the trailing average approach to account for the administrative costs associated with raising debt.

In implementing a trailing average cost of debt, the impact of the updated cost of debt may occur either through annual updates to Queensland Rail's allowable revenues, or through a true-up at the beginning of the next regulatory period. Queensland Rail does not propose arrangements in its 2025 DAU for updating allowable revenues throughout the regulatory period to reflect annual updates to the cost of debt.⁴⁷⁵ Therefore, the impact of the updated cost of debt throughout the regulatory period is to occur through a true-up at the beginning of the next regulatory period.

Annual updates would minimise mismatches between the regulatory cost of debt allowance captured in allowable revenues and the actual cost of debt from a benchmark efficient firm during the regulatory period.

We welcome further submissions from stakeholders as to whether it is appropriate for the impact of the updated cost of debt throughout the regulatory period to occur through a true-up at the beginning of the next regulatory period, or whether annual adjustments are appropriate.⁴⁷⁶

Gamma

We consider it reasonable to apply a gamma⁴⁷⁷ of 0.484 to determine our bottom-up WACC estimate for Queensland Rail's AU3.

Our approach to estimating gamma is to calculate the product of two components:

- a distribution rate, which considers the ratio of distributed imputation credits to company tax paid
- a utilisation rate, which considers the rate at which distributed imputation credits are used by investors in the market.

Consistent with our gamma value estimated as part of our rate of return review, Queensland Rail proposed a gamma of 0.484 to determine its WACC estimate for the AU3 regulatory period.⁴⁷⁸

We have recalculated gamma applying the approach adopted in our rate of return review. The most recent assessment of the utilisation rates of imputation credits involves examining equity ownership over the period 2019 to 2023. From our analysis, there has not been a material change in the gamma since our rate of return review and we do not consider there is reason to depart from a gamma estimate of 0.484 at this time.

9.4.2 Top-down assessment

In considering whether Queensland Rail's proposed WACC is appropriate to approve, we have undertaken a top-down assessment to consider whether it provides Queensland Rail with an appropriate level of compensation commensurate with the risks Queensland Rail faces.

⁴⁷⁵ As part of Queensland Rail's submission (sub. 14, p. 77), HoustonKemp reported that it understood from discussions with Queensland Rail that Queensland Rail was proposing annual updates to the trailing average cost of debt, which should result in smoother trailing average than applying an end-of-period true-up.

⁴⁷⁶ Queensland Rail's reference tariff mechanism to date has not included annual updates to re-set the prices, apart from CPI escalation. However, stakeholders have proposed an annual true-up for variance between the capital indicator and actual capital spending (see section 8.5.3). If the capital true-up were implemented, it might mitigate the extra regulatory burden of having annual tariff adjustments only for changes to the WACC.

⁴⁷⁷ Gamma is the value to investors of distributed dividend imputation credits.

 $^{^{\}rm 478}$ Based on a distribution rate of 0.88 and a utilisation rate of 0.55.

While Queensland Rail's proposed indicative rate of return of 7.39% is above our bottom-up WACC estimate, we consider that it will provide Queensland Rail with a reasonable rate of return for the AU3 regulatory period.

Queensland Rail's overall approach for calculating its proposed rate of return is consistent with that applied to calculate the approved rate of return for the 2020 access undertaking. Differences between the Queensland Rail's proposed rate of return and the rate of return approved as part of the 2020 access undertaking largely reflect timing differences in the calculation of specific timevariant WACC parameters. In updating these time variant parameters for AU3, Queensland Rail has applied the relevant methods and values set out in our rate of return review.

We consider that taking this approach to calculate the rate of return for the 2025 DAU is reasonable, given there have been no material changes in Queensland Rail's overall risk profile.

In assessing Queensland Rail's proposed rate of return, we have also compared Queensland Rail's proposal to the WACC values adopted for a number of other Australian regulated entities. In doing so, we have undertaken a normalisation exercise⁴⁷⁹ (see 'Normalisation' below) for timing differences to make for a more meaningful comparisons across regulatory decisions. This comparative analysis supports our preliminary view that Queensland Rail's proposed WACC:

- compensates Queensland Rail for the systematic risk that it faces in providing access to coal handling services on the West Moreton system
- does not result in access charges being set at a level that is inefficiently high.

Furthermore, it is important for Queensland Rail to be able to earn a rate of return that provides sufficient incentives to promote efficient investment in the network. If the rate of return is too low, it could have a 'chilling' effect on investment, leading to inadequate capacity and/or service quality. Importantly, Queensland Rail has identified that significant capital expenditure is required over the AU3 regulatory period to deliver the level of capacity requested by access holders. From our analysis, we consider it appropriate to exercise judgement beyond our bottom-up WACC estimate so that Queensland Rail's WACC provides a reasonable overall rate of return.

Normalisation

We have compared Queensland Rail's WACC against the corresponding nominal post-tax WACCs calculated for a range of Australian regulated businesses. In doing so, we have sought to calculate the time-varying parameters of the other regulators' WACC values with reference to the averaging period applied in Queensland Rail's proposed rate of return (April 2023) (Figure 10).⁴⁸⁰ Where a regulated business's WACC has also been calculated in 2023, we have made a direct comparison with that approved WACC. We will seek to update these estimates as part of our final decision.

⁴⁷⁹ To provide for a more like-for-like comparison of the WACC values, we have sought to calculate the time-varying parameters of the other regulators' WACC values with reference to the same point in time.

⁴⁸⁰ That is, we have sought to update the risk-free rate and debt risk premium applying the same method for calculating these parameters as outlined in the respective regulatory decisions.



Figure 10: Normalised WACCs for selected Australian regulated businesses (2023)

Notes: 1. The Australian Energy Regulator (AER) WACC estimate is obtained from the AER's rate of return 2023 annual update, which used a August 2023 averaging period. 2. ARTC's approved WACC for the Hunter Valley Coal Network (HVCN) reflects a negotiated outcome between ARTC and its users. We have normalised the risk-free rate using the methodology applied to estimate the risk-free rate in the ACCC's HVCN 2017 DAU draft decision for an April 2023 averaging period. We have not normalised the cost of debt, as the approach applied in the ACCC's HVCN 2017 DAU draft decision can no longer be reproduced due to relevant information no longer being produced by the RBA. 3. The Economic Regulation Authority (ERA) WACC estimates are obtained from the ERA's 2023 determination on the WACCs for the freight and urban networks and the Pilbara railways, which used a June 2023 averaging period. 4. Aurizon Network's WACC estimate for the CQCN is obtained from our reset Schedule F values decision, which used a June 2023 averaging period.

Sources: QCA analysis; AER, Rate of return annual update 2023, December 2023; ACCC, Australian Rail Track Corporation's 2017 Hunter Valley Access Undertaking, draft decision, 20 April 2017; ERA, 2023 Weighted average cost of capital for the freight and urban networks and the Pilbara railways, final determination, September 2023; QCA, Aurizon Network's reset Schedule F values, decision, October 2023.

On a normalised basis, Queensland Rail's rate of return proposal is lower than the WACCs for the two other major coal-carrying systems in Australia, ARTC's HVCN and Aurizon Network's CQCN.

Amongst other things, we consider that differences in customer base will contribute to West Moreton coal facing a higher level of risk relative to both ARTC's HVCN and Aurizon Network's CQCN. For instance, Queensland Rail serves three West Moreton mines, all producing thermal coal, albeit of high quality. In comparison, Aurizon Network's CQCN and ARTC Hunter Valley both serve many more mines, with a substantial proportion of those mines shipping metallurgical coal, which has a higher value and is likely to have a longer-term commercial future.

Therefore, all else being equal, it may be considered reasonable for West Moreton coal services to have a higher WACC than either Aurizon Network or Hunter Valley. However, as noted by New Hope and Yancoal, both of those network operators have made behavioural commitments to their customers, as part of agreed commercial arrangements that included the WACC.

In this regard, New Hope and Yancoal submitted that some of Queensland Rail's industry peers in Australia were difficult to compare directly because of different circumstances:

• The ARTC's WACC was negotiated as part of a broader package of commitments.⁴⁸¹

⁴⁸¹ New Hope, sub. 5, p. 22; Yancoal, sub. 9, p. 10.

- Aurizon Network's WACC was commercially agreed as a result of the introduction of performance targets and independent capacity assessments that did not apply to Queensland Rail.⁴⁸²
- Arc Infrastructure's WACC was benchmarked against North American railroads, which had a different risk profile from Queensland Rail's West Moreton coal services.⁴⁸³

While Queensland Rail has not negotiated an appropriate WACC with its coal-handling customers on the West Moreton system, Queensland Rail is proposing to undertake the capital expenditure required to be able to support the volumes requested by access holders and access seekers.

We consider that our comparison of other regulatory decisions supports Queensland Rail's proposed WACC being reasonable, and likely to promote efficient investment in the infrastructure required to provide the West Moreton route service.

However, a final view on the WACC will depend in part on any positions agreed between Queensland Rail and its customers as part of their reference tariff negotiations, including risk allocation and system coordination. These positions may affect our assessment of the risks faced by Queensland Rail which, in turn, may affect our assessment of an appropriate WACC. For avoidance of doubt, we will be favourably disposed to approving a WACC that forms part of an agreed package of pricing and related measures brought to us by Queensland Rail and its customers.

Queensland Rail's proposed adjustment to its bottom-up estimate

In proposing an indicative WACC of 7.39%, Queensland Rail applied an uplift of 0.6% to its bottomup estimate.⁴⁸⁴ This is consistent with the approach applied for the AU2 regulatory period, where an uplift to the cost of debt was provided to compensate for volume uncertainty.⁴⁸⁵

Queensland Rail submitted that it still faced the same volume uncertainty on the West Moreton system that it faced in the AU2 period. As such, Queensland Rail considered that a need remained for an adjustment to the benchmark bottom-up WACC.⁴⁸⁶

New Hope and Yancoal considered that Queensland Rail's proposed adjustment to the WACC was not justified. New Hope considered that the circumstances for AU3 were markedly different. In this regard, the short-term uncertainty present at the time of the AU2 decision had been substantially addressed⁴⁸⁷, while longer-term risk factors had been addressed through other elements of Queensland Rail's proposal. Yancoal also considered that Queensland Rail's exposure to volume risk had been significantly reduced.⁴⁸⁸

We consider that there is evidence that the volume uncertainty remains a concern for the AU3 regulatory period. None of the mines that use West Moreton coal haulage services is contracted for the full term of the undertaking. The recent uncertainty about the Wilkie Creek mine also highlights the risks Queensland Rail faces, particularly with only three customers. In assessing the appropriate WACC in our final decision, we will have regard to any changes that may resolve some of these volume-related matters before the start of the AU3 regulatory period.

⁴⁸² Yancoal, sub. 9, p. 10.

⁴⁸³ New Hope, sub. 5, p. 22.

⁴⁸⁴ Queensland Rail, sub. 1, p. 29.

⁴⁸⁵ This uplift to the cost of debt was equivalent to around a 0.6% adjustment to the bottom-up WACC estimate.

⁴⁸⁶ Queensland Rail, sub. 1, p. 29.

⁴⁸⁷ New Hope submitted that an adjustment to the bottom-up WACC estimate was approved for the AU2 regulatory period to specifically to address short-term uncertainty around approvals for the New Acland Stage 3 mine development.

⁴⁸⁸ New Hope, sub. 5, pp. 14–15; Yancoal, sub. 9, pp. 3, 9–11.

In this assessment of Queensland Rail's indicative WACC proposal, we have not formed a view on the merits of applying a specific adjustment or 'top-up' to the bottom-up WACC estimate consistent with the approach used to estimate Queensland Rail's WACC for AU2. Rather, we have considered whether overall the proposed WACC provides Queensland Rail with an appropriate level of compensation commensurate with the risks it faces.

Summary 9.3

We are minded to approve Queensland Rail's proposed indicative rate of return proposal with an updated risk-free rate and cost of debt.

9.5 Capital expenditure

The capital indicator is an ex ante estimate of the capital spending that will be incurred during a regulatory period. Our approval of a capital indicator does not imply that we have accepted that level of capital expenditure should be included in Queensland Rail's RAB (see section 8.3.1). Rather, at the end of each year, Queensland Rail is required to provide us with a capital expenditure claim and we then assess the claim.⁴⁸⁹ Once we make a decision about that claim, it then becomes approved capital expenditure, which is added to the RAB.

Queensland Rail proposed a total capital indicator of \$346.9 million for AU3, a significant increase from the AU2 total capital indicator of \$153 million.⁴⁹⁰ Queensland Rail said this higher capital expenditure was necessary as the network was not currently able to support the peak forecast capacity of 9.6 mtpa⁴⁹¹ and required significant investment to reduce speed restrictions and increase reliability of the network so that the forecast capacity could be delivered. Analysis of track condition data shows that, while track condition has generally improved since 2018, the Toowoomba Range crossing remains the area with the most track geometry challenges (see Appendix B).

Table 8 shows the capital indicator by year and by section of the network.

⁴⁸⁹ Further information on the assessment process is on our website at <u>https://www.qca.org.au/project/queensland-rails-</u> 2020-access-undertaking/capital-expenditure-claim-2/.

⁴⁹⁰ While not part of this decision, Queensland Rail has proposed an additional \$40.6 million in capital expenditure during the AU2 period, effectively increasing the AU2 capital indicator. We understand Queensland Rail intends to put this additional expenditure forward as part of a separate DAAU.

⁴⁹¹ See section 9.2.

-					-	
Corridor	2025-26	2026-27	2027-28	2028–29	2029-30	Total
Rosewood- Jondarayan	\$66.6	\$70.0	\$10.1	\$15.4	\$7.5	\$169.6
Jondarayan- Macalister	\$32.4	\$33.9	\$11.1	\$9.5	\$3.2	\$90.2
Macalister- Columboola	\$6.9	_	\$21.7	\$24.4	\$33.9	\$87.0
Total	\$105.9	\$104.0	\$43.0	\$49.3	\$44.6	\$346.9

Table 8: Proposed capital expenditure by year and corridor (\$2025-26, excluding IDC)

Source: Queensland Rail, sub. 1, p. 27.

Queensland Rail proposed 'triggers' for a reset of the reference tariff and works program should any contract not be renewed during the AU3 period (see section 8.5.2). This would require Queensland Rail to adjust its capital and maintenance programs to meet the revised expected demand of the remaining system users.

Stakeholders expressed concern over the amount of capital expenditure proposed.⁴⁹² They said a lack of transparency made it difficult to assess the proposal.⁴⁹³

The appropriateness of the capital expenditure program largely depends on the level of demand and system capacity. Given the volume uncertainty (see section 9.2) it is difficult to assess the appropriateness of the capital indicator proposed by Queensland Rail. The proposed capital indicator represents both a significant challenge for Queensland Rail in delivering these works and a significant potential cost to users.

As outlined in section 9.2, we do not consider the forecast volumes to be appropriate to approve at this time due to the level of uncertainty. However, we have used the volume and cost forecasts proposed by Queensland Rail to provide indicative information on efficient levels of capital, maintenance and operational allowances for Queensland Rail's West Moreton coal services.

9.5.1 Arcadis analysis

Engineering consultancy Arcadis has advised us that, based on the tonnage forecast by Queensland Rail, it considers 14 out of the 17 project types in the proposed capital indicator to be reasonable. However, it identified that the capital indicator proposed may not be the most efficient way to deliver the capacity needed. Table 9 shows the differences in capital expenditure between Queensland Rail's proposal and alternative estimates prepared by Arcadis.

⁴⁹² New Hope, sub.5, pp. 8-9; Yancoal, sub. 9, pp. 3, 13-14.

⁴⁹³ Aurizon Coal and Bulk, sub. 2, p. 72; New Hope, sub. 5, pp. 8-9; Yancoal, sub. 9, p. 3.

Table 9: Differences between Queensland Rail capital indicator and Arcadis analysis

Project	Arcadis assessment (\$m)
Track Reconditioning	-93.7
Re-sleepering	-6.9
Bridge Pier Replacement	-20.5
Total	-121.1

Source: Arcadis, <u>Review of West Moreton System Costs and Other Technical Matters in Queensland Rail's 2025 DAU</u>, May 2024, pp. 24-25.

The differences in amounts for the capital indicator are predominately due to Arcadis concluding that capital works proposed in the west of the network (between Macalister and Columboola) are not reasonable, and it would be more efficient to continue with current maintenance practices for this section of track, rather than conduct capital works.⁴⁹⁴

Arcadis questioned whether greater efficiencies could be realised by having regard to the likely asset lives. While Queensland Rail's proposed work adhered to general engineering standards, Arcadis questioned whether this proposal had fully taken into account the preferred risk profile of users and an environment where assets are expected by Queensland Rail to have reached the end of their economic lives in 19 years.⁴⁹⁵

Arcadis also highlighted issues such as Queensland Rail's long approval processes for large capital works and labour market constraints resulting from mobilisation for rail projects such as Sydney Metro, Inland Rail and Cross River Rail as risks to the delivery of the works underpinning the capital indicator.

9.5.2 Capital indicator

As discussed in section 9.2, there is a level of uncertainty around the volumes that will ultimately be transported under AU3. Given how crucial demand levels are to the capital works programs underpinning the capital indicator, it is important to first reach agreement between all parties on the most appropriate level of demand.

Stakeholders were concerned about the level of capital expenditure and considered that an annual review process could be included in the 2025 DAU, whereby Queensland Rail engaged with users regarding capital works programs (see section 8.3.2).⁴⁹⁶ Clearly, there is an appetite from users to engage with Queensland Rail on this matter, as well as an overriding concern over the level of the reference tariff.

While this section focuses on the capital indicator proposed by Queensland Rail, it cannot be considered in isolation – it must also be considered in context of how it impacts users and their investment plans. Ultimately, there is a direct but complex relationship between users' need for network investment to enable their operations and the network owner's need for user demand for it to continue to operate.

⁴⁹⁴ Our analysis of track condition data confirms that track geometry issues are largely concentrated at the eastern end of the system in and around the Toowoomba Range crossing (see Appendix B).

⁴⁹⁵ Arcadis, Review of West Moreton System Costs and Other Technical Matters in Queensland Rail's 2025 DAU, May 2024 p. 22.

⁴⁹⁶ Aurizon Network, sub. 3, p. 5; New Hope, sub. 12, p. 4; Yancoal, sub. 9, p. 15.

We consider the objectives of the QCA Act are best achieved by Queensland Rail and its customers working towards agreed approaches. Given stakeholders desire to engage and the significant price impact of these programs, we encourage Queensland Rail to engage with users and above-rail operators not only so that they may be able to agree on alternative capital programs which minimise costs through greater consideration of the interaction between service levels and above-rail costs, but also to consider arrangements to help maintain this consensus throughout the 2025 undertaking period.⁴⁹⁷

Summary 9.4

It may not be appropriate to approve the capital indicator proposed by Queensland Rail based on the information currently available.

9.6 Asset lives

As discussed in section 8.5.1, the 2025 DAU proposes an accelerated depreciation profile based on forecast mine lives estimated by AME, advising Queensland Rail. Queensland Rail proposed that:

- existing assets be depreciated over 19 years, replacing the current depreciation method of technical life. Assets that have a remaining life of less than 19 years would continue to be depreciated over that remaining period
- DAU3 capital expenditure be depreciated over 14 years, indexed to 1 July 2025. This would result in spending in subsequent years having a declining maximum depreciation life, reducing to nine years for assets built in the last year of the undertaking period.⁴⁹⁸

Applying the asset lives proposed would be a significant revision to asset lives currently applied to West Moreton system assets as shown in Table 10.

⁴⁹⁷ See also section 8.3.

⁴⁹⁸ For example, capital expenditure proposed in July 2026 would have a theoretical maximum depreciation profile of only 13 years. Capital expenditure is not depreciated until it is completed, meaning that expenditure for the July 2026 year is depreciated for only 6 months in the first year, resulting in a maximum practical depreciated life of 12.5 years. Therefore, capital expenditure proposed in the final year of the DAU3 would have a maximum practical depreciation schedule of 9.5 years.

Table 10: AU2 West Moreton asset lives

Asset lives	Years
Track (including turnouts)	35
Roads	38
Fences	20
Signals	20
Bridges	100
Tunnels	100
Culverts	100
Earthworks	100
Other	20
Land acquisition costs	50
Telecommunications	20
Land	0

Source: Queensland Rail, sub. 1, p. 34.

Stakeholders did not suggest alternative asset lives to those proposed by Queensland Rail for consideration.

We note that, so far, the asset lives proposed by Queensland Rail are the only ones put forward for consideration, and that the general principle of Queensland Rail being entitled to recover investments over the economic lives of its assets has not been challenged.

Our preliminary view is that the asset lives proposed by Queensland Rail may be reasonable and likely to promote efficient investment in and use of the West Moreton route service, and may be in the interests of Queensland Rail (s. 138(2)(a), (b)). However, as discussed in section 8.5, the first step to finding the most economically efficient outcome will be negotiation between Queensland Rail and its users that takes into account asset lives along with other risks and considerations.

Summary 9.5

The asset lives proposed by Queensland Rail may be appropriate to approve.

9.7 Appreciation

Queensland Rail proposed to continue escalating the RAB, maintenance costs and operating costs by inflation each year. For the purposes of the DAU Queensland Rail proposed to apply forecast inflation rates of 3% for the first three years of AU3, and 2.5% for the final two years. As with the

current arrangements, these figures will be updated with actual inflation each year when the RAB is rolled over. Queensland Rail's proposal was supported by New Hope.^{499, 500}

Indexation is a well-established regulatory approach to managing the real value of a RAB, and Queensland Rail's proposal is appropriate to approve.

Summary 9.6

It is appropriate to approve Queensland Rail's proposed approach to RAB appreciation based on the information currently available.

9.8 Maintenance expenditure

Queensland Rail proposed a significant increase in maintenance expenditure, driven primarily by the need to accommodate 9.6 mtpa across the West Moreton system. Figure 11 outlines historical and proposed expenditure during the AU2 and DAU3 periods, allocated by corridor.



Figure 11: Maintenance expenditure FY21-30

Source: Queensland Rail's DAU 3 explanatory document, p. 45.

The \$172.5 million proposed maintenance allowance is a \$69.2 million increase over the \$106.5 million approved for AU2. Most of the proposed increase (\$55.9m) is related to increased track

⁴⁹⁹ New Hope, sub. 12, p. 2.

⁵⁰⁰ We note that while Aurizon Network proposed cessation of RAB indexation, this was in the context of accelerated depreciation. See Aurizon Network, sub. 3, p. 21.

Queensland Rail's 2025 Draft Access Undertaking

maintenance. Queensland Rail said the increased maintenance will also require more train path possessions for maintenance activities compared to current levels. To minimise this potential impact on capacity, Queensland Rail proposed to increase staffing levels on maintenance crews, to complete more maintenance tasks within restricted possession periods.

Stakeholders were concerned with the significant increases in maintenance costs proposed by Queensland Rail⁵⁰¹, as well as the difficulty in assessing the proposal based on the amount of information provided.⁵⁰²

9.8.1 Relationship with the capital indicator

The required level of maintenance is closely linked to the proposed capital works, as upgraded or renewed sections of track generally require less maintenance. Often one reason why capital works are conducted is that in the long run it is more efficient to replace assets as that will reduce maintenance costs.

As discussed in section 9.5, Arcadis has found that it would be more efficient to retain existing maintenance practices between Macalister and Columboola, due to the capacity remaining the same as present, and due to that section of track having relatively low utilisation, so there are sufficient spare paths available for maintenance possessions.

This approach west of Macalister would be a significant change to both the capital and maintenance programs proposed by Queensland Rail in the material accompanying the 2025 DAU. Arcadis recommended maintaining existing maintenance practices, as opposed to capital works, west of Macalister. However, it said it was unable to obtain sufficiently detailed information from Queensland Rail about maintenance activities to calculate the associated costs that should be included. As a result, we anticipate the maintenance allowance may need to be revised in line with demand and capital indicator changes, taking into account further information provided by Queensland Rail on its maintenance plan.⁵⁰³

9.8.2 Maintenance expenditure

Given the size of the maintenance expenditure proposed, the potential efficiencies identified by Arcadis and the impact of the proposed maintenance allowance on the reference tariff, we consider that users, above-rail operators and Queensland Rail may be able to agree on an alternative maintenance program that reflects updated demand forecasts and capital expenditure programs that minimise costs through greater consideration of the interaction between service levels and above-rail costs.

We consider the objectives of the QCA Act are best achieved by Queensland Rail and its customers working towards agreed approaches. Given stakeholders' desire to engage, the potential efficiencies identified by Arcadis, and the significant price impact of these programs, we encourage Queensland Rail to engage with users and above-rail operators so that they may be able to agree on alternative maintenance programs that minimise costs through greater consideration of the interaction between service levels and above-rail costs.⁵⁰⁴

⁵⁰¹ Aurizon Coal and Bulk, sub. 2, p. 5; New Hope Group, sub. 5, p. 10; Yancoal, sub. 9, p. 3.

⁵⁰² New Hope Group, sub. 5, pp. 10-11.

⁵⁰³ Arcadis also examined the split between fixed and variable maintenance costs. See Arcadis, *Review of West Moreton System Costs and Other Technical Matters in Queensland Rail's 2025 DAU*, May 2024, pp. 41-44.

⁵⁰⁴ See also section 8.3.

Summary 9.7

It may not be appropriate to approve the maintenance allowance proposed by Queensland Rail based on the information currently available.

9.9 Operating expenditure

Queensland Rail proposed an 89% increase (in real terms) in operating expenditure compared to AU2 (Figure 12).



Figure 12: West Moreton operating cost allowances

Source: Queensland Rail, sub. 1, p. 48.

Queensland Rail said it had revised its approach to forecasting operating costs, and that the methodologies employed in AU1 and AU2 'underestimated long-term operating costs'.⁵⁰⁵ The approach for costs proposed in DAU3 is based on escalating Queensland Rail's claimed 2021-22 costs to \$2025-26, with individual cost categories escalated separately⁵⁰⁶ to obtain a forecast amount. Table 11shows Queensland Rail's analysis of the allowances, actuals, and differences for past West Moreton system operating costs.

⁵⁰⁵ Queensland Rail, sub. 1 p. 48.

⁵⁰⁶ In the case of train control costs, a bottom-up escalation was used.

	2016-17	2017-18	2018-19	2019-20	2020-21	2021-22
QCA allowance	7.2	7.4	7.6	7.7	7.2	7.5
Operating costs	8.1	11.7	16.0	12.3	12.4	11.4
Difference	(0.9)	(4.3)	(8.4)	(4.6)	(5.2)	(3.9)

Table 11: West Moreton operating costs allowances vs actuals (\$m)

Source: Queensland Rail, sub. 1, p. 48.

However, Queensland Rail did not provide information to explain the significantly higher operating costs starting in 2017-18, that continued during the AU2 period, or whether these higher costs were efficient. It justified changing the approach to forecasting on the basis that the new estimates corresponded to increased actual operating costs.

Stakeholders were concerned with the significant increases in operating costs proposed by Queensland Rail⁵⁰⁷, as well as the difficulty in assessing the proposal based on the amount of information provided.⁵⁰⁸ Aurizon Network said the increases in operating costs were not explained and that Queensland Rail had not provided evidence that they represented efficient costs.⁵⁰⁹ Queensland Rail provided some analysis of relative corporate overheads in a study by ARUP.⁵¹⁰ However, this analysis was based on averages based on the length of track, and there are multiple other factors that may be useful as the basis for benchmarks.

Arcadis assessed the proposed operating costs based on information provided by Queensland Rail as well as benchmarking against operating expenses in international operations. It found that 8 of the 9 proposed operating cost activities were reasonable, the exception being corporate overhead. Arcadis said the allowance for overheads had increased by 87% relative to 2020-21, but it had not received additional information on these costs from Queensland Rail. Arcadis concluded there was insufficient information to consider corporate overhead costs (18.6% of forecast operating costs) reasonable.⁵¹¹

As with capital and maintenance, operating costs will depend on the level of demand in the system, in particular the number of customers and the number of trains that must be managed. As such, the capital, maintenance, and operational programs are, to an extent, a package. Given the significant price impact of these combined work programs, and the questions raised by stakeholders over the proposed cost increases, we consider that Queensland Rail should consult with users and above-rail operators to agree on a package reflective of updated demand forecasts and revised programs that minimise costs through greater consideration of the interaction between service levels and above-rail costs.

⁵⁰⁷ New Hope Group, sub. 5, p. 11-12; Yancoal, sub. 9, p. 3, 16.

⁵⁰⁸ New Hope Group, sub. 5, p. 11-12; Yancoal, sub. 9, p. 3, 16.

⁵⁰⁹ Aurizon Network, sub. 3, p. 6.

⁵¹⁰ ARUP report reference.

⁵¹¹ Arcadis also examined the split between fixed and variable operating costs. See Arcadis, *Review of West Moreton System Costs and Other Technical Matters in Queensland Rail's 2025 DAU*, May 2024, pp. 41-44.

Summary 9.8

It may not be appropriate to approve the operating costs proposed by Queensland Rail based on the information available.

9.10 Metropolitan system tariff

Queensland Rail proposed to maintain the existing Metropolitan System tariff in real terms, escalating the current tariff charges by CPI.

The Metropolitan tariff has been developed for the past decade using a proxy approach that relies on prices derived for the coal services that use the West Moreton system. This approach avoided the complicated task of seeking to allocate costs for the Metropolitan system to coal services, which use only a small portion of what is predominantly a commuter network.

Queensland Rail proposed to continue this Metropolitan proxy pricing approach and escalate the 2020 undertaking price by actual and forecast CPI. We consider that the proxy approach remains an appropriate way of determining a price that sits between:

- the incremental cost which would be at or near zero
- the standalone cost which could be expected to be at least as high as the price that is being charged.

Continuing the existing proxy approach was supported by New Hope⁵¹², and no concerns were raised by other stakeholders.

Accordingly, our decision is that it is appropriate to approve continuing the 'proxy' approaches used in previous undertakings. We consider this simple, transparent approach continues to be in the interests of Queensland Rail, access seekers and access holders (ss. 138(2)(b), (e) and (h)).⁵¹³

Summary 9.9

It may be appropriate to approve the proposed Metropolitan system tariff approach.

⁵¹² New Hope, sub. 5, p. 23.

⁵¹³ We note this would still leave the way open for Queensland Rail to apply in the future to implement a Metropolitan-specific asset base and our decision would not predetermine our consideration of any future DAU.

Appendix A: The access regime

A.1 Queensland access regime

Part 5 of the QCA Act sets out the statutory regime for access to services in Queensland, including the process for declaration of services and the negotiation of terms of access. The regime is based on a negotiate-arbitrate framework, which envisages that, in the first instance, access to a declared service should be procured on the basis of terms and conditions that are commercially agreed between the access seeker and the provider of the declared service.

The regime also provides for developing an 'access undertaking', which is defined under the QCA Act as 'a written undertaking that sets out details of the terms on which an owner or operator of the service undertakes to provide access to the service whether or not it sets out other information about the provision of access to the service'.⁵¹⁴ It is notable that since implementation of the Queensland access regime, access to declared services has primarily been facilitated through approved access undertakings. These have been the primary means of setting out: (i) a process required for an access seeker to negotiate access; (ii) general terms and conditions that apply when negotiating access agreements; and (iii) how disputes in relation to access are to be resolved.

A.2 Queensland Rail

Queensland Rail provides access to declared services for the purposes of the access regime. The relevant services involve 'the use of rail transport infrastructure for providing transportation by rail if the infrastructure is used for operating a railway for which Queensland Rail Limited, or a successor, assign or subsidiary of Queensland Rail Limited, is the railway manager'.⁵¹⁵ On 1 June 2020, the Treasurer and Minister for Infrastructure and Planning declared the following parts of the Queensland Rail service, which parts are each a service, under section 84(1)(b) of the QCA Act: (a) the North Coast Route service; (b) the Mount Isa Route service; (c) the West Moreton Route service; (d) the Central Western Route service; (e) the Western Route service; and (f) the South Western Route service (collectively referred to in this draft decision as the 'declared services').⁵¹⁶

As a result of this service declaration, Queensland Rail is subject to various obligations under the QCA Act, including engaging in negotiations in good faith regarding access to the service (s. 100), providing information to access seekers regarding the service, and participating in access dispute resolution.

The following are previous access undertakings involving Queensland Rail:517

- 2008 access undertaking 2010-2015
- 2016 access undertaking 2016-2020
- 2020 access undertaking (AU2) 2020-2025.

⁵¹⁴ QCA Act, Sch. 2.

⁵¹⁵ The declaration of Queensland Rail's below-rail infrastructure is set out in s. 250(1)(b) of the QCA Act.

⁵¹⁶ Queensland Government, <u>Gazette</u> (Extraordinary), vol. 384, no. 31, 1 June 2020.

⁵¹⁷ Additional information is available on the QCA website at https://www.qca.org.au/project/queensland-rail/previousaccess-undertakings/.

Appendix B: Track quality

We have analysed track geometry data provided by Queensland Rail to inform our analysis and assist stakeholders to better understand track condition in the West Moreton system. Stakeholders said that limited disclosure by Queensland Rail prevented them from verifying key performance measurements, including the OTCI.⁵¹⁸

We found that the condition of the West Moreton system has improved since 2018, due in part to a reduction in the proportion of the track where the twist measure is out specification. The Toowoomba range crossing remains the primary location for rail condition hotspots.

Overall track condition index

The OTCI is a measure reported on by Queensland Rail to give a general indication of the condition of its rail infrastructure.⁵¹⁹ It combines a range of key measures of track geometry, to indicate how the infrastructure is performing and help identify areas requiring maintenance or asset upgrades. It is a broad indicator and is best used to monitor overall trends in track condition.⁵²⁰ OTCI values that have decreased over time may suggest that the track condition has improved, while the opposite applies for OTCI values that have increased. Queensland Rail provided OTCI data for 2018 to inform our investigation of the 2020 DAU and provided updated data in 2023 to assist with our consideration of the 2025 DAU. We split our analysis for both years into three main geographical sections along the West Moreton system: Rosewood-Jondaryan, Jondaryan-Macalister, and Macalister-Columboola.⁵²¹



Figure 13: OTCI values for 2018 and 2023

⁵¹⁸ Yancoal, sub. 9, p. 14.

⁵¹⁹ Measures similar to the OTCI are used by many rail operators, including Aurizon Network. However, the exact implementation can vary.

⁵²⁰ The sum of three times the standard deviation for each parameter was calculated for each 100 m section of the track. The OTCI is the average of all these 100 m sections along the entire track.

⁵²¹ Queensland Rail provided track measurements from August 2018 for the QCA's investigation for AU2 and track measurements from September 2023 for the QCA's investigation for AU3.

	Columboola-Macalister	Macalister-Jondaryan	Jondaryan-Rosewood
2018	30.75	30.76	60.92
2023	24.60	23.50	44.83

Table 12: OTCI data summary

Source: Queensland Rail data and QCA analysis.

For all three sections of track, the OTCI values have decreased between 2018 and 2023, suggesting that track quality has generally improved. While the OTCI values for Columboola-Macalister and Macalister-Jondaryan are similar, values for the Jondaryan-Rosewood corridor are significantly higher for both the 2018 and 2023 data. These elevated OTCI values may be explained by the greater amount of traffic that runs through the Jondaryan-Rosewood corridor, which also includes the steep and winding Toowoomba Range crossing.

Twist and gauge

The Civil Engineering Track Standards (CETS) provide maintenance thresholds for various track geometry parameters to maintain consistency and safety throughout the network. Should track data measurements exceed these limits, maintenance work is carried out. Based on advice from Queensland Rail, for major exceedances (>30 mm), maintenance work is carried out within 1 day, whereas minor exceedances (25-30 mm) have a window of 7 days.

During the investigation for the 2020 access undertaking (AU2), our consultant SYSTRA analysed the 2018 track quality data, with a particular focus on a parameter known as 10 m twist. Twist is defined as the rate of change of the cant (superelevation) over a particular interval. In other words, it measures whether the two rails, connected by sleepers, are at the same level over the interval, or whether one rises or falls relative to the other (i.e. whether there is a 'twist' in the track). The importance of twist among other geometric parameters has been highlighted in recent research by Abadi et al. (2023), where twist was found to play a significant role in how unbalanced loads to either the left or right rail could lead to rocking movements and even derailment.⁵²² SYSTRA's analysis included plotting the distribution of the 10 m twist and calculating the distance exceeding the CETS limits. We conducted similar analysis using the most recent data and compared results.

⁵²² 'Among geometrical parameters, twist plays an important role in the operation of railway lines and it has rarely been studied. It affects interaction of track and wheels, can cause an unbalanced load to the left and right rails and imposes more dynamic loads to ballast and sleepers. In severe cases it leads to rocking movements and even derailment. Twist along the railway track causes different forces in wheel and rail leading to geometrical failures.' (EIY Abadi, AM Sameni and M Yaghini, 'Analysis of the relationship between geometric parameters of railway track and twist failure by using data mining techniques', Engineering Failure Analysis, vol. 143, part A, 2023, doi:10.1016/j.engfailanal.2022.106862.)

Figure 14: Twist distributions over 10 m intervals



Source: Queensland Rail data and QCA analysis.

The data relating to the percentage of track exceeding the CETS thresholds is summarised in Table 13.

	1-day threshold	7-day threshold
2018	0.16%	0.44%
2023	0.03%	0.17%
Difference (2023 vs 2018)	81.25% lower	61.36% lower

Table 13: Summary of 10 m twist data

Source: Queensland Rail data and QCA analysis.

One-day (>30 mm) and 7-day (25-30 mm) CETS thresholds

Exceedances in the 1-day threshold are any twist measurements that exceed 30 mm, over a 10-metre section of track. These are to be rectified within 1 day. Similarly, exceedances in the 7-day threshold are any twist measurements that are within 25-30 mm. These are to be rectified within 7 days.

Figure 14 shows a clear narrowing of the 2023 twist distribution compared to the 2018 distribution. A higher proportion of data points are clustered around 0 mm, and the percentage of track exceeding the CETS thresholds has also decreased significantly. This decrease in the percentage of track length that exceeds the maintenance limit may suggest that the formation renewal and track reconditioning programs have been successful in achieving their aims.

Given the relatively elevated OTCI values for the Jondaryan-Rosewood corridor, we also plotted the distribution of hotspots relative to location.⁵²³ We define a hotspot as any measurement that exceeds

⁵²³ Due to the 2023 data containing significantly more data points, the relative frequency was taken to provide a fair comparison between the 2018 and 2023 datasets.

the CETS 7-day limit. Most of the hotspots are also located in the Jondaryan-Rosewood section of the network.⁵²⁴



Figure 15: Twist hotspots exceeding a 7-day response threshold

Source: Queensland Rail data and QCA analysis.

Gauge is the distance (separation) between the two rails. Using the gauge thresholds for maintenance intervention provided by Queensland Rail, we can analyse hotspots in the network for both 2018 and 2023 data. The result is similar to the twist analysis, with most of the hotspots in the Jondaryan-Rosewood corridor, clustered just east of Toowoomba on the range crossing.





Source: Queensland Rail data and QCA analysis.

Twist and gauge measurements of railway tracks are an important indicator of track quality and are affected by maintenance and formation renewal programs. While the relative frequency of twist hotspots has decreased since 2018, there does not appear to be any improvement in gauge hotspots. By providing the OTCI and an overview of track quality, we hope to assist stakeholders in gaining a clearer understanding of the track condition in the West Moreton system.

⁵²⁴ Specifically, the hotpots are clustered just east of Toowoomba. Geographically, this is on the range leading up to Toowoomba from Brisbane.

Appendix C: Beta data

To estimate the beta for any given firm, we regressed its returns data against the returns of a proxy for the market portfolio in the home economy, such as the All Ords or ASX200 for Australia and the S&P500 for the United States.

To estimate beta, data must be captured over a period of time. In choosing the time horizon of the data that is used, there is an inherit trade-off between capturing a large amount of data to try and limit the standard errors of the estimate, while also capturing information that is relevant to the risk profile of the firm.

We have used a 10-year return window. In doing so, we have used weekly data, as we consider it strikes a balance between having a large number of observations and also being unlikely to capture statistical noise that might possibly be accompanied by higher-frequency (e.g. daily) return intervals.

We note that when using weekly returns, there is the potential for estimates to vary by non-trivial amounts depending on the reference day selected. Therefore, we have taken an average of each of these weekly reference day combinations to deal with the potential variation that may arise when estimating beta.

The firms included in our regulated energy and water businesses and toll roads businesses samples, are outlined in Tables 14 and 15**Error! Reference source not found.**, respectively.

Company
Alliant Energy Group
Ameren Corporation
American Electric Power
APA Group
Avista Corp
Black Hills Corp
Canadian Utilities Ltd
CMS Energy Corp
Consolidated Edison Inc
Dominion Energy Inc
Duke Energy Corp
Edison International
Emera Inc
Eversource Energy
FirstEnergy Corp
Fortis Inc
Idacorp Inc

Table 14: Firms included in our regulated energy and water businesses sample

Company	
MGE Energy Inc	
National Grid Plc	
North Western Corp	
PNM Resources Inc	
Portland General Electric Co	
PPL Corp	
Sempra Energy	
Southern Co	
WEC Energy Group Inc	
Xcel Energy Inc	
American Water Works Co Inc	
American States Water Co	
Artesian Resources Corp	
California Water Service Grp	
Middlesex Water Co	
SJW Group	
Essential Utilities Inc	
York Water Co	
Severn Trent Plc	
United Utilities Group Plc	

Table 15: Firms included in our toll roads businesses sample

Company
Atlas Arteria
Getlink SE
Transurban Group

-

Appendix D: Submissions received

Submission	Number
Queensland Rail initial submission	
Queensland Rail's DAU3 explanatory document	1
Submissions in response to the DAU	
Aurizon Coal and Bulk	2
Aurizon Network	3
Centrex	10
GrainCorp	4
New Hope Group	5
North West Phosphate	6
Pacific National	7
Qube Logistics	8
Yancoal	9
Responsive submissions	
Glencore	11
Mount Isa Line Users	17
New Hope Group	12
Pacific National	13
Queensland Rail	14
Rail Operator Group	15
Yancoal	16

Glossary

ACCC	Australian Competition and Consumer Commission
AER	Australian Energy Regulator
ARA	Australian Railway Association
ARTC	Australian Rail Track Corporation
AU	access undertaking
AU2	Queensland Rail's 2020 access undertaking
AU3	Queensland Rail's 2025 access undertaking (once approved)
CETS	Civil Engineering Track Standards
CPI	Consumer price index
CQCN	Central Queensland coal network
DAU	draft access undertaking
DBCT	Dalrymple Bay Coal Terminal
DBI	Dalrymple Bay Infrastructure
DTMR	Department of Transport and Main Roads, Queensland Government
DTP	daily train plan
ERA	Economic Regulation Authority
HVCN	Hunter Valley coal network
MTP	master train plan
OTCI	overall track condition index
RBA	Reserve Bank of Australia
SAA	standard access agreement
SCC	supply chain calendar
QCA	Queensland Competition Authority
WACC	Weighted average cost of capital

.

References

- Abadi EIY, Sameni AM and Yaghini M, '<u>Analysis of the relationship between geometric parameters</u> of railway track and twist failure by using data mining techniques', <u>Engineering Failure Analysis</u>, vol. 143, part A, 2023, doi:10.1016/j.engfailanal.2022.106862.
- Arcadis, <u>Review of West Moreton System Costs and Other Technical Matters in Queensland Rail's</u> 2025 DAU, May 2024.
- Australian Competition and Consumer Commission (ACCC), <u>Application for revocation of</u> <u>AA1000425 and the substitution of authorisation AA1000644</u>, determination, 1 February 2024.
- Australian Competition and Consumer Commission (ACCC), <u>Australian Rail Track Corporation's</u> 2017 Hunter Valley Access Undertaking, draft decision, 20 April 2017.
- Australian Energy Regulator (AER), *Rate of return annual update 2023*, December 2023.
- Brailsford T, Handley J and Maheswaran K, 'Re-examination of the historical equity risk premium in Australia', *Accounting & Finance*, vol. 48, 2008, pp. 85-86.
- Department of Industry, Science and Resources, <u>Resources and Energy Quarterly</u>, Australian Government, March 2024.
- Economic Regulation Authority (ERA), <u>2023 Weighted average cost of capital for the freight and</u> <u>urban networks and the Pilbara railways</u>, final determination, September 2023.
- Queensland Competition Authority (QCA), <u>QR Network 2009 Draft Access Undertaking</u>, draft decision, December 2009.
- <u>Oueensland Rail's Draft Access Undertaking</u>, decision, June 2016.
- <u>Access disputes under the QCA Act</u>, summary guide, June 2019.
- <u>Oueensland Rail draft access undertaking</u>, decision, February 2020.
- <u>Pricing of shared infrastructure for the GAPE and Newlands system</u>, guidance paper, September 2021.
- ----- <u>Queensland Rail access undertaking timeline</u>, letter to Queensland Rail, 21 September 2022.
- *<u>Aurizon Network's reset Schedule F values</u>, decision, October 2023.*
- ----<u>Rate of return review</u>, final report, version 3, February 2024.

Queensland Government, *Gazette* (Extraordinary), vol. 384, no. 31, 1 June 2020.

- ——<u>Coal industry review statistical tables</u>, Coal production data by mine, coal type and financial year [dataset], Open Data Portal, accessed 2 April 2024.
- Queensland Rail, <u>Western System Information Pack</u>, August 2016.
- TheCoalTrader.com, <u>Australian thermal coal exports drop 11% MoM in Feb 2024</u>, viewed 11 April 2024.

Thompson S, Sood K and Rapaport E, '<u>Houlihan Lokey starts New Wilkie clean-up job; flyers out</u>', *Financial Review*, 28 February 2024, viewed 11 April 2024.

United Nations Framework Convention on Climate Change (UNFCCC), <u>End of Coal in Sight at</u> <u>COP26</u>, press release, 4 November 2021, UNFCCC website, viewed 11 April 2024.

Queensland Rail's 2025 Draft Access Undertaking

.